ON SOME ASPECTS OF TOTALLY DUAL INTEGRAL SYSTEMS

bу

William J. Cook

A thesis
presented to the University of Waterloo
in partial fulfillment of the
requirements for the degree of
Doctor of Mathematics
in

Combinatorics and Optimization

Waterloo, Ontario, 1983

(c) William J. Cook

I hereby declare that I am the sole author of this thesis.

I authorize the University of Waterloo to lend this thesis to other institutions or individuals for the purpose of scholarly research.

Signature William J. Cook

I further authorize the University of Waterloo to reproduce this thesis by photocopying or by other means, in total or in part, at the request of other institutions or individuals for the purpose of scholarly research.

Signature William J. Cool

The University of Waterloo requires the signatures of all persons using or photocopying this thesis. Please sign below and give address and date.

Abstract

A system of rational linear inequalities $Ax \leq b$ is defined to be a totally dual integral system if for each integer vector w for which the linear program $min\{yb: yA = w, y \geq 0\}$ has an optimal solution, the optimum can be achieved by an integer vector y. Totally dual integral systems were introduced by A.J. Hoffman and J. Edmonds and R. Giles and are closely related to combinatorial min-max theorems. This thesis deals with general properties of totally dual integral systems and with totally dual integral systems related to matching problems. The topics studied include: 1) operations that preserve total dual integrality, 2) upper bounds on the number of nonzero variables needed in integral optimal solutions to dual linear programs associated with totally dual integral systems, 3) recognition of totally dual integral systems, and 4) minimal totally dual integral systems for matching polyhedra.

<u>Acknowledgements</u>

I am indebted to my supervisor Professor U.S.R. Murty for his help and his support. For me, Professor Murty was the ideal supervisor.

I thank Professor A.J. Hoffman for his encouragement throughout my studies and Professor P. Hell for introducing me to combinatorics.

I am grateful to the members of my examining committee: Professor R.P. Anstee, Professor A.R. Conn, Professor W.R. Pulleyblank, and Professor A. Schrijver. The members of the committee made many helpful suggestions and pointed out errors in the original manuscript.

I give special thanks to Professor R.P. Anstee for his very careful reading of the thesis.

I thank Professor W.R. Pulleyblank for many valuable conversations concerning the material presented in Chapter 3 and for other help that he has given me.

The material presented in Section 8 of Chapter 2 was developed jointly with Professor J. Fonlupt and Professor A. Schrijver and the results presented in Section 9 of Chapter 2 were obtained jointly with Professor L. Lovász and Professor A. Schrijver. I thank Professor Fonlupt, Professor Lovász, and Professor Schrijver for their contributions.

I thank the faculty members, graduate students, and staff of the Department of Combinatorics and Optimization. Among the faculty members with whom I have discussed aspects of this thesis are Professor M. Farber, Professor C. Haff, and Professor D.H. Younger. Among my fellow graduate students with whom I have worked I thank especially K. Fukuda, G. Koop, E. Korach, A. Lubiw, B. Richter, and N. Sbihi. I give special thanks to A. Mandel for so often pointing out the absurdity of my arguments.

Most importantly, I thank my parents, my sisters and their families, and all of my friends.

Table of Contents

	Page
Chapter I - Introduction	
1. Definition of total dual integrality	1
2. Background material	4
3. Summary of results	13
Chapter II - Totally dual integral systems	
1. Introduction	18
2. Integral polyhedra	20
3. Totally unimodular and balanced matrices	24
4. Anti-blocking and the perfect graph theorem	28
5. Hilbert bases and totally dual integral	
defining systems	36
6. Operations that preserve total dual integrality	43
7. Integer programming	51
8. Bounds on the number of nonzero dual variables	56
9. Recognition of totally dual integral systems	63
10. Proving total dual intgrality with cross-free	
families	77
Chapter III - Minimal systems for matching problems	
1. Introduction	86
2. Matchings and separability	88
3. b-matchings	95
4. Simple b-matchings	-00
5. Capacitated b-matchings	14
6. Triangle-free 2-matchings	L27
References 1	32

CHAPTER I

INTRODUCTION

1. Definition of total dual integrality

There are many interesting theorems in combinatorics that equate the minimum value of some function over some set and the maximum value of some other function over another set. Indeed, some of the fundamental results in graph theory are of this nature, e.g. Menger's theorem (on disjoint paths), the König-Egerváry theorem (on matchings in bipartite graphs), and the Tutte-Berge theorem (on factors of graphs). Often such min-max theorems can be phrased as "a certain linear program and its dual each have integral optimal solutions." For example, consider the König-Egerváry theorem:

(1.1) In a bipartite graph the maximum number of pairwise disjoint edges is equal to the minimum number of nodes meeting all edges.

Letting A be the edge-node incidence matrix of a bipartite graph G (i.e. A has a row for each node of G, a column for each edge of G, and $a_{ve} = 1$ if v is an end node of e and $a_{ve} = 0$ otherwise), this result is equivalent to the fact that both sides of the linear programming duality equation

(1.2) $\max\{wx: Ax \le 1, x \ge 0\} = \min\{y1: yA \ge w, y \ge 0\}$

can be achieved by integer vectors if w = 1, where 1 is the vector of all 1's (see Section 2 for conventions regarding notation). In fact, a well-known "weighted" version of the König-Egerváry theorem is equivalent to the fact that for each integer vector w both sides of (1.2) can be achieved by integer vectors.

Translating combinatorial min-max theorems into linear programming results has proven to be very useful, both in unifying known min-max theorems and in obtaining new theorems.

The first major result in the application of linear programming to combinatorial min-max theorems is due to Hoffman and Kruskal [56], who, in 1956, proved that if A is a totally unimodular matrix (i.e. the determinant of every square submatrix of A is either 0,1 or -1) then, for all integer vectors b and w such that the optima exist, both sides of the equality

(1.3)
$$\max\{wx: Ax \le b\} = \min\{yb: yA = w, y \ge 0\}$$

can be achieved by integer vectors. The König-Egerváry theorem is a direct consequence of this result (see Hoffman [60] for other applications). During the late fifties and sixties, Ford and Fulkerson [62], Edmonds [65, 70], and others used the Hoffman-Kruskal result and other linear programming ideas to prove combinatorial min-max theorems where the coefficient matrix of the corresponding linear system was not necessarily totally unimodular. In each case, the min-max result was obtained by showing that for a class of linear systems each side of (1.3) can be achieved by integer vectors. Hoffman [74] was the first to see that, in fact, to prove min-max theorems it suffices to show that the minimum in (1.3) always has an integer optimal solution. Indeed, it follows from a result of Hoffman [74,82] and Edmonds and Giles [77] (see Chapter 2, Section 2), that if $Ax \leq b$ is a linear system with b integral and if the minimum in (1.3) can be achieved by an integer vector for each integer vector w for which the optima exist, then the maximum in (1.3) can also be achieved by an integer vector for each integer vector w for which the optima exist. This result plays an important role in a technique

established by Hoffman [74], Younger [69], Lovász [76], Robertson (see Lovász [76]), Johnson [75], Edmonds and Giles [77], and Hoffman and Schwartz [78] for proving combinatorial min-max theorems (see Chapter 2, Section 10). This technique has been used quite successfully in recent years in obtaining very general min-max results by Edmonds and Giles [77], Hoffman and Schwartz [78], Frank [79], Schrijver [82a], and others. These general results include as special cases such well known theorems as the König-Egerváry theorem, Menger's theorem, the max-flow min-cut theorem, Dilworth's theorem, Fulkerson's branching theorem, the Vidyasankar-Younger theorem on longest paths, and the Lucchesi-Younger theorem on directed cuts.

Motivated by the above results, a rational linear system $Ax \le b$ is defined to be a <u>totally dual integral</u> system if the minimum in (1.3) can be achieved by an integer vector for each integer vector w for which the optima exist (a linear system $\{A_1x \le b_1, A_2x \ge b_2, A_3x = b_3\}$ is called totally dual integral if $\{A_1x \le b_1, -A_2x \le -b_2, A_3x \le b_3, -A_3x \le -b_3\}$ is totally dual integral). Thus, to prove a combinatorial min-max theorem it often suffices to show that a certain linear system is totally dual integral.

This thesis deals with general properties of totally dual integral systems and with totally dual integral systems related to matching problems. A summary of the results contained in the thesis is given in Section 3, following a section on notation and background material.

Background material.

In this section a survey of some well known results, which will be used throughout the thesis, is presented, together with notational conventions and references to background material.

Throughout the thesis, linear systems and linear spaces are assumed to be rational unless otherwise specified. Rational spaces suffice for the applications mentioned in the thesis. Moreover, two of the key results in the theory of total dual integrality (the theorem of Hoffman and Edmonds-Giles and Hilbert's Finite Basis Theorem) do not hold for real linear systems (see Giles and Pulleyblank [79] and Mandel [81]). The n-dimensional rational space will be denoted by \mathbb{Q}^n .

Let $a=(a_1,\ldots,a_n)$ and $b=(b_1,\ldots,b_n)$ be vectors. The expression " $a\leq b$ " means $a_i\leq b_i$ for $i=1,\ldots,n$ and "ab" denotes the inner product $\Sigma\{a_ib_i\colon i=1,\ldots,n\}$. When matrix equations and inequalities are expressed it is assumed that the sizes of the matrices and vectors involved are compatible for matrix multiplication. The expression "0" will often be used to denote the vector with the value of zero in each component. Similarly, "1" will denote the vector of all ones.

a) Linear programming

A <u>linear programming problem</u> (or a linear program) is the problem of maximizing or minimizing a linear (objective) function subject to a set of linear inequality constraints. For a history of linear programming see Dantzig [63].

If $Ax \leq b$ is a system of linear inequalities and w is a vector, the expression $\max\{wx\colon Ax \leq b\}$ is used to denote both the problem of maximizing wx subject to $Ax \leq b$ and the number wx^* , where x^* is a vector such that $Ax^* \leq b$ and $wx^* \geq wx$ for all vectors x for which $Ax \leq b$, if such a vector x^* exists (x^* is called an optimal solution to the linear program). A basic result in linear programming (see Dantzig [63]) is that if $P = \{x\colon Ax \leq b\}$ is nonempty and there exists a number α such that $wx \leq \alpha$ for all $x \in P$, then the number $\max\{wx\colon Ax \leq b\}$ exists (i.e. there exists an optimal solution to the linear program).

With each linear program $\max\{wx: Ax \le b\}$ there is an associated dual linear program $\min\{yb: yA = w, y \ge 0\}$. J. von Neumann (see Dantzig [63]) and Gale, Kuhn, and Tucker [51] proved the following fundamental result, known as the duality theorem.

Theorem 2.1: Let $Ax \le b$ be a linear system and w a vector. If $\max\{wx \colon Ax \le b\}$ exists, then $\min\{yb \colon yA = w, y \ge 0\}$ exists and

(2.1)
$$\max\{wx: Ax \le b\} = \min\{yb: yA = w, y \ge 0\}.$$

The duality theorem implies similar results for linear programs in other forms. For instance, if $\max\{wx: Ax \le b, x \ge 0\}$ exists then

(2.2)
$$\max\{wx: Ax \le b, x \ge 0\} = \min\{yb: yA \ge w, y \ge 0\}$$

holds. It should be noted that a linear system $\{Ax \le b, x \ge 0\}$ is totally dual integral if and only if $\min\{yb: yA \ge w, y \ge 0\}$ can be achieved by an integer vector for each integral w for which it exists.

The following result, known as the complementary slackness theorem, will be used in Chapter 2.

Theorem 2.2: Let $Ax \le b$ be a linear system of m inequalities and let w be a vector. Suppose that x^* is a member of $\{x: Ax \le b\}$ and that y^* is a member of $\{y: yA = w, y \ge 0\}$. The vector x^* achieves the maximum in (2.1) and the vector y^* the minimum in (2.1) if and only if for all $i = 1, \ldots, m, y_i^* > 0$ implies that $a_i x^* = b_i$, where $a_i x \le b_i$ is the i'th inequality in the system $Ax \le b$.

For a proof of this theorem and for additional material on linear programming see Dantzig [63].

b) Polyhedra

A polyhedron is a subset of \mathbb{Q}^n of the form $\{x: Ax \leq b\}$, where A is an m×n matrix and b is a vector. For an account of the theory of polyhedra see Rockafellar [79] and Stoer and Witzgall [70], and the papers Bachem and Grotschel [82] and Pulleyblank [82].

If P is a polyhedron, then a linear system $Ax \le b$ such that $P = \{x: Ax \le b\}$ is a <u>defining system</u> for P. A polyhedron P is <u>bounded</u> if there exist vectors ℓ and u such that $\ell \le x \le u$ for all x in P. A <u>hyperplane</u> is a polyhedron of the form $\{x: \alpha x = \beta\}$, where α is a vector and β is a scalar. The intersection of finitely many hyperplanes is an <u>affine subspace</u>, i.e. a polyhedron of the form $\{x: Ax = b\}$, where Ax = b is a system of equations.

A finite set of vectors M is an <u>affinely independent</u> set if the only vector $\lambda = (\lambda_m \colon m \in M)$ such that $\Sigma\{\lambda_m \colon m \in M\} = 0$ and $\Sigma\{\lambda_m \colon m \in M\} = 0$ is the zero vector. The <u>dimension</u> of a polyhedron P (denoted by dim P) is one less than the cardinality of a maximal set of

affinely independent vectors in P. A polyhedron $P \subseteq \mathbb{Q}^n$ is of <u>full</u> dimension if dim P = n.

Let P be a polyhedron defined by a linear system $Ax \leq b$. A linear inequality $\alpha x \leq \beta$ is valid for P if each $x \in P$ satisfies $\alpha \bar{x} \leq \beta$. A hyperplane $H = \{x: \alpha x = \beta\}$ is a supporting hyperplane of P if $H \cap P \neq \emptyset$ and $\alpha x \leq \beta$ is a valid inequality for P. A face of P is a subset which is the intersection of P with a supporting hyperplane of P. Equivalently, a face of P is a nonempty set of the form $\{x \in P: A'x = b'\}$, where $A'x \le b'$ is a subset of the inequalities $Ax \le b$. A face, F, of P is called a facet of P if dim F = dim P-1. A valid inequality $\alpha x \leq \beta$ is facet-inducing for P if $\{x: \alpha x = \beta\} \cap P$ is a facet of P. An inequality $\alpha x \leq \beta$ is essential for P if every defining system of P must include some positive scalar multiple of $\alpha x \leq \beta$. If P is of full dimension, then there exists a minimal defining system for P that is unique up to positive scalar multiples. Moreover, if P is of full dimension then there exists a one to one correspondence between the facets of P and the essential inequalities of P (each essential inequality induces a different facet of P and each facetinducing inequality is an essential inequality for P).

Each minimal face of P (minimal with respect to inclusion) is of the form $\{x\colon A'x=b'\}$, for some subsystem $A'x\le b'$ of $Ax\le b$ (thus, each minimal face is an affine subspace). A face of P consisting of exactly one point is called a vertex.

The <u>convex hull</u> of a finite set of vectors, M, is the intersection of all convex sets containing M. A well known result is that the convex hull of a finite set of vectors is a polyhedron.

If w is a vector such that $\max\{wx: x \in P\}$ exists, then the

maximum is achieved by each member of some face of P. The polyhedron P is an integer polyhedron if $\max\{wx\colon x\in P\}$ can be achieved by an integer vector for each vector w for which the maximum exists. The following well known result follows immediately from this definition.

<u>Lemma 2.3:</u> A polyhedron P is an integer polyhedron if and only if each minimal face of P contains an integer point.

A polyhedral cone is a polyhedron of the form $\{x: Ax \geq 0\}$, where A is a matrix. By a theorem of Weyl [35] and Minkowski [96], a set C is a polyhedral cone if and only if there exists a finite set of vectors $\{a_1,\ldots,a_k\}$ that generate C, i.e. $C=\{\lambda_1a_1+\ldots+\lambda_ka_k: \lambda_i\geq 0, i=1,\ldots,k\}$. A polyhedral cone C is pointed if $x\in C$, $x\neq 0$ implies that $-x\notin C$. If C is a pointed polyhedral cone then there exists a vector w such that wx>0 for each nonzero $x\in C$.

c) Complexity

Results in complexity theory and formal definitions of polynomial-time algorithm, NP, and co-NP can be found in S. Cook [77], Garey and Johnson [79], and Karp [72,75] (see also Edmonds [65b]). Informally, a class of objects is in the class NP if there is some information which can be supplied about an object in the class so that one can verify in an amount of time polynomially bounded by the size of the object that the object is in the class. (The size of the information must be bounded by a polynomial in the size of the object.) Similarly, a class of objects is in the class co-NP if there is some information which can be supplied about an object not in the class so that one can verify in polynomial time that the object is not in the class. A polynomial-time algorithm is one which has a running time which is bounded by some polynomial in the size

of the input.

Most of the complexity results given in this thesis involve linear systems. The size of a linear system $Ax \le b$ is the number of binary digits needed to write the system in binary notation.

Two important polynomial-time algorithms, which will be used often in Chapter 2, are the ellipsoid algorithm for linear programming and Lenstra's integer programming algorithm.

Khachian [79] showed that it is possible to solve linear programming problems in polynomial time using the ellipsoid method of Shor [70] and Yudin and Nemirowskii [76] (see Akgül [81] and Bland, Goldfarb, and Todd [81] for surveys of results on the ellipsoid method). Grötschel, Lovász, and Schrijver [81,82] later showed that a modification of this algorithm can be used to solve many combinatorial optimization problems in polynomial time.

H.W. Lenstra [81] proved that for any fixed positive integer r, there exists a polynomial-time algorithm which, for any linear system $Ax \le b$ with A a matrix of rank r, finds an integer point in $\{x: Ax \le b\}$ or proves that no such point exists. See Grötschel, Lovász, and Schrijver [82] and Lenstra, Lenstra and Lovász [82] for results related to this algorithm of Lenstra.

d) Smith and Hermite normal forms

Let A be an $m \times n$ integer matrix of rank r. Hermite [51] showed that there exists a unimodular U (an $n \times n$ integer matrix is called unimodular if it has determinant 1 or -1) such that

$$(2.3) \qquad AU = \begin{bmatrix} L & 0 \\ B & 0 \end{bmatrix}$$

where L is an r×r lower triangular matrix, B is an (m-r)xr matrix, and the 0's are matrices of all zeros (the matrix AU is said to be in Hermite normal form). Smith [61] proved that there exist unimodular matrices V and U such that

$$(2.4) \qquad \text{VAU} = \begin{bmatrix} D & 0 \\ 0 & 0 \end{bmatrix}$$

where D is an $r \times r$ diagonal matrix and the O's are matrices of all zeros (VAU is in <u>Smith normal form</u>).

There are many applications of Smith and Hermite normal forms (see, for example, Bachem and Kannan [79]). One application that will be used in Chapter 2 is the following result, which appears as an exercise in yan der Waerden [40](cf. Bachem and von Randow [79]).

Theorem 2.4: Let Ax = b be a system of linear equations. There exists an integer vector \overline{x} such that $A\overline{x} = b$ if and only if there does not exist a vector y such that yA is integral and yb is nonintegral. \Box

This theorem follows directly from the result of Smith [61] mentioned above. It will be referred to as the Smith-van der Waerden theorem.

Polynomial-time algorithms for computing the Hermite and Smith normal forms of an integer matrix were found by Kannan and Bachem [79]. These algorithms will be used at several points in Chapter 2.

Let a_1, \ldots, a_k be vectors. The set $L = \{\lambda_1 a_1 + \ldots + \lambda_k a_k : \lambda_i \text{ is} \}$ an integer for $i = 1, \ldots, k\}$ is the <u>lattice</u> generated by a_1, \ldots, a_k .

A <u>basis</u> for a lattice L is a set of linearly independent vectors that generate L. It follows from the result of Hermite [61] that every lattice has a basis. For results on lattices see Lekkerkerker [69].

e) Graph theory

A graph G is an ordered triple (VG,EG, Ψ_G) consisting of a nonempty finite set of <u>nodes</u>, a finite set EG of <u>edges</u>, disjoint from VG, and an incidence function Ψ_G that associates with each edge an unordered pair of distinct nodes of G (loops are not permitted). Often, an edge e such that $\Psi_G(e) = (u,v)$ will be identified by the unordered pair (u,v). For an edge e with $\Psi_G(e) = (u,v)$, u and v are called the ends of e. For graph theory terminology see Bondy and Murty [76]. Special notation, which will be used in Chapter 3 is given below.

Let G be a graph. For each $v \in VG$, $\delta_G(v)$ denotes the subset of edges of G which meet v (i.e. have v as an end node) and $N_G(v)$ denotes the subset of nodes $u \in VG \setminus \{v\}$ for which there exists an edge $(v,u) \in EG$. For each $S \subseteq VG$, $\gamma_G(S)$ denotes the subset of edges of G having both ends in S. If it is clear which graph is being considered, the "G" subscripts on Ψ_G, δ_G, N_G , and γ_G will not be used.

For each $S\subseteq VG$, G[S] denotes the subgraph of G induced by S, i.e. $G[S]=(S,\gamma(S),\overline{\Psi})$, where $\overline{\Psi}$ is the restriction of Ψ to $\gamma(S)$.

If S is a set of subsets of VG and e is an edge of G, then S(e) denotes the set $\{S \in S : e \in \gamma(S)\}$.

If \mathcal{H} is a set of subgraphs of G and e is an edge of G, then $\mathcal{H}(e)$ denotes the set $\{\mathcal{H} \in \mathcal{H} \colon e \in E\mathcal{H}\}$.

If $x = (x_i : i \in I)$ and $S \subseteq I$, where I is a finite set, then x(S) denotes the sum $\Sigma\{x_i : i \in S\}$.

If β is a number, then $\lfloor \beta \rfloor$ denotes the largest integer less than or equal to β and $\lfloor \beta \rfloor$ denotes the smallest integer greater than or equal to β . If $\mathbf{x} = (\mathbf{x}_1, \dots, \mathbf{x}_n)$ is a vector then $\lfloor \mathbf{x} \rfloor$ denotes the vector $(\lfloor \mathbf{x}_1 \rfloor, \dots, \lfloor \mathbf{x}_n \rfloor)$ and $\lfloor \mathbf{x} \rfloor$ denotes the vector $(\lfloor \mathbf{x}_1 \rfloor, \dots, \lfloor \mathbf{x}_n \rfloor)$.

3. Summary of results.

This thesis deals with general properties of totally dual integral systems and with totally dual integral systems related to matching problems.

The theory of totally dual integral systems is developed in Chapter 2. The presentation is meant to be as complete as possible. Hence, in sections 2 and 5, proofs of key results due to Hoffman [74,82] and Edmonds and Giles [77], Giles and Pulleyblank [79], and Schrijver [81] are presented. The proofs of these results do not differ substantially from those given in the above papers and in Schrijver [80], but it is hoped that having the material collected here with standard terminology will be of benefit to the reader. Also, it is pointed out in two places how the results may be generalized from the lattice of all integer points to arbitrary lattices.

To apply results of total dual integrality to combinatorial problems it is necessary to find totally dual integral defining systems for classes of polyhedra. In the analogous situation for integer polyhedra certain operations which may be applied to a linear system, such that if the linear system defines an integer polyhedron then the derived system does as well, have been used often. Section 6 of Chapter 2 investigates whether or not these operations also preserve total dual integrality of linear systems. Among the results in that section, it is shown that adding slack variables to linear systems with integer left hand sides, setting some inequalities to equalities (to obtain a defining system for a face of a polyhedron), and applying Fourier-Motzkin elimination to eliminate a variable with 0 and ±1 coefficients from a linear system (to obtain a defining system for a projection of the polyhedron) all preserve total dual integrality. (The fact that setting inequalities to equalities preserves total dual integrality is also given in Schrijver [82b].) It is also shown that the only

nontrivial scalar multiplication of inequalities that preserves total dual integrality is multiplication by scalars of the form 1/k for some positive integer k.

Suppose that $Ax \le b$ is a totally dual integral system with b integral and that w is an integer vector such that $max\{wx \colon Ax \le b\}$ exists. In Section 7 of Chapter 2 the problem of finding integer vectors x and y that achieve the optima in the equation

(3.1)
$$\max\{wx: Ax \le b\} = \min\{yb: yA = w, y \ge 0\}$$

is considered. It is shown that there exists a polynomial-time algorithm which, for any linear system $Mx \le d$ and any vector c for which $max\{cx\colon Mx \le d\}$ exists, either finds an integer vector which achieves the maximum or shows that $Mx \le d$ does not define an integer polyhedron. This easy result implies that an integer vector x which achieves the maximum in (3.1) can be found in polynomial time. A result of Chandrasekaran [81] is presented which shows that for integral A an integral y which achieves the minimum in (3.1) can also be found in polynomial time

Totally dual integral systems associated with combinatorial problems are often very large with respect to the size of the combinatorial objects. Thus, the above integer programming results are often not sufficient to find, in polynomial time, combinatorial objects which achieve the maximum and minimum in a min-max equality. Grötschel, Lovász, and Schrijver [81,82] have shown in various cases that the ellipsoid algorithm can be used to overcome the difficulty of having large linear systems associated with combinatorial problems. However, unlike the case of finding integer optimal solutions to primal linear programs whese solution set is a bounded polyhedron, they were not able to derive under general conditions a polynomial-time algorithm for finding integer optimal dual solutions.

One difficulty in finding such an algorithm is the problem of expressing the integer optimal dual solution. In general, there may be exponentially many dual variables associated with a linear system for a combinatorial problem. Thus, there may be integer optimal solutions to the dual linear program that cannot be described in polynomial time. A result of Cook, Fonlupt, and Schrijver [83] presented in Section 8 resolves this difficulty for totally dual integral systems. This result is that if $Ax \le b$ is a totally dual integral system, with A an integer matrix of rank r, such that $P = \{x: Ax \le b\}$ is of full dimension, then for each integer vector w for which $min\{yb: yA = w, y \ge 0\}$ exists the minimum can be achieved by an integer vector with at most 2r-1 nonzero components (it is also shown that if P is not of full dimension then there is no upper bound on the number of nonzero dual variables needed in an integer optimal solution to $min\{yb: yA = W, y \ge 0\}$ in terms of r). This result improves several known results when specialized to particular combinatorial problems.

A basic problem in the study of any type of mathematical object is that of recognizing the object. Two easy results on recognizing linear systems that define integer polyhedra are presented in Section 9 of Chapter 2. It is shown that the class of linear systems that define integer polyhedra is in co-NP and that for any fixed positive integer r, there exists a polynomial-time algorithm which, for any linear system $Ax \le b$ with A a matrix of rank r, determines whether or not $Ax \le b$ defines an integer polyhedron. Cook, Lovász, and Schrijver [82] showed that the class of linear systems $Ax \le b$ with A integral that are totally dual integral is in co-NP and that for any fixed positive integer r, there exists a polynomial-time algorithm which, for any linear system $Ax \le b$ with A an integer matrix of rank r, determines whether or not $Ax \le b$ is totally dual integral. These two results

are also presented in Section 9.

Schrijver [81] showed that each polyhedron of full dimension is defined by a unique minimal totally dual integral system with integer left hand sides. A result given in Section 9 of Chapter 2 is that for any fixed positive integer r, there exists a polynomial-time algorithm which, for any linear system $Ax \le b$ with A a matrix of rank r which defines a polyhedron of full dimension and any inequality $\alpha x \le \beta$ determines whether or not $\alpha x \le \beta$ is in the unique minimal totally dual integral defining system with integer left hand sides for P.

A well known "uncrossing" technique for proving that combinatorially described linear systems are totally dual integral is presented in Section 10.

The results on linear systems for matching problems center around the objects known as simple b-matchings. If G is a graph and $b = (b_v: v \in VG)$ is a nonnegative integer vector, a simple b-matching of G is a subset of the edges of G such that each node v of G meets at most b_v edges in the subset. A simple b-matching of G which meets each node v in exactly b_v edges is often called a b-factor of G (b-factors have been studied by Tutte[47,52], Gallai[50], Lovász[70] and others). A description of a totally dual integral defining system for the convex hull of (the incidence vectors of) the simple b-matchings of a graph follows easily from a result of Edmonds and Johnson [70]. In Chapter 3, the unique minimal totally dual integral defining system with integer left hand sides for this convex hull is characterized. This result yields a "best possible" min-max theorem for simple b-matchings. A similar result for the slightly more general "capacitated" b-matching problem is also given. This general result is then used to prove a

theorem which characterizes the unique minimal totally dual integral defining system for the convex hull of the matchings of a graph due to Cunningham and Marsh [78], and to prove a theorem which characterizes the unique minimal totally dual integral defining system for the convex hull of the b-matchings of a graph due to Cook [81] and Pulleyblank [81]. The proof technique used is simpler than those used by Cook [81] and Pulleyblank [81] in that it does not require the knowledge of a complete list of the facet-inducing inequalities for the polyhedron in question. An essential result used throughout Chapter 3 is a connection between minimal totally dual integral systems and separability. This simple result is presented in Section 2 of that chapter and is used to prove a result of R. Giles on matching separability, a result of Edmonds and Giles (see Pulleyblank [82]) on matroid polyhedra, a result of Giles [75] on matroid intersection polyhedra, and a result on b-matching separability.

In the final section of Chapter 3, the unique minimal totally dual integral system with integer left hand sides that defines the convex hull of the triangle-free 2-matchings of a graph is characterized.

CHAPTER II

TOTALLY DUAL INTEGRAL SYSTEMS

1. <u>Introduction</u>

Properties of totally dual integral systems are presented in this chapter. For completeness, the presentation includes proofs of key results due to Hoffman [74,82], Edmonds and Giles [77], Giles and Pulleybank [79], and Schrijver [81].

The relationship of totally dual integral systems and integer polyhedra is presented in Sections 2 and 5. This relationship provides motivation for the study of totally dual integral systems. motivation is provided in Sections 3 and 4, which contain some well known results on classes of totally dual integral systems. Results of Hoffman and Kruskal [56] and Fulkerson, Hoffman, and Oppenheim [74] on totally unimodular and balanced matrices are presented in Section 3, without proof. Fulkerson's anti-blocking theory is presented in Section 4, including proofs of most of the results (Fulkerson developed this theory before general totally dual integral systems were studied, so his papers do not use the terminology of total dual integrality nor do they make use of the results on general totally dual integral systems). Section 4 includes a generalization of a theorem of Fulkerson [71,72] to h-perfect graphs. Section 6 deals with the problem of determining what operations can be performed on a totally dual integral system such that the derived system is also totally dual integral. Integer programming problems related to totally dual integral systems are studied in Section 7. Section 8 contains results on the number of nonzero dual variables needed in an integer optimal solution to the dual linear program of a linear program with a totally dual integral constraint set. Section 9 deals with the problem of recognizing a

totally dual integral system. In Section 10, a well known "uncrossing" technique for proving that combinatorially described linear systems are totally dual integral is presented. This technique is illustrated with several examples which make use of results on totally unimodular and balanced matrices.

A property of linear systems that is stronger than total dual integrality and which will not be dealt with here is box total dual integrality. A linear system $Ax \le b$ is called box totally dual integral if for all vectors c and d, the linear system $\{Ax \le b, c \le x \le d\}$ is totally dual integral. For results on box total dual integrality see Edmonds and Giles [77].

Another related concept that will not be dealt with here, except for an application to matroid theory given in Section 8, is that of integer rounding. A linear system is said to have the <u>integer rounding</u> property (Baum and Trotter [77], Chandrasekaran [81]) if

(1.1)
$$\min\{yb: yA = w, y \ge 0, y \text{ integral}\} = [\min\{yb: yA = w, y \ge 0\}]$$

for each integer vector w for which the right hand side exists. One reason for not dealing with integer rounding is that Giles and Orlin [81] have shown that a linear system has the integer rounding property if and if only if a corresponding system is totally dual integral. Indeed, it follows from the definitions of total dual integrality and integer rounding that $Ax \leq b$ has the integer rounding property if and only if $\{Ax-bx_0 \leq 0, x_0 \leq 0\}$ is totally dual integral. Thus, results on recognition, integer programming, etc., for integer rounding follow from the corresponding results for total dual integrality. It should be noted that Giles and Orlin [81] also proved a converse result: Given a linear system $Ax \leq b$, there exists a positive integer M such that $Ax \leq b$ is totally dual integral if and only if $Ax \leq Mb$ has the integer rounding property.

2. Integral Polyhedra

Motivation for the study of total dual integrality is provided by the relation of totally dual integral systems to integer polyhedra. This relationship is presented in this section and in Section 5. A good reference for the material presented here is Schrijver [80a].

The starting point for the study of integer polyhedra and total dual integrality is the Smith-van der Waerden Theorem (Theorem 2.4, Chapter 1). This theorem implies the following geometric result (cf. Schrijver [80]).

Theorem 2.1: Let K be an affine subspace of \mathbb{Q}^n that does not contain integer points. Then K lies in a hyperplane that contains no integer points.

<u>Proof:</u> Since K is an affine subspace, $K = \{x: Ax = b\}$ for some integral A and b. By the Smith-van der Waerden Theorem, there exists a rational vector α such that $\alpha Ax = \alpha b$ has no integral solutions. Since K is contained in $\{x: \alpha Ax = \alpha b\}$, the result follows.

It is quite easy to extend this result from the lattice of all integer points to arbitrary lattices.

Corollary 2.2: Let L be a lattice in \mathbb{Q}^n and let K be an affine subspace of \mathbb{Q}^n . If K does not contain any points in L, then K lies in a hyperplane that contains no points in L.

<u>Proof:</u> Suppose that the dimension of L is n. Let $\{a_1,\dots,a_n\}$ be a basis for L. Consider the linear transformation T given by the matrix with columns a_1,\dots,a_n . A point $x\in \mathbb{Q}^n$ is an integer point if and only if T(x) is a point in L. The pre-image of K, $T^{-1}(K)$,

is an affine subspace of \mathbb{Q}^n which does not contain any integer points. By Theorem 2.1, $\mathbf{T}^{-1}(\mathbb{K})$ lies in a hyperplane H that contains no integer points. So K lies in the hyperplane $\mathbf{T}(\mathbf{H})$ and $\mathbf{T}(\mathbf{H})$ contains no points in L.

Now suppose that the dimension of L is d and d < n. Let $\{a_1,\dots,a_d\}$ be a basis for L and let M be the linear space spanned by a_1,\dots,a_d . By the above argument, the affine subspace KnM lies in an affine subspace H of dimension d-1 that contains no points in L. The subspace H can be extended to a hyperplane that contains K but contains no points in L.

The Smith-van der Waerden Theorem will be used to obtain the following fundamental theorem of Hoffman [74,82] and Edmonds and Giles [77] (this particular form of the theorem is due to Schrijver [80]).

Theorem 2.3: A polyhedron P is an integer polyhedron if and only if every supporting hyperplane of P contains integer points.

<u>Proof.</u> Suppose that P is an integer polyhedron. The intersection of P with a supporting hyperplane is a face and every face of P contains integer points, so every supporting hyperplane of P contains integer points. Conversely, suppose that every supporting hyperplane of P contains integer points. Suppose also that P contains a minimal face F that contains no integer points. Since F is an affine subspace, by Theorem 2.1 F lies in a hyperplane H that contains no integer points. Let $P = \{x: Ax \le b\}$, where A and b are integral. By adding a large positive integer to each component of the vector α in the proof of Theorem 2.1, it can be assumed that H is of the form

This theorem is a special case of a result of Schrijver [80], which gives a geometric description of the work of Chvátal [75] related to the cutting plane algorithm of Gomory[63].

As in Corollary 2.2, Theorem 2.3 can be used to show that if L is a lattice of dimension n in \mathbb{Q}^n and P is a polyhedron in \mathbb{Q}^n , then every face of P contains points in L if and only if every supporting hyperplane contains points in L. As examples, L can be taken to be the set of all half integer points in \mathbb{Q}^n or the set of all even integer points in \mathbb{Q}^n .

Theorem 2.3 has several equivalent forms which are given in the following theorems.

Theorem 2.4: A polyhedron P is an integer polyhedron if and only if $\max\{wx\colon x\in P\}$ is an integer for each integral w for which the maximum exists.

<u>Proof:</u> If P is an integer polyhedron then $\max\{wx\colon x\in P\}$ is achieved by an integral x for each w for which the maximum exists so if w is integral then $\max\{wx\colon x\in P\}$ is an integer. Now suppose that $\max\{wx\colon x\in P\}$ is an integer for each integral w for which the maximum exists. Let H be a supporting hyperplane of P and let $H = \{x\colon \alpha x = \beta\}$ for some integral α . Now $\max\{\alpha x\colon x\in P\} = \beta$, so β is an integer. But since it can be assumed that the components of are integers with greatest common divisor 1, β can be expressed

as an integer combination of the components of α . So H contains an integer point. By Theorem 2.3, P is an integer polyhedron.

Theorem 2.5: If $Ax \le b$ is a totally dual integral system and b is integral, then $P = \{x: Ax \le b\}$ is an integer polyhedron.

Proof: This follows from the duality theorem and Theorem 2.4.

Theorem 2.6: If $Ax \le b$ is a totally dual integral system and b is integral, then each side of the equation

 $\max\{wx: Ax \le b\} = \min\{yb: yA = w, y \ge 0\}$

can be achieved by an integer vector for each integral w for which the optima exist.

<u>Proof:</u> This is just an interpretation of Theorem 2.5.

3. Totally unimodular and balanced matrices.

It is not clear from the definition of total dual integrality that nontrivial totally dual integral systems exist. Therefore, in this section, some familiar classes of totally dual integral systems will be studied. Further examples will be considered in Section 10.

Totally unimodular and balanced matrices are two important classes of matrices that give rise to totally dual integral systems. These classes can be used to build classes of totally dual integral systems which include systems arising from many different combinatorial problems — see Section 10.

A matrix A is totally unimodular if the determinant of every square submatrix of A is either 0,1, or -1. Hoffman and Kruskal [56] showed the importance of total unimodularity in the study of integer polyhedra by proving the following theorem.

Theorem 3.1: An integer matrix A is totally unimodular if and only if for every integral b the polyhedron $\{x: Ax \le b, x \ge 0\}$ is an integer polyhedron.

Thus, if A is totally unimodular then $Ax \leq b$ is totally dual integral for every vector b.

A short proof of this theorem is given in Veinott and Dantzig [68], where the following result is also proven.

Theorem 3.2: Let A be an $m^{X}n$ integer matrix of rank m. The polyhedron $\{x: Ax = b, x \ge 0\}$ is an integer polyhedron for every integral b if and only if each $m^{X}m$ submatrix of A has determinant 0,1, or -1.

The following corollary is a restatement of Theorem 3.2.

Corollary 3.3: Let A be an m^Xn matrix of rank m. The linear system $Ax \le b$ is totally dual integral for every integral b if and only if each m^Xm submatrix of A has determinant 0,1, or -1.

<u>Proof:</u> The system $Ax \le b$ is totally dual integral for every integral b if and only if for every integral w the polyhedron $\{y: yA = w, y \ge 0\}$ is an integer polyhedron.

For applications of total unimodularity see the papers of Baranyai [73], Hoffman [60,76,75], and Lovász [79].

It does not follow directly from the definition that there exists a method to determine whether or not a matrix is totally unimodular in time polynomial in the size of the matrix - there are too many square submatrices. However, a polynomial time recognition algorithm for totally unimodular matrices does follow from Seymour's deep characterization of regular matroids - see Seymour [80].

Much work has been done on finding simple sufficient conditions for total unimodularity as well as finding characterizations of total unimodularity - see Padberg [75] and Truemper [78] for surveys. A sufficient condition for total unimodularity due to Hoffman and Kruskal will be used in Chapter 3. The following well known result is due to Heller and Tompkins [56] (see also Egervary [31]).

Lemma 3.4: If A is a 0-1 matrix whose rows can be partitioned into two sets A_1 and A_2 such that each column has at most one nonzero component in each part of the partition, then A is totally unimodular.

This result will be used to prove the following theorem of Hoffman and Kruskal [56].

Theorem 3.5: Let E be a finite set and F_1, F_2 families of subsets of E. For i=1,2, suppose that if $C,D \in F_i$ then either $C \cap D = \emptyset$, $C \subseteq D$, or $D \subseteq C$. The incidence matrix A of the family $F_1 \cup F_2$ is totally unimodular.

<u>Proof:</u> Let M be a square submatrix of A and let A_i be the rows of A corresponding to sets in F_i (i=1,2). Let D be a minimal row of A_i (i=1 or 2). Subtract D from each other row in A_i that is greater than D. Repeating this procedure, a matrix which statisfies the conditions in Lemma 3.4 is obtained. So the matrix obtained has determinant 0,1, or -1, which implies that M has determinant 0,1, or -1.

Total unimodularity has been generalized to the notion of local unimodularity by Hoffman and Oppenheim [78]. For results on local unimodularity see Hoffman [79] and Truemper and Chandrasekaran [78].

By Lemma 3.4, the edge-node incidence matrix of a bipartite graph is totally unimodular. Berge [72] introduced balanced matrices as a generalization of these edge-node incidence matrices. A 0-1 matrix A is balanced if A contains no square submatrix of odd size which has exactly two ones in every column and exactly two ones in every row. Since a bipartite graph contains no odd cycles, it is clear that the edge-node incidence matrix of a bipartite graph is balanced. In fact, since the determinant of any minimal (with respect to taking submatrices) 0-1 matrix of odd size with exactly two ones in every row and column is two, every 0-1 totally unimodular matrix is balanced. The following example of Berge [72] shows that not all balanced matrices are totally unimodular.

$$\begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & 1 & 0 & 0 \\ 1 & 0 & 1 & 0 \\ 1 & 0 & 0 & 1 \end{bmatrix}$$

Many nice min-max theorems for bipartite graphs generalize to the context of balanced matrices - see Berge [72,73,80], Lovász [72], and the survey in Schrijver [79]. The importance of balanced matrices to total dual integrality lies in the following theorem of Fulkerson, Hoffman and Oppenheim [74].

Theorem 3.6: If A is a balanced matrix, then $Ax \le 1$, $x \ge 0$ is totally dual integral and $Ax \ge 1$, $x \ge 0$ is totally dual integral.

Berge [72,80], Berge and Las Vergnas [70], Lovász [72], and others have given characterizations of balanced matrices, but, as yet, there is no known way to test whether a given matrix is balanced or not in polynomial time. Polynomial time recognition algorithms have been found for a class of balanced matrices that generalize the edge-node incidences matrices of trees (the so called totally balanced matrices), by Anstee and Farber [82], Hoffman, Kolin, and Sakarovitch [82] and Lubiw [82].

4. Anti-blocking and the perfect graph theorem.

The theory of blocking and anti-blocking polyhedra was developed by Fulkerson [68,70,71,72] and Lehman [79]. A brief account of the anti-blocking portion of this theory is presented in this section, since it contains several interesting results concerning total dual integrality which provide further motivation for the study of totally dual integral systems. These results include Fulkerson's pluperfect graph theorem and the perfect graph theorem of Lovász. The theory of blocking polyhedra will not be studied here, but several differences in this theory and anti-blocking theory will be pointed out.

Let M be a nonnegative matrix with no column of all zeros and let P be the polyhedron $\{x \colon Mx \le 1, x \ge 0\}$. Define the <u>anti-blocking polyhedron</u> bedron of P to be the polyhedron

$$A(P) = \{y: y \ge 0 \text{ and } yd \le 1 \text{ for each } d \in P\}.$$

It is possible to give a more compact description of A(P). Let a_1, \ldots, a_k be the vertices of P and let A be the matrix with rows a_1, \ldots, a_k . Since every point in P can be expressed as a convex linear combination of a_1, \ldots, a_k , A(p) = {y: Ay \leq 1, y \geq 0}. Since every point in P is nonnegative, A is nonnegative. Furthermore, if m_{ij} is the largest element in column j of the matrix M, then a vector with $1/m_{ij}$ in the jth position and zero in every other position is a vertex of P. So A has no column of all zeros and the expression A(A(P)) is meaningful. Fulkerson [72] gave the following result and proof.

Theorem 4.1: If M and P are defined as above, then A(A(P)) = P.

<u>Proof:</u> It follows immediately from the definition of A(P) that $P \subseteq A(A(P))$. Suppose that there exists $b \in A(A(P))$ such that $b \notin P$. Since $b \notin P$, there exists some row, m_i , of M such that $m_i b > 1$. But this is a contradiction, since $m_i \in A(P)$.

Theorem 4.1, together with the alternative description of A(P) given above, implies that if a row m_i is an essential row of M (that is, if $m_i x \le 1$ is a facet inducing inequality of P), then m_i is a vertex of A(P). It is not true that every vertex of A(P) must be a row of M. However, Fulkerson [72] has shown that every vertex of A(P) that is not an essential row of M can be obtained from an essential row of M by setting some components of the row to zero. Consequently, if every row of M is integral then every vertex of A(P) is integral.

To prove the above mentioned result of Fulkerson, first note that Farkas' Lemma implies that a row m_i of M is an essential row of M if and only if it is not less than or equal to a convex linear combination of the other rows of M. So all that needs to be shown is the following theorem of Fulkerson [72].

Theorem 4.2: Let M and P be defined as above and let d be a vertex of P. If d is less than or equal to a convex linear combination of $f^1, \ldots, f^s \in P$, then d can be obtained from some $f^i(1 \le i \le s)$ by setting some components of f^i to zero.

<u>Proof:</u> Suppose that d is less than or equal to $c = \Sigma\{\alpha_i f^i : i = 1, \ldots, s\}$, where $f^1, \ldots, f^s \in P$, $\Sigma\{\alpha_i : i = 1, \ldots, s\} = 1$, and $\alpha_1, \ldots, \alpha_s \ge 0$. The result is trivial if d = 0. So it may be assumed that d_1, \ldots, d_k are the positive components of d. Let $d' = (d_1, \ldots, d_k)$ and $c' = (c_1, \ldots, c_k)$.

Since d is a vertex of P, there exists a k×k submatrix E of M such that Ex' = 1 has d' as its unique solution. But $Ec' \le 1$ and $d' \le c'$. So Ed' = Ec' and d' = c'. Thus d can be expressed as a convex linear combination of $\overline{f}^1, \ldots, \overline{f}^s$, where \overline{f}^i is obtained from f^i by setting the components other than f^i_1, \ldots, f^i_k to zero. Since d is a vertex, d is equal to \overline{f}^i for some $i \in \{1, \ldots, s\}$ and the result follows.

The theory of anti-blocking polyhedra is connected to total dual integrality by the following theorem of Fulkerson [72], which is known as the "Pluperfect graph theorem." If M is a nonnegative matrix with no zero column, call a matrix A an <u>anti-blocking matrix</u> of M if $A(P) = \{x: Ax \le 1, x \le 0\}$, where $P = \{x: Mx \le 1, x \ge 0\}$.

Theorem 4.3: Let M be a 0-1 matrix having no zero column, let $P = \{x \colon Mx \le 1, \ x \ge 0\}$, and let A be an anti-blocking matrix of M. The polyhedron P is an integer polyhedron if and only if $Mx \le 1$, $x \ge 0$ is totally dual integral and $Ax \le 1$, $x \ge 0$ is totally dual integral.

<u>Proof:</u> By Theorem 2.5, if $Mx \le 1$, $x \ge 0$ is totally dual integral then P is an integer polyhedron (Fulkerson [72] gives a direct proof of this). Since M is integral, it follows from Theorem 4.1 and Theorem 4.2 that A(P) is an integer polyhedron. Suppose that P is an integer polyhedron. Let A' consist of the essential rows of A. If $A'x \le 1$, $x \ge 0$ is totally dual integral then $Ax \le 1$, $x \ge 0$ is also totally dual integral. Now since each vertex of P is 0-1 valued, A' is a 0-1 matrix. Thus, to prove the theorem it suffices to show that the fact that P is an integral polyhedron implies that

 $M_X \le 1$, $x \ge 0$ is totally dual integral. The following short proof of this is due to Baum and Trotter [82] and Korach [82].

It must be shown that for each nonnegative integral w, $\min\{1^{\bullet}y\colon yM\geq w,\;y\geq 0\}$ can be achieved by an integral vector. The proof is by induction on

$$z^* = \max\{wx: Mx \le 1, x \ge 0\} = \min\{1 \cdot y: yM \ge w, y \ge 0\}.$$

Clearly z* is nonnegative and integral. If $z^* = 0$, then the result is true since y = 0 attains the minimum in (4.1). Suppose that for all integral w such that $z^* \le k-1$ the result is true and that w is such that $z^* = k(k \ge 1)$. Since $z^* > 0$, there exists some row m_i of M such that $m_i x = 1$ for each vector x that achieves the maximum in (4.1) (by the complementary slackness theorem). Consider the linear programming duality equation

 $(4.2) \qquad v^* = \max\{(w-m_{\underline{i}})x \colon Mx \le 1, \ x \le 0\} = \min\{1 \cdot y \colon yM \ge w-m_{\underline{i}}, \ y \ge 0\}.$ By the choice of $m_{\underline{i}}$, $v^* = k-1$. So the minimum in (4.2) is achieved by an integral vector y'. Let $y''_{\underline{i}} = y'_{\underline{i}} + 1$ and $y''_{\underline{i}} = y'_{\underline{j}}$ for all other \underline{j} , where $y_{\underline{i}}$ is the variable corresponding to row $m_{\underline{i}}$. Since $1 \cdot y' = k-1$, $1 \cdot y'' = k$ and y'' achieves the minimum in (4.1).

A generalization of this theorem is given in Korach [82].

Lovász [72] proved the following result, which is closely related to Theorem 4.3.

Theorem 4.4: Let M be a 0-1 matrix with no zero column and let A

be an anti-blocking matrix of M. If $\min\{1 \cdot y : yM \ge w, y \ge 0\}$ is achieved by an integral vector for each 0-1 valued w, then $Mx \le 1$, $x \ge 0$ is totally dual integral and $Ax \le 1$, $x \ge 0$ is totally dual integral.

This result is the "Perfect graph theorem." It can be proven using Theorem 4.3 and a duplication lemma of Lovász [72]. A proof of this theorem which does not use Theorem 4.3 is also given in Lovász [72]. Other proofs can be found in Chvátal [75] and Lovász [79].

Lovász' Perfect graph theorem is so named for the following reason. Berge [62] defined a perfect graph as a graph G such that for each induced subgraph H of G the maximum cardinality of an independent set in H is equal to the minimum cardinality of a family of cliques such that each node of G is in some clique of the family. Based on the properties of some known classes of perfect graphs, Berge [62] conjectured that if a graph G is perfect then its complement \overline{G} is also perfect. Lovász' Perfect graph theorem proves this conjecture. Indeed, if M is the incidence matrix of maximal cliques of a perfect graph G, then $\min\{1 \cdot y \colon yM \ge w, \ y \ge 0\}$ is achieved by an integral vector for each 0-1 valued w. Thus, the vertices of $\{x \colon Mx \le 1, \ x \le 0\}$ are precisely the incidence vectors of the independent sets of G. So the incidence matrix, A, of maximal cliques in \overline{G} is an anti-blocking matrix of M. Since $\{x \colon Ax \le 1, \ x \ge 0\}$ is totally dual integral, it follows that \overline{G} is perfect.

So, by specializing M to clique-node incidence matrices, Theorem 4.3 and Theorem 4.4 give results on perfect graphs. In fact, Fulkerson [71,73] showed that no generality is lost by assuming that M is the clique-node incidence matrix of a graph.

Theorem 4.5: Let M be a 0-1 matrix with no zero column. If $P = \{x \colon Mx \le 1, x \ge 0\}$ is an integer polyhedron, then there exists a graph G such that the essential rows of M are precisely the incidence vectors of the maximal cliques of G.

Proof: Suppose that $P = \{x: Mx \le 1, x \ge 0\}$ is an integer polyhedron. Define a graph G as follows. The nodes of G correspond to the columns of M and nodes j and k are joined by an edge if there exists an essential row m_i of M such that $m_{ik} = m_{ij} = 1$. Clearly, each essential row of M is the incidence vector of a clique of G. Suppose that there is a clique C of G which is not contained in any clique that corresponds to an essential row of M. Let $\overline{x}_i = 1/(k-1)$ for each $i \in C$ and $\overline{x}_i = 0$ for each $i \in VG-C$, where k is the cardinality of C. The vector \overline{x} is contained in P, since $m_i \overline{x} \le 1$ for each essential row m_i of M. Let $w_i = 1$ for each $i \in C$ and $w_i = 0$ for each $i \in VG-C$. Now $w\overline{x} = k/(k-1)$, but $wx \le 1$ for each integral point in P. So P is not an integer polyhedron, a contradiction.

Fulkerson [71,73] gives a different proof of Theorem 4.5, based on the faithful graph representation theorem of Gilmore.

The above proof technique can be used to generalize Theorem 4.5 to h-perfect graphs. Sbihi and Uhry [81] and Fonlupt and Uhry [82] define an h-perfect graph as a graph G such that the following linear system defines an integer polyhedron:

$$\Sigma\{\mathbf{x}_{\mathbf{v}} \colon \mathbf{v} \in \mathbf{C}\} \leq 1 \ \forall \ \mathbf{C} \in \mathbf{C}$$

$$\Sigma\{\mathbf{x}_{\mathbf{v}} \colon \mathbf{v} \in \mathbf{H}\} \leq (|\mathbf{H}| - 1)/2 \ \forall \ \mathbf{H} \in \mathbf{H}$$

$$\mathbf{x}_{\mathbf{v}} \geq 0 \qquad \forall \ \mathbf{v} \in \mathbf{V}\mathbf{G}$$

where C is the set of maximal cliques of G and $H \subseteq V$ is a member of H if the subgraph G[H] induced by H is an odd cycle (i.e. H is an "odd hole"). Clearly, perfect graphs are h-perfect and if G is h-perfect then (4.3) defines the convex hull of the incidence vectors of the independent sets of G. The concept of h-perfection is motivated by a result of Boulala and Uhry [79] which shows that series-parallel graphs are h-perfect. The following generalization of Theorem 4.7 shows that if a system appears to come from an h-perfect graph then it actually does come from an h-perfect graph.

Theorem 4.6 Let C and H be families of subsets of V such that if $H \in H$, then |H| = 2k+1 for some $k \ge 2$, $|H \cap C| \le 2$ for each $C \in C$, and the elements of H may be labeled h_0, \ldots, h_{n-1} such that for each $i = 0, \ldots, n-1$ there exists $C \in C$ with $h_i, h_{i+1} \in C$ (subscripts should be taken modulo n). If the linear system (4.3) defines an integer polyhedron P, then there exists a graph G such that the maximal cliques of G are the sets in C that define facet-inducing inequalities for P and the odd holes of G are the sets in H that define facet-inducing inequalities for P.

<u>Proof:</u> Suppose (4.3) defines an integer polyhedron P. Let G be the graph with node set V with two distinct nodes u,w joined by an edge if there is a set $C \in C$ such that u,w $\in C$. As in the proof of Theorem 4.5, each maximal clique of G is a set in C that defines a facet-inducing inequality for P. Similarly, each odd hole of G is a set in H that defines a facet-inducing inequality for P. Also, each $C \in C$ is a clique of G. It is straightforward to check that each set $H \in H$ which defines a facet-inducing inequality for P is an odd hole of G (use the labelling h_0, \ldots, h_{n-1} of H to show that

G[H] contains a cycle which meets all of the nodes in H and note that if G[H] contains an edge which is not in this cycle then $\Sigma\{x_v\colon v\in H\}\le (|H|-1)/2 \text{ is implied by other valid inequalities for P}),$ which completes the proof. \square

Analogous to anti-blocking theory, Fulkerson developed a theory of blocking polyhedra. Blocking theory involves systems of the form $Bx \ge 1$, $x \ge 0$ where B is a nonnegative matrix. One of the main differences in blocking theory and anti-blocking theory is that the analogue of the result in Theorem 4.3 that $Mx \le 1, x \ge 0$ is totally dual integral if and only if $Ax \le 1$, $x \ge 0$ is totally dual integral does not hold in the blocking case. Nevertheless, some very interesting results involving blocking theory and total dual integrality can be found in Fulkerson, Hoffman, and Oppenheim [74] and Seymour [77].

5. Hilbert bases and totally dual integral defining systems.

Let S be a finite set of integer vectors in Qⁿ. Suppose that S corresponds to some set of combinatorial objects (S could be the set of incidence vectors of matchings in a graph, for instance). Let $\mathtt{w} \in \mathtt{Q}^\mathtt{n}$ be an integer "weight" vector — the weight of an object s \in S being ws. A well known method for finding a min-max relation for the maximum weight of an object in S is to find a linear system that defines the convex hull of S and then apply the duality theorem of linear programming (see Chapter 1). If this linear system is also totally dual integral, then the min-max relation can be strengthened by requiring the dual variables to take on integer values (often integer solutions to the dual linear program correspond to combinatorial objects, such as "coverings" or "cuts"). There always exists a linear system that defines the convex hull of S, but does there always exist a totally dual integral defining system for this polyhedron? A result of Giles and Pulleyblank [79] given below shows that the answer to this question is yes and, in fact, that this system can be taken to have integer left hand sides.

Let C be a polyhedral cone and let A be an integer matrix such that $C = \{x : Ax \le 0\}$. Consider the linear program

(5.1) $\max\{wx: Ax \le 0\}$,

where w is an integer vector. An optimal solution to the dual linear program of (5.1) corresponds to an expression of w as a nonnegative linear combination of a_1, \ldots, a_m (the rows of A). So $Ax \le 0$ is totally dual integral if and only if each integer vector in the cone C* generated by $\{a_1, \ldots, a_m\}$ can be expressed as a nonnegative integral

combination of a_1, \dots, a_m . Furthermore, if $\{b_1, \dots, b_t\}$ is any set of vectors that generates C*, then $C = \{x \colon Bx \le 0\}$, where the rows of B are b_1, \dots, b_t . So there exists a totally dual integral defining system for C if and only if there exists a set of vectors $\{c_1, \dots, c_q\}$ that generates C* such that each integral w in C* can be expressed as a nonnegative integer combination of c_1, \dots, c_q . Hilbert [90] proved that such vectors always exist.

Theorem 5.1 (Hilbert's Finite Basis Theorem): If C is a polyhedral cone then there exists a finite set of integer vectors $\{h_1,\ldots,h_k\}$ such that each integral $v\in C$ can be expressed as a nonnegative integer combination of h_1,\ldots,h_k .

Proof: By the theorem of Weyl[35] and Minkowski[96], there exists a finite set of integer vectors, $\{d_1,\ldots,d_j\}$, that generate C. Let H be the set of integer vectors in the bounded set $\{\lambda_1d_1,+\ldots+\lambda_jd_j\colon 0\leq \lambda_i\leq 1,$ $i=1,\ldots,j\}$. Let $v\in C$ be an integer vector. Since $\{d_1,\ldots,d_j\}$ generates C, v can be expressed as $\gamma_1d_1+\ldots+\gamma_jd_j$ for some nonnegative rational γ_i , $i=1,\ldots,j$. The vector $v'=(\gamma_1-\lfloor\gamma_1\rfloor)d_1+\ldots+(\gamma_j-\lfloor\gamma_j\rfloor)d_j$ is in H, so $v'+(\lceil\gamma_1\rceil d_1+\ldots+\lceil\gamma_j\rceil d_j)$ expresses v as a nonnegative integer combination of vectors in H. Thus, H is the required set of vectors.

The proof of Theorem 5.1 given above is contained in Giles and Pulleyblank [79]. Proofs of this theorem which do not make use of the Weyl-Minkowski Theorem can be found in Bachem [77] and Jeroslow [78].

In view of Theorem 5.1, a set of integer vectors $\{h_1,\ldots,h_k\}$ is called a <u>Hilbert basis</u> if each integer vector in the cone generated by $\{h_1,\ldots,h_k\}$ can be expressed as a nonnegative integer combination of h_1,\ldots,h_k (so Theorem 5.1 implies that every polyhedral cone is generated by a finite Hilbert basis). Theorem 5.1 will now be used to prove a result of Giles and Pulleyblank [79].

Let $Ax \leq b$ be a linear system with A integral. If F is a face of the polyhedron $P = \{x : Ax \leq b\}$, then a row a_i of A is active for F if $a_i \bar{x} = b_i$ for each $\bar{x} \in F$ (if a_i is active for F then $a_i x \leq b_i$ is an active inequality for F). Let C_F be the cone generated by the active rows of A for F. It follows from the complementary slackness theorem that C_F is the set of all those vectors w such that $\max\{wx : Ax \leq b\}$ is achieved by each point in F. So C_F does not depend on the defining system for P. The complementary slackness theorem also implies the following lemma.

<u>Lemma 5.2:</u> If A is an integral matrix, then $Ax \le b$ is totally dual integral if and only if for each minimal face of $\{x: Ax \le b\}$ the set of active rows of A is a Hilbert basis.

Giles and Pulleyblank [79] proved the following theorem.

Theorem 5.3: Let P be a polyhedron. There exists a totally dual integral system $Ax \le b$, with A integral, such that $P = \{x: Ax \le b\}$. Furthermore, if P is an integer polyhedron then b can be taken to be integral.

<u>Proof:</u> Let A'x \leq b' be a defining system for P. For each minimal face F of P let C_F denote the cone generated by the active rows of A' for F and let H_F be a Hilbert basis for C_F (i.e. H_F generates C_F). Also, for each $h \in H_F$ let $\alpha_h = h \cdot \bar{x}$ where \bar{x} is some point in F (note that if x^1 and x^2 are any two points in F then $hx^1 = hx^2$). The inequality $hx \leq \alpha_h$ is valid for P since it can be obtained by taking nonnegative multiples of the active inequalities in A'x \leq b' for F.

Let $Ax \leq b$ be the system of inequalities $hx \leq \alpha_h$ for each $h \in H_F$ and each minimal face F of P. Each facet-inducing inequality for P in $A'x \leq b'$ can be obtained by taking nonnegative multiples of inequalities in $Ax \leq b$ (since the inequality must be active for some minimal nonempty face of P). So $Ax \leq b$ is a defining system for P. If F is a minimal face of P, then each $h \in H_F$ is active for P in P. So Lemma 5.2 implies that P is totally dual integral.

Finally, if P is an integer polyhedron then each minimal nonempty face F of P contains an integer point, which implies that α_h is integral for each h ϵ H_F. So if P is an integer polyhedron then b is integral.

Consider again the finite set S of integer vectors. Suppose that a min-max relation for the maximum weight of an object in S has been obtained by finding a totally dual integral defining system, $Ax \leq b$, for the convex hull of S, with A integral. As was mentioned earlier, the min-max relation can be strengthened by restricting the dual variables to integer values. The min-max relation can be further

strenghtened if some of the inequalities in $Ax \le b$ can be removed to obtain a smaller totally dual integral defining system for the convex hull of S. This process can be repeated until a minimal totally dual integral defining system for the convex hull of S is obtained, i.e. until a totally dual integral system is obtained such that if any inequality is removed from the system then it either no longer defines the convex hull of S or is no longer totally dual integral. If the convex hull of S is a polyhedron of full dimension, then a result of Schrijver [81] implies that such a minimal totally dual integral system is uniquely determined by S. So, if the convex hull of S is of full dimension, a "best possible" min-max relation for S can be obtained in this way (see Chapter 3).

The following theorem, given in Jeroslow [78] and Schrijver [81], is needed in order to prove the above mentioned result of Schrijver (the proof is taken from Schrijver [80a]).

Theorem 5.4: If C is a pointed polyhedral cone then C is generated by a unique minimal Hilbert basis.

Proof: Suppose that C is a pointed polyhedral cone (recall that C is pointed if $x \in C$ implies $-x \notin C$ for each $x \neq 0$) and that C is generated by distinct minimal Hilbert bases H_1, H_2 . Since C is pointed, there exists a vector w such that wc > 0 for each nonzero $c \in C$. Let h be a vector in the symmetric difference of H_1 and H_2 with wh minimized. It can be assumed that $h \in H_1$. Since H_2 is a Hilbert basis, h can be expressed as $\lambda_1 h_1 + \ldots + \lambda_k h_k$ where $\{h_1, \ldots, h_k\} \subseteq H_2$ and λ_i is a positive integer, $i = 1, \ldots, k$. Since $wh = \lambda_1 wh_1 + \ldots + \lambda_k wh_k$ and $h \notin \{h_1, \ldots, h_k\}$, $wh_i \leq wh$, $i = 1, \ldots, k$.

So, by the choice of h, for $i=1,\ldots,k$ h_i is in $H_1\cap H_2$, which implies that H_1 is not minimal, a contradiction.

Suppose that $P = \{x \colon Ax \le b\}$ is of full dimension. Let F be a minimal face of P and let C_F be the cone generated by the active rows of A for F. Suppose that there exists a nonzero $c \in C_F$ such that $-c \in C_F$. Let $\alpha = c\overline{x}$ where \overline{x} is a point in F. Now $cx \le \alpha$ and $-cx \le -\alpha$ are valid inequalities for P. So P lies on the hyperplane $cx = \alpha$, which is impossible since P is of full dimension. So C_F is a pointed cone. This fact will be used in the proof of the following theorem of Schrijver [81].

Theorem 5.5: Let P be a polyhedron of full dimension. There exists a unique minimal totally dual integral system $Ax \le b$ with A integral such that $P = \{x: Ax \le b\}$. Furthermore, P is an integer polyhedron if and only if b is integral.

<u>Proof:</u> Proceed as in the proof of Theorem 5.3, but for each minimal nonempty face F of P let H_F be the unique minimal Hilbert basis for the pointed cone C_F . The system $Ax \leq b$ that is obtained as in the proof of Theorem 5.3 is a totally dual integral defining system for P with A integral. Suppose that $A''x \leq b''$ is a totally dual integral defining system for P with A'' integral. It will be shown that each inequality in $Ax \leq b$ must be present in the system $A''x \leq b''$ (it will then follow that $Ax \leq b$ is the unique minimal totally dual integral defining system for P with integer left hand sides).

Let F be a minimal face of P and let $h\in H_F$. Suppose that $hx \leq \alpha_h \text{ is not an inequality in the system } A''x \leq b''\text{.} \text{ Consider the } A''x \leq b''\text{.}$

linear program

(5.2) $\max\{hx: A''x \le b''\}.$

Each vector in F is an optimal solution to this linear program, so the complementary slackness theorem implies that each optimal solution to the dual linear program of (5.2) corresponds to an expression of h has a nonnegative linear combination of the rows in A" that are active for F. Each active row in A" for F is in the cone C_F . Since h cannot be expressed as a nonnegative integral combination of other integer vectors in C_F , there does not exist an integer optimal solution to the dual linear program of (5.2). So A"x \leq b" is not totally dual integral, a contradiction.

It only remains to be shown that P is an integer polyhedron if and only if b is integral, but this follows directly from the theorem of Hoffman and Edmonds-Giles (Corollary 2.5 of this chapter) and the second part of Theorem 5.3.

This result is the analogue of the well known theorem that there exists a unique (up to positive scalar multiples of the inequalities) minimal defining system for each full dimensional polyhedron. If P is a polyhedron of full dimension and $Ax \le b$ is the unique minimal totally dual integral defining system for P with integer left hand sides, then call $Ax \le b$ the <u>Schrijver system</u> for P. The Schrijver systems for various combinatorial polyhedra associated with matchings will be studied in Chapter 3.

It should be noted that a result for non-full dimensional polyhedra, analogous to Theorem 5.5, is also given in Schrijver [81].

6. Operations that preserve total dual integrality.

The result of Giles and Pulleyblank [79] given in the previous section shows that for any finite set of integer vectors a min-max relation for S can be obtained by finding a totally dual integral defining system for the convex hull of S. However, how can such a system be found? Consider the analogous problem of finding a linear system that defines the convex hull of S. Certain operations that can be performed on a linear system such that if the linear system defines an integer polyhedron then the derived system does as well are quite useful. Some of these operations, such as adding slack variables or taking scalar multiples of inequalities, are useful because they allow certain assumptions to be made on the form of a linear system without losing any generality. Others, such as setting inequalities to equalities or applying Fourier-Motzkin elimination, are useful because often a system of interest arises by applying the operations to a system which is known to define an integer polyhedron. The problem of determining whether or not these operations preserve total dual integrality of linear systems is dealt with in this section (cf. Cook [83]). Other aspects of finding totally dual integral defining systems are studied in Section 10 of this chapter.

The first type of operations that will be looked at are those that allow assumptions to be made on the form of a system. It may be assumed, without loss of generality, that systems are of the form $Ax \leq b$. Throughout this section let f denote a vector and f a scalar.

a) Adding slack variables.

In linear programming, linear systems are often transformed by adding "slack" variables to obtain problems which are in the correct

form for the simplex algorithm. The following result shows that, with integral data, such a transformation maintains total dual integrality of a linear system.

<u>Proposition 6.1:</u> Let f be integral. The system $\{Ax \le b, fx \le \beta\}$ is totally dual integral if and only if the system $\{Ax \le b, fx+s=\beta, s \ge 0\}$ is totally dual integral.

<u>Proof:</u> Let D1 denote the linear program $\min\{yb+z\beta\colon yA+zf=w,\ y\geq 0\}$, $z\geq 0\}$ and let D2 denote the linear program $\min\{yb+z\beta\colon yA+zf=w,\ z\geq w_s,\ y\geq 0\}$. Suppose that the system $\{Ax\leq b,\ fx+s=\beta,\ s\leq 0\}$ is totally dual integral and let w be an integer vector such that D1 has an optimal solution. An integral optimal solution to D1 can be found by setting $w_s=0$ in D2. Now suppose that $\{Ax\leq b,\ fx\leq \beta\}$ is totally dual integral and let (w,w_s) be an integral vector such that D2 has an optimal solution. An integral optimal solution to D2 is $(y*,z*+w_s)$, where (y*,z*) is an integral optimal solution to D1 with right hand side $w'=w-w_sf$.

This result does not hold in general for nonintegral f, e.g. $\{x_1 \le 0, (2/3)x_1 \le 0\}$ is totally dual integral but $\{x_1 \le 0, (2/3)x_1+x_2 = 0, x_2 \ge 0\}$ is not.

b) Splitting unrestricted variables.

Another transformation that is often used in linear programming is the replacement of variables which are not restricted to nonnegative values by the difference of two nonnegative variables. This transformation allows the assumption to be made that all variables are nonnegative.

However, it does not preserve total dual integrality of linear systems, e.g. the system $\{x_1+5x_2 \le 1, x_1+6x_2 \le 1\}$ is totally dual integral but the system obtained by replacing x_1 by $x_1' - x_1''$ and x_2 by $x_2' - x_2''$, where $x_1' \ge 0$, $x_1'' \ge 0$, $x_2'' \ge 0$, is not totally dual integral (this example also shows that the property of defining an integer polyhedron is not preserved under such a splitting operation, since $x_1' = 0$, $x_1'' = 0$, $x_2' = (1/6)$, $x_2'' = 0$ is a nonintegral vertex of the derived polyhedron). So, when working with totally dual integral systems, it cannot be assumed that all variables are nonnegative.

c) Scalar multiplication of inequalities.

It is trivial that if a system A'x \leq b' is obtained by multiplying some inequality in Ax \leq b by a positive scalar then the two systems define the same polyhedron. However, it is not possible to multiply an inequality by an arbitrary positive scalar and maintain total dual integrality of a system, e.g. $\mathbf{x}_1 \leq 0$ is totally dual integral but $2\mathbf{x}_1 \leq 0$ is not. In fact, Giles and Pulleyblank [79] have shown that for any system $\mathbf{A}\mathbf{x} \leq \mathbf{b}$ there exists a positive scalar d such that $\mathbf{d}\mathbf{A}\mathbf{x} \leq \mathbf{d}\mathbf{b}$ is totally dual integral. However, an inequality in a totally dual integral system can be multiplied by a scalar of the form $1/\mathbf{k}$, where k is a positive integer, since solutions to the new dual linear program can be found by setting a single component \mathbf{y}_i to $\mathbf{k}\mathbf{y}_i$ in a solution to the original dual linear program. For linear systems with integer left hand sides that define full dimensional polyhedra, the following proposition shows that this is the only type of nontrivial scalar multiplication that is possible.

<u>Proposition 6.2:</u> Let A and f be integral and let d be a positive scalar. Suppose that $Ax \le b$, $fx \le \beta$ is totally dual integral and that $P = \{x: Ax \le b, fx \le \beta\}$ is of full dimension. The system $Ax \le b$, $dfx \le d\beta$ is also totally dual integral if and only if either $fx \le \beta$ is not in the Schrijver system for P or d is of the form 1/k for some positive integer k.

Proof: The sufficiency of either condition is easily seen. To show necessity, suppose $\{Ax \leq b, dfx \leq d\beta\}$ is totally dual integral and $fx \leq \beta$ is in the Schrijver system for P. There exists a convex cone, C, generated by the active rows of some minimal face, F, of P, that contains f in its unique minimal Hilbert basis (C is pointed since P is of full dimension). Let λ be the multiplier of df in an expression of f as a nonnegative integer combination of the active rows of F in $\{A,df\}$. The multiplier λ must be nonzero since f cannot be expressed as a nonnegative integer combination of other integer vectors in C. Since $-f \notin C$ (C is pointed), λd must be less than or equal to 1. Since λd must be integral, $\lambda df = f$ (f is the first nonzero integral point on the ray $\{y_1 f \colon y_1 \leq 0\}$). So $d = 1/\lambda$.

d) Unimodular transformations.

Let U be a square integral matrix such that $\det(U) = \pm 1$. A simple and well known result is that $Ax \le b$ is totally dual integral if and only if $AUx \le b$ is totally dual integral. (This corresponds to the result that P is an integer polyhedron if and only if $P' = \{U^{-1}x: x \in P\}$ is an integer polyhedron.) This result is used in Section 8 of this chapter in connection with Hermite normal forms.

The remaining two operations that will be looked at are the type that can be used to prove that a given system is totally dual integral

by showing that it can be obtained via the operation from a system that is known to be totally dual integral (of course, this remark can be made about operations a) through d) as well, but those operations are usually not used in this way).

e) Setting inequalities to equalities.

The following proposition is given in Schrijver [82b]. For completeness a proof is given here.

Proposition 6.3: If $\{Ax \le b, \ x \le \beta\}$ is totally dual integral, then $\{Ax \le b, \ fx = \beta\}$ is also totally dual integral.

<u>Proof:</u> Note that $\{Ax \le b, fx = \beta\}$ is totally dual integral if and only if $\{Ax \le b, fx \le \beta, -fx \le -\beta\}$ is totally dual integral. Also, if $a_1, \ldots, a_k, f, -f$ are the active rows of a minimal face of $\{x: Ax \le b, \}$ $fx \leq \beta$, $-fx \leq -\beta$ then a_1, \dots, a_k, f are the active rows of a minimal face of $\{x: Ax \leq b, fx \leq \beta\}$. So it suffices to show that if each integer vector in the cone generated by a_1, \ldots, a_k can be expressed as a nonnegative integer combination of a_1, \dots, a_k , then each integer vector in the cone generated by $a_1, \dots, a_k, -a_1$ can be expressed as a nonnegative integer combination of $a_1, \dots, a_k, -a_1$. Suppose that r is an integral vector and $r = \Sigma\{\lambda_i a_i : i = 1,...,k\} - \gamma a_i$, where $\lambda_i \ge 0$, i = 1,...,kand $\gamma \geq 0$. Choose γ' such that $r+\gamma'a_1$ is in the cone generated by a_1,\ldots,a_k and such that $\gamma'a_1$ is integral and γ' is integral. By assumption, there exist nonnegative integers λ_i^{\dagger} , $i=1,\ldots,k$, such that $r+\gamma'a_1 = \Sigma\{\lambda_i'a_i: i=1,...k\}$. Now $\Sigma\{\lambda_i'a_i: i=1,...,k\} - \gamma'a_1$ expresses ras a nonnegative integer combination of $a_1, \dots, a_k, -a_1$.

This result corresponds to the well known result that any face of

an integer polyhedron is also an integer polyhedron. It can be used to find a totally dual integral defining system for any face of a polyhedron for which such a system is known, e.g. totally dual integral defining systems for the convex hull of the perfect matchings of a graph and the convex hull of the common bases of two matroids can be derived from the totally dual integral defining systems for the matching polyhedron and the matroid intersection polyhedron respectively (cf. Pulleyblank [82] and Chapter 3).

f) Fourier-Motzkin elimination.

Let K be a set in \mathbb{Q}^n and let $I \subseteq \{1,\ldots,n\}$. If $x \in \mathbb{Q}^n$, let x_I denote the vector $(x_i \colon i \in I)$. The <u>projection</u> of K onto the I coordinates is the set $\{x_I \colon x \in K\}$. A well known and useful result is that any projection of an integer polyhedron is again an integer polyhedron (for an example of its application see Balas and Pulleyblank [82]). To obtain a corresponding result for total dual integrality, linear systems must be considered.

If a defining system for a polyhedron is given then a defining system for any projection of the polyhedron can be found by repeated application of Fourier-Motzkin elimination (cf. Stoer and Witzgall [70]). The version of Fourier-Motzkin elimination used here is such that if the original system has integral data then the resulting system does as well (the result of Giles and Pulleyblank [79] mentioned in subsection c shows that the scaling of the resulting system must be specified). Consider the linear system

(6.1)
$$a_{i}x - \alpha_{i}x_{o} \leq b_{i} \quad i \in I = \{1, ..., i_{o}\}$$

$$c_{j}x + \gamma_{j}x_{o} \leq d_{j} \quad j \in J = \{1, ..., j_{o}\}$$

$$f_{k}x \qquad \leq g_{k} \quad k \in K = \{1, ..., k_{o}\}$$

where a_i, c_j, f_k are vectors, α_i, γ_j are positive scalars, and b_i , d_j, g_k are scalars, for $i \in I$, $j \in J$, $k \in K$. The following system is obtained by applying Fourier-Motzkin elimination to eliminate the variable x_0

$$(\gamma_{j}a_{i}^{\dagger}+\alpha_{i}c_{j})x \leq \gamma_{j}b_{i}^{\dagger}+\alpha_{i}d_{j} \qquad i \in I, j \in J$$

$$(6.2) \qquad f_{k}x \leq g_{k} \qquad k \in K.$$

It is not true that (6.2) must be totally dual integral if (6.1) is, even if all data is integral and the coefficients of any row of (6.2) have greatest common divisor 1. For example, $\{x_1-x_3 \le 0, x_2-x_3 \le 0, -x_2+2x_3 \le 0\}$ is totally dual integral since the coefficient matrix is unimodular but $\{2x_1-x_2 \le 0, x_2 \le 0\}$ is not totally dual integral. However, the result is true in a special case.

Theorem 6.4: Let $Ax \le b$ be a totally dual integral system. If each coefficient of the variable x_0 is either 0,1, or -1 then the system obtained by eliminating x_0 by Fourier-Motzkin elimination is also totally dual integral.

<u>Proof:</u> What must be shown is that if (6.1) is totally dual integral and α_i, γ_j are equal to 1 for $i \in I$, $j \in J$, then (6.2) is also totally dual integral. Let D1 denote the linear program

$$\min \quad \sum \{b_{i}y_{i}^{!}: i \in I\} + \sum \{d_{j}y_{j}^{"}: j \in J\} + \sum \{g_{k}z_{k}: k \in K\}$$

$$\text{s.t.} \quad \sum \{a_{i}y_{i}^{!}: i \in I\} + \sum \{c_{j}y_{j}^{"}: j \in J\} + \sum \{f_{k}z_{k}: k \in K\} = w$$

$$-\sum \{y_{i}^{!}: i \in I\} + \sum \{y_{j}^{"}: j \in J\} = w_{0}$$

$$y_{i}^{!} \geq 0, y_{j}^{"} \geq 0, z_{k} \geq 0 \quad i \in I, j \in J, k \in K$$

and let D2 denote the linear program

min
$$\Sigma\{(b_i+d_j)y_{ij}: i \in I, j \in J\}+\Sigma\{g_kz_k: k \in K\}$$

(6.4) s.t. $\Sigma\{(a_i+c_j)y_{ij}: i \in I, j \in J\}+\Sigma\{f_kz_k: k \in K\} = w$

$$y_{ij} \geq 0, z_k \geq 0 \quad i \in I, j \in J, k \in K.$$

Suppose that w is integral and D2 has an optimal solution (\bar{y},\bar{z}) . An optimal solution to D1 with $w_0=0$ is \bar{z} together with

(6.5)
$$\overline{y}_{i}' = \Sigma \{\overline{y}_{ij} : j \in J\} \qquad i \in I$$

$$\overline{y}_{i}'' = \Sigma \{\overline{y}_{ij} : i \in I\} \qquad j \in J$$

since any solution (y',y'',z) to D1 with $w_0 = 0$ corresponds to a solution to D2 with the same objective value, by finding nonnegative y_{ij} 's such that

(6.6)
$$\Sigma\{y_{ij}: j \in J\} = y'_{i} \qquad i \in I$$
$$\Sigma\{y_{ij}: i \in I\} = y''_{i} \qquad j \in J$$

 $(\mathbf{w}_0 = 0 \text{ implies that } \Sigma\{\mathbf{y}_1' \colon \mathbf{i} \in \mathbf{I}\} = \Sigma\{\mathbf{y}_j'' \colon \mathbf{j} \in \mathbf{J}\})$. An integral optimal solution to D2 can be found by finding an integral optimal solution, $(\mathbf{y}, \mathbf{y}, \mathbf{z})$, to D1 with $\mathbf{w}_0 = 0$ and then finding a nonnegative integral y that satisfies (6.6).

7. Integer programming.

Suppose that $Ax \le b$ is totally dual integral and that b is integral. The results of Hoffman and Edmonds-Giles given in Section 2 show that both sides of the equation

(7.1)
$$\max\{wx: Ax \le b\} = \min\{yb: yA = w, y \ge 0\}$$

can be achieved by integer vectors for each integer vector \mathbf{w} for which the optima exist. The problem dealt with in this section is that of finding integer \mathbf{x} and \mathbf{y} that obtain the optima for a given integer vector \mathbf{w} .

The left hand side of (7.1) can be handled easily. All that is needed is the fact that $Ax \le b$ defines an integer polyhedron (which follows from the results of Section 2 since b is integral) and the following theorem.

Theorem 7.1: There exists a polynomial-time algorithm which, for any linear system $Ax \le b$ and any vector w for which the maximum in (7.1) exists, either finds an integer vector which achieves the maximum in (7.1) or shows that $Ax \le b$ does not define an integer polyhedron.

<u>Proof:</u> Let $Ax \le b$ be a linear system, w a vector such that the maximum in (7.1) exists, and P the polyhedron $\{x: Ax \le b\}$. A minimal face $F = \{x: A'x = b'\}$ of P will be found such that every point in F achieves the maximum in (7.1). Once F is found, the problem is reduced to solving a system of linear diophantine equations, which can be done in polynomial-time by the algorithm of Kannan and Bachem [79] for finding the Smith normal form of a matrix (other polynomial-time algorithms for solving systems of linear diophantine equations are given

in Frumkin [76] and Gathen and Sieveking [76]). If the linear system A'x = b' has no integer solution, then F is a face of P which does not contain an integer point, which proves that P is not an integer polyhedron. To find F in polynomial time, the ellipsoid method of Khachiyan [79] will be used repeatedly. First, find an optimal solution to $\max\{wx\colon Ax\leq b\}$ and let $\beta=\max\{wx\colon Ax\leq b\}$. Consider the linear system

Any solution to (7.2) achieves the maximum in (7.1). Solve the linear program $\max\{a_1x: (7.2)\}$, where $a_1x \le b_1$ is the first inequality in $Ax \le b$. If this maximum is b_1 , then add $a_1x = b_1$ to (7.2). Repeat this procedure until each inequality in $Ax \le b$ has been considered. Now letting A'x = b' consist of the equalities $a_1x = b_1$ that were added to (7.2), F has been found.

When A is integral, the right hand side of (7.1) can be handled by the following result of Chandrasekaran [81] (the description of the algorithm given below is from Cook, Lovász and Schrijver [82]).

Theorem 7.2: There exists a polynomial-time algorithm which, for any linear system $Ax \le b$ with A integral and any integer vector w for which the minimum in (7.1) exists, either finds an integer vector which achieves the minimum in (7.1) or shows that $Ax \le b$ in not totally dual integral.

<u>Proof:</u> Let $Ax \le b$ a linear system with A integral and let w be an integer vector such that the minimum in (7.1) exists. Again, the ellipsoid method will be used repeatedly. Let a_1, \ldots, a_k be those rows

of A such that $a_1\bar{x}=b_1$ for each vector \bar{x} which achieves the maximum in (7.1), where $a_1x\leq b_1$ is the corresponding inequality in $Ax\leq b$ (these rows can be found by considering the linear program $\min\{a_1x\colon Ax\leq b,\ wx\leq z^*\}$, where $z^*=\max\{wx\colon Ax\leq b\}$). By the complementary slackness theorem, each vector which achieves the minimum in (7.2) corresponds to an expression of w as a nonnegative linear combination of a_1,\ldots,a_k .

Let C be the cone generated by $\{a_1,\ldots,a_k\}$. Relabel a_1 through a_k if necessary so that a_1,\ldots,a_j are those vectors a_i in $\{a_1,\ldots,a_k\}$ such that $-a_i \notin C$. Let λ_1 be the largest number such that $w-\lambda_1a_1$ is in C, i.e. $\lambda_1=\max\{z_1\colon y\mathbb{A}=w-z_1a_1,\ldots y\ge 0\}$ (since $-a_1\notin C$, λ_1 exists). Let $w^1=w-l\lambda_1Ja_1$ and let $\lambda_2=\max\{z_2\colon y\mathbb{A}=w^1-z_2a_2\}$. Let $w^2=w^1-\lfloor\lambda_2Ja_2\rfloor$ and so forth until w^j is found. Now use the algorithm of Kannan and Bachem [79] to solve a system of linear diophantine equations to find integer $\lambda_{j+1},\ldots,\lambda_k$ such that $w^j=\lambda_{j+1}a_{j+1}+\ldots+\lambda_ka_k$, if such $\lambda_{j+1},\ldots,\lambda_k$ exist. (Note that the λ_i 's may be negative.)

Suppose that no such integer $\lambda_{j+1},\ldots,\lambda_k$ exist. If $\gamma_1 a_1 + \ldots + \gamma_k a_k$ is an expression of w^j as a nonnegative integer combination of a_1,\ldots,a_k , then $\gamma_1=\gamma_2=\ldots=\gamma_j=0$, by the choice of $\lambda_1,\ldots,\lambda_j$. So w^j cannot be expressed in this way. Now w^j is in C, so $\min\{yb\colon yA=w^j,\ y\ge 0\}$ exists and the minimum can be achieved by an integer vector if and only if w^j can be expressed as a nonnegative

integer combination of a_1, \dots, a_k . Since w^j is integral, this shows that $Ax \le b$ is not totally dual integral.

Now suppose that integer $\lambda_{j+1},\dots,\lambda_k$ have been found. By expressing $-a_i$ as a nonnegative linear combination of a_{j+1},\dots,a_k (excluding a_i) for $i=j+1,\dots,k$, find positive $\gamma_{j+1},\dots,\gamma_k$ such that $0=\gamma_{j+1}a_{j+1}+\dots+\gamma_ka_k$. Find an integer M, of size polynomial in the size of $Ax \leq b$ and w, such that $\lambda_i+M\gamma_i$ is a nonnegative integer for $i=j+1,\dots,k$. Since $w^j=(\lambda_{j+1}+M\gamma_{j+1})a_{j+1}+\dots+(\lambda_k+M\gamma_k)a_k$ and $w=\lfloor\lambda_1\rfloor a_1+\dots+\lfloor\lambda_j\rfloor a_j+w^j$, this gives an expression of w as a nonnegative integer combination of a_1,\dots,a_k (which gives an integer vector which achieves the minimum in (7.1)).

A consequence of this result is that given a Hilbert basis $\{a_1,\ldots,a_k\}$ and an integer vector w in the cone generated by the basis, an expression of w as a nonnegative integer combination of a_1,\ldots,a_k can be found in polynomial time.

In their outstanding paper on consequences of the ellipsoid method, Grötschel, Lovász, and Schrijver [81] have shown that polynomial-time algorithms for finding optimum integer dual solutions to linear programs associated with several different combinatorial problems can be derived from the ellipsoid algorithm. The linear systems involved are often very large, but the algorithms derived are polynomial in the size of the given combinatorial problem (the linear systems are given by an encoding whose size is polynomial in the size of the combintorial problem). It would be interesting to generalize their methods to

obtain a result on finding optimum integer dual solutions that is similar to their general results on finding optimum integer primal solutions.

8. Bounds on the number of nonzero dual variables.

Suppose that we have a linear system $Ax \le b$ encoded in some way and we wish to consider the problem mentioned at the end of the previous section. That is, is there a polynomial-time algorithm which, for any integer vector w such that $\max\{wx\colon Ax\le b\}$ exists, either finds an integer optimal solution to the dual of this linear program or proves that $Ax\le b$ is not totally dual integral? If such an algorithm exists then it must be possible to describe the integer optimal solution in polynomial time. For example, let G be a graph and C the set of maximal cliques of G and suppose that the linear system

(8.1)
$$\Sigma\{x_{\mathbf{v}}: \mathbf{v} \in \mathbf{C}\} \leq 1 \qquad \forall \quad \mathbf{C} \in \mathbf{C}$$
 $x_{\mathbf{v}} \geq 0 \qquad \forall \quad \mathbf{v} \in \mathbf{VG}$

is encoded by the graph G. A by-product of the polynomial-time algorithm of Grotschel, Lovász, and Schrijver [81,81a] for finding an integer optimal solution to the dual linear program of $\max\{wx\colon (8.1)\}$ for any perfect graph G and any nonnegative integer vector w (recall that G is perfect if and only if (8.1) is totally dual integral) is that for any perfect graph G and any nonnegative integer vector w there exists an integer optimal solution to the dual linear program of $\max\{wx\colon (8.1)\}$ with at most n^2+n nonzero variables, where n is the number of nodes of G. The problem considered in this section is whether or not there exists a similar polynomial bound on the number of nonzero dual variables for general totally dual integral systems.

The results presented in this section are due to Cook, Fonlupt, and Schrijver [83].

A more precise statement of the problem is now given. Let h be

the least integer such that for any integer vector w for which

(8.2) $\min\{yb: yA = w, y \ge 0\}$

exists, the minimum can be achieved by an integral solution with at most h nonzero variables. The problem is to ask if, for any totally dual integral system $Ax \le b$, where A is an integral matrix of rank n, is there an upper bound for h in terms of n?

Suppose that $\{a_1,\ldots,a_k\}$ is a Hilbert basis for a cone C of dimension n. By Lemma 5.2, the above problem is equivalent to: Is there an upper bound for h in terms of n, where h is the least integer such that each integer vector in C can be expressed as a nonnegative integer combination of h vectors in $\{a_1,\ldots,a_k\}$? Carathéodory's Theorem implies that each vector w in C can be expressed as a nonnegative linear combination of n vectors in $\{a_1,\ldots,a_k\}$, so the problem dealt with here is an integer analogue of the one solved by Carathéodory (see Rockafeller [70]). It will be shown that in general there is no upper bound for h in terms of n. However, if C is pointed, the following result gives a bound for h.

Theorem 8.1: Let $\{a_1, \ldots, a_k\}$ be a Hilbert basis for a pointed cone C of dimension n. If w is an integer vector in C then w can be expressed as a nonnegative integer combination of 2n-1 vectors in $\{a_1, \ldots, a_k\}$.

<u>Proof:</u> It can be assumed that 0 is not a vector in $\{a_1, \ldots, a_k\}$. Suppose that w is an integer vector in C. Let A be the matrix with rows a_1, \ldots, a_k . Since C is pointed, the linear program $\max\{1y\colon yA=w,\ y\geq 0\}$ has a basic optimal solution \overline{y} (it has a

solution since w ϵ C and it is easy to see that it is bounded by considering a vector d such that dz > 0 for each nonzero z ϵ C). Let $|\bar{y}|$ denote the vector $(|\bar{y}_1|, \dots, |\bar{y}_k|)$. The vector $w-|\bar{y}|A$ is an integer vector in C and so can be expressed as $\lambda_1 a_1 + \dots + \lambda_k a_k$ where λ_i is a nonnegative integer for $i=1,\dots,k$. Now \bar{y} has at most n nonzero components, since it is a basic solution. So $1 \cdot (\bar{y}-|\bar{y}|) < n$. Now, since \bar{y} is an optimal solution and $(\lambda_i + |\bar{y}_i|: i=1,\dots,k)$ is a feasible solution to the linear program $\max\{1y\colon yA=w,\ y\geq 0\}$, it follows that $\Sigma\{\lambda_i + |\bar{y}_i|:\ i=1,\dots,k\}\leq 1\bar{y}$ and $\Sigma\{\lambda_i:\ i=1,\dots,k\}\leq 1 \cdot (\bar{y}-|\bar{y}|) < n$. So $(\lambda_1 + |\bar{y}_1|)a_1 + \dots + (\lambda_k + |\bar{y}_k|)a_k$ expresses w as a nonnegative integer combination of a_1,\dots,a_k with at most 2n-1 nonzero multipliers.

The above upper bound can be improved slightly if all vectors in the Hilbert basis are 0-1 valued.

Theorem 8.2: Let $\{a_1,\ldots,a_k\}$ be a Hilbert basis for a cone C of dimension n>1, where a_i is 0-1 valued for $i=1,\ldots,k$. If w is an integer vector in C then w can be expressed as a nonnegative integer combination of 2n-2 vectors in $\{a_1,\ldots,a_k\}$.

<u>Proof:</u> Since a_i is 0-1 valued for $i=1,\ldots,k$, C is pointed. Suppose that w is an integer vector in C. Find \bar{y} as in the proof of Theorem 8.1. Suppose that $1\cdot(\bar{y}-\bar{y})\geq n-1$. Let A be the matrix with rows a_i,\ldots,a_i where $\bar{y}_i,\ldots,\bar{y}_i$ are the nonzero variables in the solution \bar{y} (it may be assumed that there are no such variables).

The only possible 0-1 vectors whose inner product with $(\bar{y}_{i_1} - [\bar{y}_{i_1}], \dots, \bar{y}_{i_n} - [\bar{y}_{i_1}])$ is an integer are 0 and the vector of all 1's. But $(\bar{y}_{i_1} - [\bar{y}_{i_1}], \dots, \bar{y}_{i_n} - [\bar{y}_{i_1}]) A = w - ([\bar{y}_{i_1}], \dots, [\bar{y}_{i_n}]) A$, which is integral. So each column of A is either all zeros or all 1's which implies that n is 1, a contradiction. So $1 \cdot (\bar{y} - [\bar{y}]) < n - 1$, which improves the upperbound to 2n - 2.

It is an open question whether or not the upper bound given in Theorem 8.1 can be lowered to n. (It is easy to see that it cannot be less than n.) To see that in general there is no upper bound in terms of n, consider the following example. Let $\mathbf{p}_1,\dots,\mathbf{p}_k$ be distinct primes, $k\geq 2$. For $i=1,\dots,k$ let \mathbf{q}_i be the product of \mathbf{p}_1 through \mathbf{p}_k , excluding \mathbf{p}_i . The greatest common divisor of $\mathbf{q}_1,\dots,\mathbf{q}_k$ is 1, so there exists integer λ_i , $i=1,\dots,k$ such that $\lambda_1\mathbf{q}_1+\dots+\lambda_k\mathbf{q}_k=1$. For $i=1,\dots,k$ let $\mathbf{r}_i=\mathbf{q}_i$ if $\lambda_i\geq 0$ and $\mathbf{r}_i=-\mathbf{q}_i$ if $\lambda_i<0$. Now $\{\mathbf{r}_1,\dots,\mathbf{r}_k\}$ is a Hilbert basis for a cone of dimension 1, but, since the greatest common divisor of any proper subset of $\{\mathbf{r}_1,\dots,\mathbf{r}_k\}$ is greater than 1, any expression of 1 as a nonnegative integer combination of $\mathbf{r}_1,\dots,\mathbf{r}_k$ requires all k vectors.

The above example shows that in general there is no upper bound in terms of rank (A) on the number of nonzero variables needed in an integral solution that achieves (8.2), where $Ax \le b$ is a totally dual integral system with A an integer matrix and w is an integer vector such that the minimum exists. However, as was shown in Section 5, if the polyhedron $P = \{x: Ax \le b\}$ is of full dimension then the active rows in A of any

minimal nonempty face of P generate a pointed cone. Thus, Theorem 8.1 implies the following result.

Theorem 8.3: Let $Ax \le b$ be a totally dual integral system, with A an integer matrix of rank n, such that $P = \{x: Ax \le b\}$ is of full dimension. If w is an integer vector such that (8.2) exists then (8.2) can be achieved by an integer optimal solution with at most 2n-1 nonzero variables.

In a similar manner, Theorem 8.2 implies that if A is a 0-1 matrix of rank n > 1 then the upperbound given above can be lowered to 2n-2. A slight modification of this result is needed to improve a result given in Cunningham [81] (recall the definition of integer rounding given in Section 1).

Theorem 8.4: Let A be a 0-1 matrix with no row of zeros and b a 0-1 vector, such that $Ax \le b$ has the integer rounding property and the matrix [A]b has rank n > 1. If w is a nonnegative integer vector, then there exists an optimal solution to the integer program

(8.3)
$$\min\{yb: yA = w, y \ge 0, y \text{ integral}\}\$$

with at most 2n-2 nonzero variables.

<u>Proof:</u> Suppose that w is a nonnegative integer vector. Let z^* be equal to $\min\{yb: yA = w, y \ge 0\}$ and let y^1 be a basic optimal solution to $\max\{1y: yb = [z^*], yA = w, y \ge 0\}$. Consider the linear program

(8.4)
$$\min\{yb: yA = w-[y^1]A, y \ge 0\}.$$

By the choice of z^* , a lower bound for (8.4) is z^*-y^1 b. Also,

 y^1-ly^1] is a solution to (8.4) with objective value $[z*]-[y^1]b$. So, by the integer rounding property, there exists an integer solution y^2 to (8.4) with objective value $[z*]-ly^1]b$. Now $y^3=[y^1]+y^2$ is an optimal solution to (8.3). Since y^1 is a basic solution, $[y^1]$ has at most n nonzero components. As in the proof of Theorem 8.2, it can be assumed that $1 \cdot (y^1-ly^1) < n-1$. So y^2 has at most n-2 nonzero components and y^3 at most 2n-2 nonzero components.

The problem of Cunningham [81] mentioned above is the following. Let M be a matroid and w a nonnegative integer vector. What is the least integer k such that of all minimum cardinality families of independent sets of M having w as the sum of the incidence vectors of the sets in the family, there exists a family with at most k distinct members? This problem can be stated in terms of linear systems as follows. Let A be a matrix whose rows are the incidence vectors of independent sets of M. The problem is: What is the least k such that, for each nonnegative integer vector w, $min\{1y: yA = w, y \ge 0, y\}$ integral} can be achieved by a vector with at most k nonzero components? Cunningham [81] showed that his algorithm for testing membership in matroid polyhedra gives an upper bound of n +1 for k, where n is the number of elements in M. This bound was improved to 2n by Schrijver (see Cunningham[81]). The matroid partition theorem of Edmonds [65b] shows that $Ax \le 1$ has the integer rounding property, so Theorem 8.4 lowers the upper bound to 2n-2.

It has already been mentioned that a by-product of an algorithm of Grötschel, Lovász, and Schrijver [81,81a] is that if G is a perfect graph and A is the clique-node incidence matrix of G then for each nonnegative integer w, $\min\{1:y: yA \ge w, y \ge 0\}$ can be achieved by an integer vector with at most n^2+n nonzero components,

where n is the number of nodes of G. Another way of stating this is, if G is perfect and w is a nonnegative integer vector then there exists a minimum w-covering of G with at most n^2+n distinct cliques (a minimum w-covering is a minimum cardinality family of cliques such that each node v of G is in at least w cliques in the family). Since the polyhedron $\{x: Ax \le 1, x \ge 0\}$ is of full dimension, Theorem 8.3 improves this bound to 2n-1. In fact, Cook, Fonlupt, and Schrijver [83] showed that n is an upperbound (it is clear that the upper bound cannot be less than n).

There are other combinatorially described totally dual integral systems $Ax \le b$, $x \ge 0$ such that $min\{yb\colon yA \ge w,\ y \ge 0\}$ can be achieved by an integer vector with at most n nonzero components for each integer vector w for which the minimum exists, where n is the number of variables in $Ax \le b$, $x \ge 0$. The systems arising from crossfree families (Schrijver [82a]) have this property — see Section 10. W. Cunningham observed that the algorithm for optimum matchings given in Cunningham and Marsh [78] shows that the totally dual integral matching system (see Chapter 3) has this property (which also follows from the proof of total dual integrality of matching systems given in Schrijver and Seymour [77]).

9. Recognition of totally dual integral systems.

In the previous sections properties of totally dual integral systems, integer polyhedra, and min-max relations were presented. Also, sufficient conditions for a linear system to be totally dual integral were presented in Section 3 (concerning totally unimodular and balanced matrices). However, the general problem of recognizing totally dual integral systems and systems that define integer polyhedra has not yet been dealt with. Lemma 2.3 of Chapter 1 and Lemma 5.2 of this chapter show that there exist finite algorithms to determine whether or not a given linear system is totally dual integral and whether or not it defines an integer polyhedron. The complexity of these recognition problems is studied in this section. Several easy results on integer polyhedra will be presented first, then the corresponding results for totally dual integral systems will be given.

Lemma 2.3 of Chapter 1 is very useful in obtaining results on the problem of recognizing when a linear system defines an integer polyhedron.

<u>Proposition 9.1:</u> The class of linear systems, $Ax \le b$, that define integer polyhedra is in co-NP.

<u>Proof:</u> By Lemma 2.3 of Chapter 1, if $Ax \le b$ does not define an integer polyhedron then there exists a subset, $A'x \le b'$, of the inequalities such that A'x = b' has a solution but does not have an integral solution and $F = \{x: A'x = b'\}$ is a minimal face of $\{x: Ax \le b\}$. Using the algorithm of Kannan and Bachem [79], and the ellipsoid algorithm, it is possible to check in polynomial time that $A'x \le b'$ has the above properties.

It is an open question whether or not the class of linear systems

that define integer polyhedra is in NP (it is also an open question whether or not the problem of recognizing linear systems that define integer polyhedra is co-NP complete), but the following result is true.

<u>Proposition 9.2:</u> For any fixed positive integer r, there exists a polynomial-time algorithm which, for any linear system $Ax \le b$ with A a matrix of rank r, determines whether or not $Ax \le b$ defines an integer polyhedron.

<u>Proof:</u> Suppose that r is a fixed positive integer and $Ax \le b$ is a linear system with A a matrix of rank r. Each minimal nonempty face F of the polyhedron $P = \{x: Ax \le b\}$ is determined by r linearly independent rows of A in the sense that $F = \{x: A'x = b'\}$ $A'x \le b'$ is a subsystem of r inequalities from the system $Ax \le b$ and A' is a matrix of rank r (the rank of A' can be found in polynomial time, for instance, by Edmonds' [67] version of Gaussian elimination). For each such subset of r linearly independent inequalities, check whether $A^{\dagger}x = b^{\dagger}$ has a solution and, if so, whether this solution is in the polyhedron P - this can be done in polynomial time using the ellipsoid algorithm. If A'x = b' has a solution and it is in P, then $F = \{x: A'x = b'\}$ is a minimal nonempty face of P and using the algorithm of Kannan and Bachem [79] it can be checked in polynomial time whether or not F contains integer points. In this way, it can be determined in polynomial time whether or not each minimal nonempty face of P contains integer points and hence whether or not P is an integer polyhedron.

To obtain the corresponding results for totally dual integral systems Lemma 5.2 will be used together with the integer programming results of Section 7. The following result is given in Cook, Lovász,

and Schrijver [82].

Theorem 9.3: The class of linear systems $Ax \le b$ with A integral that are totally dual integral is in co-NP.

Proof: If $Ax \le b$ is not totally dual integral then, by Lemma 5.2, there exists a minimal nonempty face F of $P = \{x: Ax \le b\}$ such that the active rows in A of F do not form a Hilbert basis. Suppose that a_1, \ldots, a_k are the active rows of F (using the ellipsoid algorithm it can be checked in polynomial time that a_1, \ldots, a_k are the active rows of a minimal face of P). Since $\{a_1, \ldots, a_k\}$ is not a Hilbert basis, there exists an integer vector w in $\{\lambda_1 a_1 + \ldots + \lambda_k a_k \colon 0 \le \lambda_i \le 1, i = 1, \ldots, k\}$ such that w cannot be expressed as a nonnegative integer combination of a_1, \ldots, a_k (see the proof of Theorem 5.1). By the results of Section 7, in time polynomial in the size of w and $\{a_1, \ldots, a_k\}$, either an expression of w as a nonnegative integer combination of a_1, \ldots, a_k can be found or it can be shown that $\{a_1, \ldots, a_k\}$ is not a Hilbert basis. Since the size of w is polynomial in the size of $\{a_1, \ldots, a_k\}$ it can be proven in time polynomial in the size of $\{a_1, \ldots, a_k\}$ it can be proven in time polynomial in the size of $\{a_1, \ldots, a_k\}$ it can be proven in time polynomial in the size of $\{a_1, \ldots, a_k\}$ it can be proven in time polynomial in the size of $\{a_1, \ldots, a_k\}$ it can be proven in time polynomial in the size of $\{a_1, \ldots, a_k\}$ that $\{a_1, \ldots, a_k\}$ is not a Hilbert basis. \square

It is not necessary to use Hilbert bases in the above proof, since the theorem also follows from Theorem 7.2 by noting that if $Ax \le b$ is not totally dual integral then there exists an integer vector in $\{w\colon yA=w,\ 0\le y\le 1\}$ such that $\min\{yb\colon yA=w,\ y\ge 0\}$ exists but cannot be achieved by an integer vector. The use of Hilbert bases however corresponds more naturally to the proof of Proposition 9.1. The following theorem and its proof are given in Cook, Lovász, and

Schrijver [82].

Theorem 9.4: For any fixed positive integer r, there exists a polynomial-time algorithm which, for any linear system $Ax \le b$ with A an integer matrix of rank r, determines whether or not $Ax \le b$ is totally dual integral.

<u>Proof:</u> The proof is split into three stages. First, the problem is reduced to determining whether or not a given set of integer vectors $\{a_1,\ldots,a_m\}$ which generate a cone of dimension r is a Hilbert basis. Next, the problem is reduced to the case that the cone generated by $\{a_1,\ldots,a_m\}$ is pointed and of full dimension. Finally, it is shown that Lenstra's integer programming algorithm [81] can be used to test in polynomial time whether or not $\{a_1,\ldots,a_m\}$ is a Hilbert basis.

Suppose that r is a fixed positive integer and that $Ax \le b$ is a linear system with A an integer matrix of rank r. By Lemma 5.2, it must be determined whether or not the active rows of each minimal nonempty face of $P = \{x: Ax \le b\}$ form a Hilbert basis. As in the proof of Proposition 9.2, the minimal nonempty faces of P can be found by considering each subset of r linearly independent rows of A. Once a minimal nonempty face F of P is found, the active rows in A of F can be found by finding a vector in F and checking which inequalities in $Ax \le b$ hold as an equality for that vector (since F is a minimal nonempty face, if an inequality in $Ax \le b$ holds as an equality for one vector in F then it holds as an equality for every vector in F). Since r is fixed, there are only polynomially many subsets of r rows of A to consider, so it suffices to prove that there exists a polynomial-time algorithm to test whether or not a a given set of integer vectors $\{a_1, \ldots, a_m\}$ is a Hilbert basis,

where $\{a_1, \ldots, a_m\}$ generates a cone of dimension r (since there are r linearly independent vectors in $\{a_1, \ldots, a_m\}$).

Suppose that $\{a_1,\ldots,a_m^{}\}$ is a set of integer vectors which generates a cone $C \subseteq \mathbb{R}^n$ of dimension r. Relabel the vectors in $\{a_1, \dots, a_m\}$ if necessary so that a_1, \dots, a_k are those vectors a_i in $\{a_1, \dots, a_m\}$ such that $-a_i \in C$. The vectors a_1, \dots, a_k generate a linear space F which is the unique minimal nonempty face of C. Let d be the dimension of F. Select d linearly independent vectors v_1, \dots, v_d from $\{a_1, \dots, a_k\}$ and select r-d vectors v_{d+1}, \dots, v_r from $\{a_{k+1},\ldots,a_m^{}\}$ such that v_1,\ldots,v_r are linearly independent. Let Vbe the matrix with rows v_1, \dots, v_r . Using the algorithm of Kannan and Bachem [79], find a unimodular matrix U such that VU is in Hermite normal form. The last n-d components of each vector in $\{v_1 v_1, \dots, v_d v\}$ are zero, so for each i = 1, ..., k the last n-d components of a_iU are zero. Similarly, the last n-r components of each vector in $\{a_{k+1}^{}U,\ldots,a_{m}^{}U\}$ are zero. Since $\{a_{1}^{},\ldots,a_{m}^{}\}$ is a Hilbert basis if and only if $\{a_1^{\ U}, \dots, a_m^{\ U}\}$ is a Hilbert basis, it can be assumed that $a_i = a_i U$ for i = 1, ..., m.

If w is a vector in F and $\gamma_1 a_1 + \ldots + \gamma_m a_m$ is an expression of w as a nonnegative linear combination of a_1, \ldots, a_m , then $\gamma_{k+1} = \gamma_{k+2} = \ldots = \gamma_m = 0$. Thus, if $\{a_1, \ldots, a_m\}$ is a Hilbert basis for C, then $\{a_1, \ldots, a_k\}$ is a Hilbert basis for F. For $i = k+1, \ldots, n$ let a_i' be the vector consisting of the components d+1 through r of a_i . Since the last n-d components

of each vector in $\{a_1,\ldots,a_k\}$ are zero, if $\{a_1,\ldots,a_m\}$ is a Hilbert basis then $\{a'_{k+1},\ldots,a'_m\}$ is a Hilbert basis. Conversely, it is easy to see that if $\{a_1,\ldots,a_k\}$ is a Hilbert basis for F and $\{a'_{k+1},\ldots,a'_m\}$ is a Hilbert basis, then $\{a_1,\ldots,a_m\}$ is a Hilbert basis. It follows that $\{a_1,\ldots,a_m\}$ is a Hilbert basis if and only if $\{a_1,\ldots,a_k\}$ is a Hilbert basis and $\{a'_{k+1},\ldots,a'_m\}$ is a Hilbert basis.

Since F is a linear space and F is the cone generated by $\{a_1,\ldots,a_k\}$, this set is a Hilbert basis for F if and only if each integer vector in F can be expressed as an integer combination of a_1,\ldots,a_k (which can be checked in polynomial time using the algorithm of Kannan and Bachem [79]). As in the proof of Theorem 7.2, this can be seen as follows. The condition is clearly necessary. For the converse, suppose that the condition holds and that w is an integer vector in F. It must be shown that w can be expressed as a nonnegative integer combination of a_1,\ldots,a_k . Let $\lambda_1a_1+\ldots+\lambda_ka_k$ be an expression of w as an integer combination of a_1,\ldots,a_k . By expressing $-a_1$ as a nonnegative linear combination of a_1,\ldots,a_k (excluding a_1) for $i=1,\ldots,k$ find positive γ_1,\ldots,γ_k such that $0=\gamma_1a_1+\ldots+\gamma_ka_k$. There exists an integer M such that $(\lambda_1+M\gamma_1)a_1+\ldots+(\lambda_k+M\gamma_k)a_k$ is an expression of w as a nonnegative integer combination of a_1,\ldots,a_k .

Since $\{a'_{k+1},\ldots,a'_m\}$ generates a pointed cone of full dimension (which is less than or equal to r), it can be assumed that $\{a_1,\ldots,a_m\}$ generates a pointed cone of full dimension.

Now for the final stage of the proof. The set $\{a_1,\ldots,a_m\}$ is a

Hilbert basis for C if and only if the only integer vector in

(9.1)
$$C_0 = \{w \in C: w-a_i \notin C, i = 1,...,m\}$$

is the zero vector. This can be seen as follows. Suppose that $\{a_1,\ldots,a_m\}$ is a Hilbert basis for C and that w is an integer vector in C_0 . Since w is in C, $w=\lambda_1a_1+\ldots+\lambda_ma_m$ for some nonnegative integers $\lambda_1,\ldots,\lambda_m$. But $w-a_1\not\in C$ for $i=1,\ldots,m$, so $\lambda_1=0$ for $i=1,\ldots,m$ and w=0. Conversely, suppose that the only integer vector in C_0 is 0 and that w is an integer vector in C. Let λ_1 be the greatest rational such that $w-\lambda_1a_1$ is in C (λ_1 exists, since it has been assumed that C is pointed) and let $w^1=w-l\lambda_1|a_1$. Let λ_2 be the greatest rational such that $w^1-\lambda_2a_2$ is in C and let $w^2=w^1-l\lambda_2|a_2$. Continue in this manner until w^m is found. Now w^m is an integer vector in C_0 , so $w^m=0$. So $l\lambda_1|a_1+\ldots+l\lambda_m|a_m$ is an expression of w as a nonnegative integer combination of a_1,\ldots,a_m .

The cone C is a polyhedron of full dimension. The facet-inducing inequalities for C can be found as follows. For each subset of r-l linearly independent vectors from $\mathbf{a}_1,\dots,\mathbf{a}_m$ find the hyperplane H which contains these vectors and the origin. If each vector in $\mathbf{a}_1,\dots,\mathbf{a}_m$ lies in one of the closed half spaces determined by H, then H \cap C is a facet of C. In this way (since r is fixed), vectors $\mathbf{b}_1,\dots,\mathbf{b}_t$ such that $\mathbf{C}=\{\mathbf{x}\colon \mathbf{b}_i\mathbf{x}\geq 0,\ i=1,\dots,t\}$ can be found in polynomial time.

It follows from the definition of C_0 that

(9.2) $C_0 = \{ w \in C: \text{ for each } i = 1, ..., m \text{ there exists } j, \\ 1 \le j \le t, \text{ such that } b_j w < b_j a_j \}.$

Thus, if Φ denotes the collection of all functions ϕ from $\{1,\ldots,m\}$ to $\{1,\ldots,t\}$, then

(9.3)
$$\begin{array}{c} C_0 = U \left\{x \colon b_j x \geq 0 \quad \text{for } j = 1, \ldots, t \quad \text{and} \quad b_{\varphi(i)} x < b_{\varphi(i)}^a i \right. \\ \text{for } i = 1, \ldots, m \right\}. \end{array}$$

Using (9.3) a concise description of C_0 will now be found.

Let $w \in C_0$. By (9.3), there exists a function φ in Φ such that w is contained in

(9.4) $P = \{x: b_i x \ge 0 \text{ for } i = 1, \ldots, t \text{ and } b_{\phi(i)} a_i < b_{\phi(i)} a_i = 1, \ldots, m \}$. Since C_0 is bounded, P is bounded. So the polyhedron \overline{P} obtained by replacing each "<" by " \le " in (9.4) is bounded (P is nonempty since $W \in P$, so \overline{P} is the closure of P). There exists an E > 0 such that (1+E)W is contained in P (and, so, also in \overline{P}). Now since $0 \in \overline{P}$, there exists linearly independent vertices z_1, \ldots, z_r of \overline{P} such that (1+E)W can be expressed as a convex combination of 0 and z_1, \ldots, z_r . So W can be expressed as a convex combination of 0 and z_1, \ldots, z_r with the multiplier of 0 being positive. Since $0, z_1, \ldots, z_r$ are affinely independent, this expression of W as a convex combination of $0, z_1, \ldots, z_r$ is unique. Thus, $W \in \text{convex hull } \{0, z_1, \ldots, z_r\} \setminus \text{convex hull } \{z_1, \ldots, z_r\}$. This observation can be used as follows.

Let Z be the sets of all vectors z determined by r linearly

independent equations from

(9.5)
$$b_{j}z = 0 j = 1,...,t$$
$$b_{j}z = b_{j}a_{i} j = 1,...,t; i = 1,...,m$$

such that z also satisfies $b_jz \ge 0$, $j=1,\ldots,t$. Let S be the set of all subsets $\{z_1,\ldots,z_r\}$ of Z such that z_1,\ldots,z_r are linearly independent and there exists a function $\phi \in \Phi$ such that for every $k=1,\ldots,r$ we have $b_{\phi(i)}z_k \le b_{\phi(i)}a_i$ (since r is fixed, both Z and S can be found in polynomial time). For each $\{z_1,\ldots,z_r\}$ in S, let $\sigma(z_1,\ldots,z_r)=\text{convex hull }\{0,z_1,\ldots,z_1\}\setminus \text{convex hull }\{z_1,\ldots,z_r\}$.

$$\underline{\text{Claim:}} \quad \mathbf{C_0} = \underbrace{\mathbf{U}}_{\{\mathbf{z_1, \dots, z_r}\} \in \mathbf{S}} \quad \sigma(\mathbf{z_1, \dots, z_r}).$$

Once this claim has been established the proof will be complete, since the integer programming algorithm of Lenstra [81] can be used to test for each $\{z_1,\ldots,z_r\}\in S$ whether or not 0 is the only integer vector in $\sigma(z_1,\ldots,z_r)$ in polynomial time. Indeed, since z_1,\ldots,z_r are linearly independent, $z_1,\ldots,z_r,0$ are affinely independent. Thus, if H is a hyperplane which contains $\{z_1,\ldots,z_r\}$, then 0 is not contained in H (recall that the dimension of the space we are working in is r). Let $H=\{x\colon gx=p\}$ be such a hyperplane. It follows that $\sigma(z_1,\ldots,z_r)=\text{convex hull}\{0,z_1,\ldots,z_r\}\setminus H$. It may be assumed that g is integral and that p is a postive integer. Each vector in $\sigma(z_1,\ldots,z_r)$ satisfies gx< p, so each integer point in $\sigma(z_1,\ldots,z_r)$ satisfies gx< p-1. Letting $Bx\le f$ be a defining system for the convex hull of $\{0,z_1,\ldots,z_1\}$, it only needs to be

checked that the only integer vector in $T = \{x : Bx \le f, gx \le p-1\}$ is 0. One way to do this in polynomial time, using Lenstra's integer programming algorithm, is to solve the integer programs $\max\{x_i : x \in T, x \text{ integer}\}$ and $\min\{x_i : x \in T, x \text{ integer}\}$ for $i = 1, \ldots, r$. If 0 is the optimal solution for each of these linear programs, then the only integer vector in T is 0.

The "⊆" inclusion in the claim has already been shown. To see the "2" inclusion, let $w \in \sigma(z_1, ..., z_r)$ for some $\{z_1, ..., z_r\} \in S$. By definition of Z, w is in C. Since the multiplier of 0 in the expression of w as a convex combination of $0, z_1, \ldots, z_r$ is positive, there exists an $\varepsilon > 0$ such that $(1+\varepsilon)w$ is in $\sigma(z_1, \ldots, z_r)$. By the definition of S, there exists a function $\phi \in \Phi$ such that $b_{\phi(i)}^{z_k} \leq b_{\phi(i)}^{a_i}$ for i = 1, ..., m and k = 1, ..., r. Suppose that $b_{\phi(i)}a_i \le 0$ for some i, $1 \le i \le m$. Since $b_i z_k \ge 0$ for all j = 1,...,m and all k = 1,...,r, $b_{\phi(i)}a_i = 0$. So $b_{\phi(i)}z_k = 0$ for k = 1,...,r, which implies that $b_{\phi(i)} = 0$, since $z_1,...,z_r$ are linearly independent. Since this is impossible, it must be the case that $b_{\phi(i)}^{a_i} > 0$ for all i = 1, ..., m. Thus, $(1+\epsilon)b_{\phi(i)}^{w \leq b_{\phi(i)}^{a_i}}$ for i = 1, ..., m (which follows from the fact that $(1+\epsilon)w \in \sigma(z_1, ..., z_r)$) implies that $b_{\phi(i)}^{w < b_{\phi(i)}^{a_i}}$ for i = 1,...,m. So w is in c_0 , which completes the proof of the claim.

An immediate consequence of this theorem is that for any fixed r, there exists a polynomial-time algorithm to test whether or not a set of integer vectors $\{h_1,\ldots,h_k\}$ is a Hilbert basis if the dimension

of the cone generated by $\{h_1,\ldots,h_k\}$ is less than or equal to r. Also, it follows from the second stage of the above proof that if the cone generated by a set of integer vectors $\{h_1,\ldots,h_k\}$ is a linear space, then it is possible to test in polynomial time whether or not $\{h_1,\ldots,h_k\}$ is a Hilbert basis (no assumption on the dimension of the linear space is needed).

It should be noted that Chandrasekaran and Shirali [82] obtained a polynomial-time algorithm to test whether or not $\{Ax = 0, x \ge 0\}$ is totally dual integral, where A is an integral matrix of fixed rank.

As in the case of integer polyhedra, it is an open question whether or not the class of totally dual integral linear systems is in NP. One possible application of an affirmative answer to either this question or the corresponding question for integer polyhedra would be that for any fixed polynomial f the class of perfect graphs with less than or equal to f(n) maximal cliques, where n is the number of nodes in the graph, is in NP (for appropriate choices of f, classes of perfect graphs that contain many well known classes can be obtained in this way). See Burlet [81], Burlet and Fonlupt [82], Burlet and Uhry [82], Cameron [82], and Chvátal [81] for work on recognizing perfect graphs.

A problem related to the recognition of totally dual integral systems will now be considered.

If P is a polyhedron of full dimension then a theorem of Schrijver [81] presented in Section 5 implies that there is a "best possible" min-max theorem for the maximum weight of a vector in P where the minimum is

taken over all subsets a certain set of integer vectors, such that the subset satisfies some condition (such as a covering condition). Is it possible to test whether or not a given integer vector is in the set? In other words, given a polyhedron P of full dimension, is a given inequality in the Schrijver system for P (recall that the Schrijver system is the unique minimal totally dual integral system with integer left hand sides that defines P)? Once again, if P is defined by a linear system $Ax \leq b$ with A a matrix of bounded rank, then this recognition question can be settled in polynomial time.

Theorem 9.5: Let r be a fixed positive integer. There exists a polynomial-time algorithm which, for any linear system $Ax \le b$ with A a matrix of rank r which defines a polyhedron P of full dimension and any inequality $\alpha x \le \beta$, determines whether or not $\alpha x \le \beta$ is in the Schrijver system for P.

<u>Proof:</u> Suppose that r is a fixed positive integer, that $Ax \leq b$ defines a polyhedron P of full dimension in \mathbb{Q}^n and that $\alpha x \leq \beta$ is an inequality. By multiplying the system by a positive constant if necessary, it can be assumed that A is integral. Also, it can be assumed that α is integral since otherwise $\alpha x \leq \beta$ is not in the Schrijver system for P.

If $\max\{\alpha x\colon Ax\le b\}$ is not equal to β , then clearly $\alpha x\le \beta$ is not in the Schrijver system for P (this can be checked in polynomial time with the ellipsoid algorithm). Suppose that this maximum is equal to β . Using the ellipsoid algorithm, find the active rows in A for some minimal face F of P such that $\max\{\alpha x\colon Ax\le b\}$ is achieved by each vector in F. Let a_1,\ldots,a_n be the active rows for F and let C be the pointed cone generated by a_1,\ldots,a_m . Now

 $\alpha x \leq \beta$ is in the Schrijver system for P if and only if α is in the unique minimal Hilbert basis for C.

Let v_1, \dots, v_r be r linearly independent vectors chosen from a_1, \dots, a_m and let V be the matrix with rows v_1, \dots, v_r . Using the algorithm of Kannan and Bachem [79], find a unimodular matrix U such that VU is in Hermite normal form. As in the proof of Theorem 9.4 the last n-r components of each vector in $\{a_1U, \dots, a_mU\}$ are zero. Since α is in the unique minimal Hilbert basis for C if and only if αU is in the unique minimal Hilbert basis for the cone generated by $\{a_1U, \dots, a_mU\}$, it can be assumed that $a_i = a_iU$ for $i = i, \dots, m$ and $\alpha = \alpha U$. Now since the last n-r components of each vector in $\{a_1U, \dots, a_mU\}$ are zero, it can be assumed that C is of full dimension.

Let w be a vector such that wz > 0 for each nonzero z in C (such a vector w can be found in polynomial time). Let $\mathbb{P}^1=\{z\in C\colon wz\leq w\alpha\}$ and let $\mathbb{P}^2=\{\alpha-z\colon z\in \mathbb{P}^1\}$. Now α is in the unique minimal Hilbert basis for C if and only if the only integer vectors in $\mathbb{P}^1\cap\mathbb{P}^2$ are 0 and α . This can be seen as follows. If z^1 is an integer vector in $\mathbb{P}^1\cap\mathbb{P}^2$ other than 0 and α , then z^1 is in C and there exists an integer vector z^2 in C such that $z^1=\alpha-z^2$. So z^1+z^2 is an expression of α as a nonnegative integer combination of other integer vectors in C Conversely, if α is not in the unique minimal Hilbert basis for C then there exist integer vectors z^1 and z^2 in $\mathbb{P}^1\setminus\{0,\alpha\}$ such that $\alpha=z^1+z^2$. So z^1 and z^2 are integer vectors in $\mathbb{P}^1\cap\mathbb{P}^2$ other than 0 and α .

Since r is fixed, with Lenstra's integer programming algorithm it can be checked in polynomial time whether or not the only integer

vectors in $P^1 \cap P^2$ are 0 and α .

It should be noted that, even for linear systems $Ax \le b$ with A a matrix of fixed rank, the number of inequalities in the Schrijver system for the polyhedron defined by $Ax \le b$ is not bounded by a polynomial in the size of $Ax \le b$.

Edmonds, Lovász, and Pulleyblank [82] have shown that given a "proper" encoding of a polyhedron P and an optimization oracle for P, it can be determined in polynomial time whether or not a given inequality $\alpha x \leq \beta$ is facet inducing for P (see also Grötschel, Lovász, and Schrijver [81,82]). Again, it would be interesting if a similar result for minimal totally dual integral defining systems with integer left hand sides could be obtained.

10. Proving total dual integrality with cross-free families.

To this point, very little has been mentioned on the application of total dual integrality to proving combinatorial min-max theorems. Of course, some classic min-max results, such as the König-Egerváry Theorem, follow from the result of Hoffman and Kruskal [56] on total unimodularity given in Section 3, but the use of total unimodularity is somemwhat limited. In this final section of this chapter a well known technique for proving that combinatorially described linear systems are totally dual integral is presented. This technique has been quite successful in proving many combinatorial results which do not follow directly from the Hoffman-Kruskal Theorem. It involves restricting the set of nonzero dual variables by "uncrossing" them and showing that if no dual variables "cross" then the existence of an integer optimal solution to the dual linear program is implied by the Hoffman-Kruskal The technique has evolved over a number of years. of using "uncrossing" to prove min-max theorems appears in Younger [69]. Combining "uncrossing" with the Hoffman-Kruskal Theorem was done by Robertson (see Lovász [76]) and Johnson [75] (cf. Edmonds [70]). Bringing total dual integrality into this scheme for proving min-max results is due to Hoffman [74], Edmonds and Giles [77], and Hoffman and Schwartz [78] and developing a general framework for proving theorems in this way is due to Schrijver [82a].

Some of the many interesting general min-max theorems that have been proven by the above mentioned technique are given in Edmonds and Giles [79], Frank [79,81a], Frank and Tardos [81], Gröflin and Hoffman [82], Hassin [78,82], Hoffman [76a,78], Hoffman and Schwartz [78], Lawler [82], Lawler and Martel [82], and Schrijver [82,83] (see Schrijver [82b] for the interrelations of these models). Algorithmic aspects of these models are dealt with in Frank [81,82], Cunningham and Frank [82], Fujishige [78], Lucchesi [76], Schösleben [80], and Zimmerman [82].

Let V be a finite set and let F be a family of subsets of V. For each $f \in F$ let $a_f x \le b_f$ be a linear inequality, where x is a vector. The linear system

(10.1)
$$a_{f}^{x \leq b_{f}} \quad \forall f \in F$$
$$x \geq 0$$

is totally dual integral if for each integer vector w such that

min
$$\Sigma\{y_f b_f : f \in F\}$$

(10.2) s.t.
$$\Sigma\{y_f^a_f: f \in F\} \ge w$$

$$y_{f} \ge 0 \quad \forall f \in F$$

exists, the minimum can be achieved by an integer vector. It is a simple observation that if for some $F' \subseteq F$ there exists an optimal solution \bar{y} to (10.2) with $\bar{y}_f = 0$ for each $f \in F \setminus F'$, then any vector y' which achieves the minimum in

min
$$\Sigma\{y_fb_f: f \in F'\}$$

(10.3) s.t.
$$\Sigma \{ y_f a_f : f \in F' \} \ge w$$

$$y_f \ge 0 \quad \forall f \in F'$$

can be extended to an optimal solution to (10.2) by setting $y_{\rm f}^*=0$ for each $f\in F\backslash F'$. This observation suggests a method for proving that (10.1) is totally dual integral: For each integral w such that (10.2) exists, find a family $F'\subseteq F$ as above such that (10.3) can be achieved by an integer vector. Such a program will now be described.

The program involves "uncrossing" and will be illustrated with several examples once a lemma which sets up the technique is proven. (The lemma is implicit in the papers cited above.)

Let \mathcal{B} be some binary relation on \mathcal{F} and say that $A,B \in \mathcal{F}$ form a $\underline{\mathcal{B}}$ -intersection if $A \not\subseteq B$, $B \not\subseteq A$, and (A,B) satisfies \mathcal{B} (\mathcal{B} could be that $A \cap B \neq \emptyset$, for instance). Consider the following conditions on (10.1):

- (I) If $A, B \in \mathcal{F}$ form a \mathcal{B} -intersection then
 - (i) $A \cap B \in \mathcal{F}$ and $A \cup B \in \mathcal{F}$
 - (ii) $b_{A \cap B} + b_{A \cup B} \le b_A + b_B$
 - (iii) $a_{A\cap B} + a_{A\cup B} \ge a_A + a_B$. ((iii) is equivalent to $if \ \overline{y} \ge 0 \ then \ \Sigma\{y_f^!a_f^!: f \in F\} \ge \Sigma\{\overline{y}_f^!a_f^!: f \in F\},$ where $y_A^! = \overline{y}_A^!-\varepsilon$, $y_B^! = \overline{y}_B^!-\varepsilon$, $y_{A\cap B}^! = \overline{y}_{A\cap B}^!+\varepsilon$, $y_{A\cup B}^! = \overline{y}_{A\cup B}^!+\varepsilon$, $y_D^! = \overline{y}_D$ for all D in $F\setminus\{A,B,A\cap B,A\cup B\}$, and $\varepsilon = \min\{\overline{y}_A^!,\overline{y}_B\}$.)
- (II) If $F' \subseteq F$ and no two sets in F' form a \mathcal{B} -intersection, then (10.3) has an integer optimal solution for each integer vector \mathbf{w} for which it has an optimal solution. (For example, the coefficient matrix of (10.3) may be totally unimodular.)

Lemma 10.1: If F,B, and (10.1) satisfy (I) and (II), then (10.1) is totally dual integral.

<u>Proof:</u> Let w be an integer vector such that (10.2) exists. For any solution y to (10.2) let $g(y) = \Sigma\{|f| \cdot |V \setminus f| y_f \colon f \in F\}$. Let \overline{y} be an optimal solution to (10.2) and let t be the least common denominator of $\{y_f \colon f \in F\}$. Let $F' = \{f \in F \colon y_f > 0\}$. Suppose that there exists $A,B \in F'$ that form a B-intersection. By I-(i), $A \cup B \in F$ and $A \cap B \in F$. Define y' as in I-(iii). By I-(ii) and I-(iii), y' is an optimal

solution to (10.2). It can be checked that $g(y') \leq g(y)-1/t$, and that t is still a common denominator of $\{y'_f: f \in F\}$. Repeating this procedure, a solution can be found such that no B-intersections occur in F'. Now, by II, there exists an integer optimal solution to (10.3) So, by the observation made above,

(10.2) can be achieved by an integer vector.

Of course, a similar result holds for systems of the form

(10.1)'
$$a_{\mathbf{f}} \mathbf{x} \ge b_{\mathbf{f}} \quad \forall \ \mathbf{f} \in \mathcal{F}.$$

Indeed, letting (I)' be (I) with the " \leq " sign in I-(ii) replaced by " \geq " and with the " \geq " sign in " $\Sigma\{y_f^ia_f\colon f\in F\} \geq \Sigma\{\bar{y}_f^ia_f\colon f\in F\}$ " replaced by " \leq ", it follows that if F, \mathcal{B} , and (10.1)' satisfy (I)' and (II) then (10.1)' is totally dual integral.

Condition I-(ii) indicates that submodular functions fall naturally into theorems proven using Lemma 10.1. To illustrate this, the first example that will be considered is the Matroid Intersection Theorem of Edmonds [70].

Theorem 10.2: Let $M_i = (E, I_i)$ be a matroid for i = 1, 2 and let $r(S) = \min\{r_1(S), r_2(S)\}$ for each $S \subseteq E$, where r_i is the rank function of M_i . If $w = (w_e : e \in E)$ is a nonnegative integer vector then the maximum value of $\Sigma\{w_e : e \in S\}$ where $S \in I_1 \cap I_2$ is equal to the minimum value of $r(A_1) + \ldots + r(A_k)$ where A_1, \ldots, A_k are subsets of E(repetition is permitted) such that each $e \in E$ is in at least w_i of these subsets.

<u>Proof:</u> Letting F be the family of all subsets of E, $b_S = r(S)$ for all $S \subseteq E$, and a_S the 0-1 incidence vector of S for all $S \subseteq E$, the theorem is equivalent (by Corollary 2.6) to the total dual integrality of (10.1). For $A,B\subseteq E$, let B be the property that $A\cap B\neq\emptyset$ and for some $i\in\{1,2\}$, $r(A)=r_i(A)$ and $r(B)=r_i(B)$. It is easy to check that (10.1) satisfies (I). Now suppose that no two sets in $F'\subseteq F$ B-intersect and let $F_i=\{S\subseteq F': r(S)=r_i(S)\}$ for i=1,2. By Theorem 3.5, the incidence matrix of the family $F_1\cup F_2$ is totally unimodular, which implies that the incidence matrix of (10.3) is totally unimodular. So (10.1) satisfies (II) and the result follows from Lemma 10.1.

Trivially, the above also shows the following well known extension. If for i = 1, 2, F_i is a family of subsets of E such that if $A, B \in F_i$ and $A \cap B \neq \emptyset$ then $A \cap B \in F_i$ and $A \cup B \in F_i$ and r_i is a rational valued function on F_i which is submodular on intersecting pairs then the system

(10.4)
$$\Sigma\{x_e : e \in S\} \leq r_1(s) \quad \forall S \in F_1$$
$$\Sigma\{x_e : e \in S\} \leq r_2(s) \quad \forall S \in F_2$$
$$x_e \geq 0 \quad \forall e \in E$$

is totally dual integral.

Several examples involving directed graphs will now be considered. A directed graph is a graph where each edge has one end specified as

the $\underline{\text{head}}$ and the other specified as the $\underline{\text{tail}}$. An edge enters a set S if its head is in S and its tail is not in S.

Let G be a directed graph and let $r \in VG$. A set $B \subseteq EG$ is called a <u>branching</u> rooted at r if B forms a spanning tree and every node in $V\setminus\{r\}$ is the head of eactly one edge in B. Alternatively, a branching can be described as a minimal set of edges such that each non-empty $S \subseteq VG\setminus\{r\}$ is entered by at least one edge in the set. If $w = (w_e \colon e \in EG)$ is a vector, say that the "weight" of a branching B is $\Sigma\{w_e \colon e \in B\}$. Fulkerson [74] proved the following min-max relation.

Theorem 10.3: Let G be a directed graph and let $r \in VG$. Given a nonnegative integral weight vector $w = (w_e : e \in EG)$ the minimum weight of a branching rooted at r is equal to the maximum number of nonempty subsets of $VG\setminus\{r\}$, with repetition allowed, such that every edge e enters no more than w_e of these subsets.

Proof: Letting F be the set of all nonempty subsets of VG\{r}, b_S =1 for all $S \in F$, and $a_S = (a_S) : e \in EG$, where $a_S = 1$ if e enters S and $a_S = 0$ otherwise, for all $S \in F$; all that needs to be shown is that (10.1)' is totally dual integral. For $A, B \in F$, let B be the property that $A \cap B \neq \emptyset$. Again, it is easy to check that (10.1)' satisfies (I)'. Suppose that no two sets in $F' \subseteq F$ B-intersect and let M be the matrix with rows $(a_S: S \in F')$. Since $b_S = 1$ for all $S \in F$, to show (II) it suffices, by Theorem 3.6, to show that M is balanced. Suppose that M is not balanced. There must exist $e_0, \dots, e_{k-1} \in EG$ and $A_0, \dots, A_{k-1} \in F'$ such that for $i = 0, \dots, k-1$ e_i enters A_i and A_{i+1} , but does not enter A_j for $j \neq i$, i+1 (all subscripts

should be taken modulo k), where $k \ge 3$ is odd. Since no B-intersections occur in F', either $A_0 \subseteq A_1$ or $A_1 \subseteq A_0$. Since the ordering is not important, it can be assumed that $A_1 \subseteq A_0$. But this implies that $A_j \subseteq A_{j-1}$ for $j=2,\ldots,k$, a contradiction. So M is balanced and (II) is satisfied.

Actually, the matrix M given above is totally unimodular (cf. Schrijver [83]), but the proof that M is balanced is quite simple and illustrates the use of Theorem 3.6. Again, the above proof also shows the extension due to Frank [79] that, for any directed graph G, if F is a family of subsets VG such that if $A,B \in F$ and $A\cap B \neq \emptyset$, then $A\cap B \in F$ and $A\cup B \in F$, then the linear system

(10.5)
$$\Sigma\{x_e : e \text{ enters } S\} \ge 1 \quad \forall S \in F$$
$$x_e \ge 0 \quad \forall e \in EG$$

is totally dual integral (This is a special case of Frank's result, since he allows the right hand side of (10.5) to be any supermodular function.) This extension includes a theorem of Fulkerson [68] on the minimum length of a directed path between two specified nodes in a directed graph. A related result on the length of a longest directed path in an acylic directed graph is given in Vidyasankar and Younger [75].

As a final example of the use of the "uncrossing" technique, a proof of the Lucchesi-Younger Theorem, on directed cuts, will be given. A directed cut of a directed graph G is a nonempty set of edges that is the set of edges entering some set $S \subseteq VG$ such that no edge enters $VG \setminus S$.

The following result is due to Lucchesi and Younger [78].

Theorem 10.4: Let G be a directed graph. The maximum number of disjoint directed cuts of G is equal to the minimum cardinality of a set of edges that meets every directed cut of G.

Proof: It may be assumed that there is a path, not necessarily directed, between each pair of nodes of G. Let F be the set of all those subsets S of VG such that no edges enters VG\S and at least one edge enters S, let $b_S = 1$ for all $S \in F$, and let $a_S = (a_S : e \in EG)$ where $a_{S} = 1$ if e enters S and $a_{S} = 0$ otherwise, for all $S \in \mathcal{F}$. Again, all that needs to be shown is that (10.1)' is totally dual integral. Let B be the property that $A \cap B \neq \emptyset$ and $A \cup B \neq V$ for $A, B \in F$. As before, it is easily checked that (10.1)' satisfies (I)'. Now suppose that no two sets in $F' \subseteq F$ B-intersect and let M be the matrix with rows (a_S : $S \in F'$). It again suffices to show that M is balanced. Suppose that there exist $a_0, \dots, a_{k-1} \in EG$ and $A_0, \dots, A_{k-1} \in F'$ as in the preceding proof. For each i = 0, ..., k-1, either $A_i \subset A_{i+1}$ $A_i \supset A_{i+1}$. Since k is odd, there exists an i such that $A_i \subseteq A_{i+1} \subseteq A_{i+2}$ or $A_{i} \supset A_{i+1} \supset A_{i+2}$. The ordering of the sets is not important, so it may be assumed that $A_0 \supset A_1 \supset A_2$. Since it cannot be the case that $A_2 \supset A_3 \supset \ldots \supset A_{k-1} \supset A_0$, it may be assumed that $A_2 \subset A_3$. Since the tail of a_1 must be in A_3 , $V\setminus A_3\subseteq A_1\setminus A_2$. But then for all $j=3,\ldots,k-1$, $A_1 \subseteq A_1 \setminus A_2$ or $V \setminus A_1 \subseteq A_1 \setminus A_2$, a contradiction since a_{k-1} must enter only A_0 and A_{k-1} .

As in the proof of Theorem 10.3, it can be shown that the matrix M given above is totally unimodular (see Edmonds and Giles [77]).

For a general framework for proving min-max results using the "uncrossing" technique see Schrijver [82a].

It should be noted that there is class of min-max theorems for directed graphs related to the above min-max results which involve "packing", such as the theorem of Edmonds [72] on the maximum number of disjoint branchings in a directed graph, and which do not seem to fit into the "uncrossing" scheme given above. For other results of this type see Feofiloff and Younger [82], Frank [79], Hoffman and Schwartz [77], and Schrijver [82,82b,83].

CHAPTER III

MINIMAL SYSTEMS FOR MATCHING PROBLEMS

1. Introduction.

A <u>matching</u> in a graph G is a subset of the edges such that each node of G is met by at most one edge in the subset. Fundamental results in matching theory were proven by Tutte [47,52,54]. Tutte's results provide a min-max relation for the cardinality of a largest matching in a graph (see Berge [58]). Edmonds [65] described a linear system which defines the convex hull of the matchings of a graph. In this chapter, "best possible" min-max relations for various matching problems are obtained by finding minimal totally dual integral systems for the relevant convex hulls as outlined in Section 5 of Chapter 2.

In Section 2, a connection between minimal totally dual integral systems and a type of separability is presented, together with a description of the unique minimal totally dual integral defining system for the convex hull of the matchings of a graph due to Cunningham and Marsh [78]. A result of Cook [81] and Pulleyblank [81] which describes the minimal totally dual integral system for b-matchings is presented in Section 3. Sections 4 and 5 contain a description of the minimal totally dual integral systems for simple b-matchings and capacitated b-matchings. The minimal totally dual integral system for triangle-free 2-matchings is described in Section 6.

For completeness, a short proof, due to Schrijver [81a], of a result that characterizes a totally dual integral defining system for the convex hull of the matchings of a graph is presented in Section 2. Also, detailed proofs of results that characterize totally dual integral defining systems for b-matchings, simple b-matchings, and capacitated b-matchings are given, based on the characterization for matchings

and constructions of Tutte [52,54] (see Schrijver [82c]).

Two variations of matchings not dealt with here are matchings on bidirected graphs and T-joins. For min-max results on matchings on bidirected graphs see Edmonds and Johnson [70], Green-Krótki [80], and Schrijver [82c]. For min-max results on T-joins and T-cuts see Edmonds and Johnson [73], Korach [82], Lovász [75], and Seymour [77,79,81].

2. Matchings and separability.

Let G be a graph. A matching M of G will be identified with its incidence vector $\mathbf{x} = (\mathbf{x}_e \colon e \in EG)$, where $\mathbf{x}_e = 1$ if $e \in M$ and $\mathbf{x}_e = 0$ if $e \in EG\backslash M$. The fundamental result in the study of polyhedral aspects of matching theory was proven by Edmonds [65]:

Theorem 2.1: Let G be a graph. A defining system for the convex hull of the matchings of G is

$$x_{e} \geq 0 \qquad \forall e \in EG$$

$$(2.1) \qquad x(\delta(v)) \leq 1 \qquad \forall v \in VG$$

$$x(\delta(s)) \leq \lfloor \frac{1}{2} \rfloor s \rfloor \quad \forall s \subseteq VG, \quad |s| \text{ odd.} \qquad \Box$$

Edmonds [65] proved this result by means of a polynomial-time algorithm for the weighted matching problem, known as the blossom algorithm (the weighted matching problem is: for a given graph G and weight vector $w = (w_e: e \in EG)$, maximize wx over all matchings of G). Other proofs of this result are given in Balinski [72], Hoffman and Oppenheim [78], Lovász [79], Schrijver [81a], and Seymour [79].

Cunningham and Marsh [78] proved the following result which, together with Corollary 2.5 of Chapter 2, implies the result of Edmonds given above.

Theorem 2.2: For each graph G, the linear system (2.1) is totally dual integral.

To prove this theorem it suffices to show that for each graph G and each nonnegative integer vector w the linear program







·			
•			
	•		

$$\min \quad \Sigma\{y_{v} \colon v \in VG\} + \Sigma\{\frac{1}{2}(|S|-1)Y_{S} \colon S \in S\}$$
s.t.
$$y(\Psi(e)) + Y(S(e)) \geq w_{e} \quad \forall e \in EG$$

$$y_{v} \geq 0 \quad \forall v \in VG$$

$$Y_{S} \geq 0 \quad \forall S \in S$$

where $S = \{S \subseteq VG: |S| \ge 3 \text{ and } |S| \text{ is odd}\}$, has an integer optimal solution. The following lemma is a well-known result that follows from Edmonds' blossom algorithm. (The construction given here is due to Pulleyblank [73] and Schrijver and Seymour [77] and will be used in Section 3.)

<u>Lemma 2.3:</u> For each graph G and each vector w, there exists an optimal solution (y,Y) to (2.2) such that if $A,B \in S$ with $A \cap B \neq \emptyset$ and $A \not\subseteq B \not\subseteq A$, then either $Y_A = 0$ or $Y_B = 0$.

Proof: Let G be a graph and w a vector. Let (y,Y) be an optimal solution to (2.2). Suppose that there exist sets A and B in S with $Y_A \geq Y_B > 0$, $A \cap B \neq \emptyset$, and $A \notin B \notin A$. If $|A \cap B|$ is odd, let $\overline{Y}_{A \cap B} = Y_{A \cap B} + Y_B$, $\overline{Y}_{A \cup B} = Y_{A \cup B} + Y_B$, $\overline{Y}_{A} = Y_A - Y_B$, $\overline{Y}_{B} = 0$, $\overline{Y}_{S} = Y_{S}$ for all S in $S \setminus \{A, B, A \cup B, A \cap B\}$, and $\overline{y} = y$. If $|A \cap B|$ is even, let $\overline{Y}_{A \setminus B} = Y_{A \setminus B} + Y_B$, $\overline{Y}_{B \setminus A} = Y_{B \setminus A} + Y_B$, $\overline{Y}_{A} = Y_A - Y_B$, $\overline{Y}_{B} = 0$, and $\overline{Y}_{S} = Y_{S}$ for all S in $S \setminus \{A, B, A \setminus B, B \setminus A\}$ and let $\overline{y}_{V} = y_{V} + Y_{B}$ for all $V \in A \cap B$ and $\overline{y}_{V} = y_{V}$ for all $V \in V \cap A \cap B$. In either case, $(\overline{y}, \overline{Y})$ is an optimal solution to (2.2). Also, following Schrijver and Seymour [77], in either case (2.3) $\sum \{\overline{Y}_{S} \mid S \mid (|VG \setminus S| + 1) : S \in S\} \leq \sum \{Y_{S} \mid S \mid (|VG \setminus S| + 1) : S \in S\} - \frac{1}{t}$

where t is the least common denominator of $\{Y_S: S \in S\}$. Repeating

this procedure, an optimal solution can be found such that no such sets ${\tt A}$ and ${\tt B}$ exist.

A family of sets F is called <u>laminar</u> if $A,B \in F$ and $A\cap B \neq \emptyset$ implies that either $A \subseteq B$ or $B \subseteq A$. Thus, the above lemma states that there always exists an optimal solution (y,Y) to (2.2) such that $\{S \in S\colon Y_S > 0\}$ is laminar.

The following short proof of Theorem 2.2 is due to Schrijver [81a]. It should be noted that the proof makes uses of Theorem 3.1.

<u>Proof of Theorem 2.2</u> (Schrijver [81a]): Suppose that it is not true that for each graph G and each nonnegative integer vector w the linear program (2.2) has an integer optimal solution. Let G and w be a counterexample with $|VG|+|EG|+\Sigma\{w_e\colon e\in EG\}$ as small as possible.

By the choice of G and w, G is connected, $w_e \ge 1$ for all $e \in EG$ and for each $v \in VG$ there exists a maximum weight matching of G which does not meet v. Thus, if (y,Y) is any optimal solution to (2.2) then $y_v = 0$ for each $v \in VG$ (by the complementary slackness theorem). By Lemma 2.3 it can be assumed that $\{S \in S \colon Y_S > 0\}$ is laminar. Let A be a maximal set in $\{S \in S \colon Y_S \text{ is not integer}\}$ (by assumption, Y is nonintegral) with respect to inclusion and let A_1, \ldots, A_k be the maximal sets of $\{S \in S \colon Y_S > 0 \text{ and } S \subset A\}$ with respect to inclusion. Let $e = Y_A - [Y_A]$ and let $\overline{Y}_A = Y_A - e$, $Y_A = Y_A + e$ if A_1 for $i = 1, \ldots, k$, and $\overline{Y}_S = Y_S$ for all S in $S \setminus \{A, A_1, \ldots, A_k\}$. Now (y, \overline{Y}) is a feasible solution to (2.2) with objective value less than that of (y, Y), a contradiction to the optimality of (y, Y).

Other proofs of Theorem 2.2 are given in Hoffman and Oppenheim [78], Schrijver [82c], and Schrijver and Seymour [77]. The result also follows from the observation that Edmonds blossom algorithm finds, for any graph G and any nonnegative integer vector w, an optimal solution to (2.2) such that Y is integral and y is half-integral.

A matching M of G is called <u>perfect</u> if each node in VG is met by an edge in M. If G is connected, then G is called <u>hypomatchable</u> if for each $v \in VG$ the graph obtained by deleting v from G has a perfect matching. Let V' be the set of nodes $v \in VG$ such that either $|N(v)| \geq 3$ or |N(v)| = 2 and $\gamma(N(v)) = \emptyset$ or |N(v)| = 1 and v is a node of a two node connected component of G. The following result was proven by Pulleyblank and Edmonds [74].

Theorem 2.4: The unique minimal defining system (unique up to positive scalar multiplication of the inequalities) for the convex hull of the matchings of G is

$$x_e \ge 0 \quad \forall e \in EG$$

(2.4)
$$x(\delta(v)) \le 1 \ \forall \ v \in V'$$

 $x(\gamma(S)) \le \lfloor \frac{1}{2} |S| \ \exists \ \forall \ S \subseteq VG, \ |S| \ge 3, \ G[S]$ hypomatchable with no cutnode.

A short proof of this result due to L. Lovász can be found in Cornuéjols and Pulleyblank [80a].

Cunningham and Marsh [78] showed that the linear system (2.4) is totally dual integral, which immediately implies the following result.

Theorem 2.5: The unique minimal totally dual integral system with integer left hand sides that defines the convex hull of the matchings of G is (2.4).

The content of this result of Cunningham and Marsh [78] may be stated in terms of separability of graphs. Let k be the cardinality of a largest matching of G. Let E_1 and E_2 be subsets of EG with $E_1 \cup E_2 = EG$ and $E_i \neq \emptyset$, i = 1,2. Let k_i be the cardinality of a largest matching of G contained in E_i , i = 1,2. If $k_1 + k_2 = k$, then (E_1, E_2) is a matching separation of G. Call G matching separable if there exists a matching separation of G and matching nonseparable otherwise. G. Giles proved the following result.

Theorem 2.6: A graph G is matching nonseparable if and only if either G is isomorphic to $K_{1,n}$ for some n or G is hypomatchable with no cutnode.

In order to show the connection of this theorem to the result of Cunningham and Marsh and to obtain a new proof of this theorem, a general result on separability will be proven.

Let E be a finite set and let I be a finite set of nonnegative integer vectors $\mathbf{a} = (\mathbf{a}_e \colon \mathbf{e} \in \mathbf{E})$. Call (E,I) a general independence system if $0 \in \mathbf{I}$ and for each $\mathbf{a} \in \mathbf{I}$ and nonnegative integer $\mathbf{b} \leq \mathbf{a}$ it is the case that $\mathbf{b} \in \mathbf{I}$ (so (E,I) is an independence system if each $\mathbf{a} \in \mathbf{I}$ is 0-1 valued). Suppose that (E,I) has the above property. The rank, $\mathbf{r}(\mathbf{A})$, of a set $\mathbf{A} \subseteq \mathbf{E}$ is the maximum value of $\mathbf{x}(\mathbf{A})$ over all vectors $\mathbf{x} \in \mathbf{I}$. A set $\mathbf{A} \subseteq \mathbf{E}$ is closed if for each $\mathbf{e} \in \mathbf{E} \setminus \mathbf{A} \cdot \mathbf{r}(\mathbf{A} \cup \{\mathbf{e}\}) > \mathbf{r}(\mathbf{A})$. A separation of a set $\mathbf{A} \subseteq \mathbf{E}$ is a pair of nonempty subsets $\mathbf{A}_1, \mathbf{A}_2$ of A such that $\mathbf{A}_1 \cup \mathbf{A}_2 = \mathbf{A}$ and $\mathbf{r}(\mathbf{A}_1) + \mathbf{r}(\mathbf{A}_2) = \mathbf{r}(\mathbf{A})$. If there exists a separation of $\mathbf{A} \subseteq \mathbf{E}$ then A is separable (otherwise A is nonseparable). Let $\mathbf{C}(\mathbf{I})$ denote the convex hull of I.

<u>Lemma 2.7:</u> Let (E,I) be a general independence system and suppose that the linear system

(2.5)
$$x(A) \le r(A) \ \forall \ A \subseteq E, \quad A \neq \emptyset$$
$$x_e \ge 0 \quad \forall \ e \in E,$$

is a totally dual integral defining system for C(I). An inequality $x(A) \le r(A)$ is in the Schrijver system for C(I) if and only if $A \ne \emptyset$ is a nonseparable closed set.

<u>Proof:</u> If $x(A) \le r(A)$ is in the Schrijver system for C(I) then clearly the two conditions hold. Conversely, suppose that $A \ne \emptyset$ is a nonseparable closed set. By assumption, for each integral w the linear program

min
$$\Sigma\{r(A)Y_A: A \subseteq E, A \neq \emptyset\}$$

(2.6) subject to
$$\Sigma\{Y_A: A \subseteq E, e \in A\} \ge w_e \quad \forall e \in E$$

$$Y_A \ge 0 \quad \forall A \subseteq E, A \ne \emptyset$$

has an integral optimal solution. To show that $x(A) \leq r(A)$ is in the Schrijver system for C(I) it suffices to show that for some integer vector w the linear program (2.6) has no integer optimal solution with $Y_A = 0$. Let $w_e = 1$ for each $e \in A$ and let $w_e = 0$ for each $e \in EG\backslash A$. An optimal solution to (2.6) is $\overline{Y}_A = 1$ and $\overline{Y}_B = 0$ for all other $B \subseteq E$, $B \neq \emptyset$, with objective value r(A). Any integer optimal solution to (2.6) with $Y_A = 0$ corresponds to a collection of nonempty sets $A_1, \ldots, A_j \subseteq E$ with $A_i \neq A$, $i = 1, \ldots, j$, $A \subseteq A_1 \cup \ldots \cup A_j$, and $r(A_1) + \ldots + r(A_j) = r(A)$. Since A is closed, j must be at least 2. Since A is nonseparable, j must be equal to 1. So there does

not exist such a solution.

This lemma combined with the Cunningham and Marsh result gives a new proof of the characterization of matching nonseparable graphs given in Theorem 2.5. Conversely, using Lemma 2.7 and Theorem 2.2, Theorem 2.6 implies the result of Cunningham and Marsh. It will be shown later that the Cunningham and Marsh result follows from a result proven in Section 5 of this chapter.

Edmonds[71] proved that if M' = (E',I') is a matroid on the ground set E', with independent sets I' and rank function r', then a totally dual integral defining system for P(M'), the convex hull of the independent sets of M' is

(2.7)
$$x(A) \le r'(A) \quad \forall \quad A \subseteq E', \quad A \neq \emptyset$$
$$x_e \ge 0 \qquad \forall \quad e \in E'.$$

Thus, Lemma 2.6 implies that the Schrijver system for P(M') is (2.7) with an inequality $x(A) \le r'(A)$ only for those $A \subseteq E'$, $A \ne \emptyset$ that are closed and nonseparable in the matroidal sense (note that Cunningham [73] has given an algorithm which may be used to test in polynomial time whether or not A has this property). This result of Edmonds can be found in Pulleyblank[82]. A similar result of Giles [75] for matroid intersection polyhedra follows in the same manner from Lemma 2.7 using a theorem of Edmonds [70] (Theorem 10.2 of Chapter 2). Lemma 2.7 also gives a characterization of the Schrijver system for the convex hull of the independent sets of the "perfect independence systems" of Euler [81].

3. b-Matchings.

Let G be a graph and b = $\{b_v: v \in VG\}$ a positive integer vector. A b-matching of G is a nonnegative integer vector $\mathbf{x} = (\mathbf{x}_e: e \in EG)$ such that $\mathbf{x}(\delta(\mathbf{v})) \leq \mathbf{b}_v$ for each $\mathbf{v} \in VG$ (so a b-matching with $\mathbf{b}_v = 1$ for each $\mathbf{v} \in VG$ is a matching). If $\mathbf{w} = (\mathbf{w}_e: e \in EG)$ is a vector of edge weights then the b-matching problem for G is to find a b-matching of G which maximizes wx. A construction of Tutte [54] reduces b-matching problems to matching problems. Replace each node $\mathbf{v} \in VG$ by the new nodes $\mathbf{v}_1, \mathbf{v}_2, \ldots, \mathbf{v}_b$ and replace each edge $(\mathbf{u}, \mathbf{v}) \in EG$ by the new edges $(\mathbf{u}_i, \mathbf{v}_j)$ with $\mathbf{w}_{(\mathbf{u}_i, \mathbf{v}_j)} = \mathbf{w}_{(\mathbf{u}, \mathbf{v})}, \quad i = 1, \ldots, b_{\mathbf{u}}, \quad j = 1, \ldots, b_{\mathbf{v}}.$ A matching in the new graph corresponds to a b-matching in the original graph and vice versa. As stated in Schrijver [82c], applying Theorem 2.2 to the new graph implies the following result, which is an easy consequence of Edmonds' b-matching algorithm [65] (cf. Pulleyblank [73,80]) and has also been proven by Hoffman and Oppenheim [78] and Schrijver and Seymour [77].

Theorem 3.1: A totally dual integral defining system for the convex hull of the b-matchings of G is

$$x_{e} \ge 0 \qquad \forall e \in EG$$

$$(3.1) \qquad x(\delta(v)) \le b_{v} \qquad \forall v \in VG$$

$$x(\gamma(S)) \le \lfloor \frac{1}{2}b(S) \rfloor \forall S \subseteq VG, |S| \ge 3.$$

Proof: It must be shown that for each integral w, the linear program

$$\min \Sigma \{b_{v}\overline{y}_{v} : v \in VG\} + \Sigma \{\frac{1}{2}b(S)\}\overline{y}_{S} : S \in T\}$$

$$\text{s.t.} \overline{y}(\Psi(e)) + \overline{Y}(T(e)) \ge w_{e} \ \forall \ e \in EG$$

$$\overline{y}_{v} \ge 0 \ \forall \ v \in VG$$

$$\overline{Y}_{S} \ge 0 \ \forall \ S \in T,$$

where $T = \{S \subseteq VG: |S| \ge 3\}$, has an integer optimal solution. Let w be a nonnegative integer vector and let G' be the graph obtained from G by splitting the nodes as described above.

Let (y,Y) be an integer optimal solution to (2.2) relative to G' and $S = \{S \subseteq VG' \colon |S| \ge 3\}$ (such a solution exists by Theorem 2.2). Using the construction given in the proof of Lemma 2.3 it may be assumed that $S' = \{S \in S \colon Y_S > 0\}$ is laminar. Furthermore, it may be assumed that for each $q \in VG'$, $\Sigma\{Y_S \colon q \in S \text{ and } S \in S\} = \max\{0, w_{(q,p)}^{-y} - y_q^{-y} p \}$ for each $(q,p) \in \delta(q)$.

Suppose that for some $v \in VG$, it is not the case that $y_{v_1} = y_{v_j}$ for all $j = 1, \ldots, b_v$. Let $i \in \{1, \ldots, b_v\}$ be such that $y_{v_i} = \min\{y_{v_j}: 1 \le j \le b_v\}$ and let $j \in \{1, \ldots, b_v\}$ be such that $y_{v_i} < y_{v_j}$. Let $\bar{S} = \{S \in S': v_i, v_j \in S\}$. Suppose that there exists a set $T \in S'$ such that $v_j \in T$ and $v_i \notin T$. There must exist a node q adjacent to v_j such that $q \in T$ and $w_{(v_j,q)} > Y(\bar{S}((v_j,q)) + y_{v_j} + y_q)$. Now since $y_{v_i} < y_{v_j}$ and since $w_{(v_i,q)} = w_{(v_j,q)}$, there must also exist a set $S \in S'$ such that $v_i \in S$, $v_j \notin S$, and $q \in S$, a contradiction, since S' is laminar. So no such set T exists. Setting y_{v_j} equal to y_{v_i} , $Y_{S \cup \{v_j\}}$ equal to Y_S for each $S \in S$ with $v_i \in S$, and $y_i \in S$, a new optimal solution (y,Y) is obtained with S' remaining laminar By repeating this procedure, it may be assumed that for all $v \in VG$,

$$y_{v_1} = y_{v_j}$$
 for all $j \in \{1, \dots, b_v\}$.

Using the above argument, it may also be assumed that for each $S \in S' \quad \text{and each} \quad v \in VG, \quad \text{if} \quad v_i \in S \quad \text{for some} \quad i \in \{1,\dots,b_v\}, \quad \text{then} \quad v_i \in S \quad \text{for all} \quad j \in \{1,\dots,b_v\}.$

For each $S \in T$ let $\overline{Y}_S = Y_S$, where $S' = \bigcup \{\{v_1, \dots, v_{b_{\overline{V}}}\}: v \in S\}$ and let $\overline{y}_v = y_v$ for each $v \in VG$. Since (y,Y) is an optimal solution to (2.2), $(\overline{y},\overline{Y})$ is an optimal solution to (3.2).

The b-matching analogue of a hypomatchable graph is a b-critical graph. If G is connected, then G is b-critical if for each $v \in VG$ there exists a b-matching \bar{x} of G such that $\bar{x}(\delta(v)) = b_v - 1$ and $\bar{x}(\delta(u)) = b_u$ for each $u \in VG\setminus\{v\}$. Let V'' be those nodes v in VG such that $b(N(v)) \geq b_v + 2$ or $b(N(v)) = b_v + 1$ and $\gamma(N(v)) = \emptyset$ or v belongs to a two node connected component of G and $b_v = b_u$, where v is the other node of the component. Pulleyblank [73] proved the following characterization of the unique minimal defining system for the convex hull of the b-matchings of G.

Theorem 3.2: The unique minimal defining system (unique up to positive scalar multiples of the inequalities) for the convex hull of the b-matchings of G is

$$x_{e} \geq 0 \qquad \forall \quad e \in EG$$

$$(3.3) \qquad x(\delta(v)) \leq b_{v} \qquad \forall \quad v \in V''$$

$$x(\gamma(S)) \leq \lfloor \frac{1}{2}b(S) \rfloor \quad \forall S \subseteq VG, \quad |S| \geq 3,$$

$$G[S] \quad b\text{-critical with no cutnode} \quad v \quad \text{having} \quad b_{v} = 1.$$

Unlike the matching case, (3.3) is not totally dual integral. This can be seen by considering a triangle with $b_v = 2$ for each node v and weight $w_e = 1$ for each edge e. Such a triangle is an example of a b-bicritical graph. If G is connected, then v = 0 is b-bicritical if for each $v \in V$ G there exists a b-matching v = 0 G such that v = 0 and v = 0 for each $v \in V$ G there exists a b-matching v = 0 for each $v \in V$ G that v = 0 and v = 0 for each $v \in V$ G there exists a b-matching v = 0 for each $v \in V$ G that $v \in V$ G that

Theorem 3.3: The unique minimal totally dual integral system with integer left hand sides that defines the convex hull of the b-matchings of G is (3.3) with the additional inequalities

 $x(\gamma(S)) \le \frac{1}{2}b(S) \quad \forall \ S \subseteq VG, \ |S| \ge 3, \ G[S] \ b-bicritical \ and$ $(3.4) \quad \text{each node} \quad v \in VG \setminus S \quad \text{which is adjacent to a node in } S$ $\text{has} \quad b_{V} \ge 2.$

Since $b_{\mathbf{V}} \ge 2$ for each $\mathbf{v} \in \mathrm{VG}$ if G is b-bicritical, Theorem 3.3 implies the theorem of Cunningham and Marsh [78] (Theorem 2.5 of this chapter). It will be indicated later how Theorem 3.3 follows from a result of Section 5.

With b-matching separability defined analogously to matching separability, Lemma 2.7 and Theorem 3.3 imply the following result.

Theorem 3.4: A graph G is b-matching nonseparable if and only if either G is $K_{1,n}$ for some n and either $n \le 1$ or $b(N(v)) \ge b_v + 1$ where v is the node of degree n or G is b-critical with no cut node v having $b_v = 1$ or G is b-bicritical.

To close this section, a fundamental theorem on b-matchings due to Tutte [47,52] will be presented. This theorem will be used later in this chapter. A b-matching \bar{x} of G is called perfect if $\bar{x}(\delta(v)) = b_v$ for each $v \in VG$. If $S \subseteq VG$, let

- (3.5) $C^{\circ}(S) = \{v \in VG \setminus S : G[\{v\}] \text{ is a connected component of } G[VG \setminus S]\}$ and let
- (3.6) $C^1(S) = \{R \subseteq VG \setminus S : |R| \ge 2, b(R) \text{ is odd and } G[R] \text{ is a connected component of } G[VG \setminus S]\}.$

Tutte's b-matching theorem is as follows:

Theorem 3.5: A graph G has a perfect b-matching if and only if for each $S \subseteq VG$, $b(S) \ge b(C^{\circ}(S)) + |C^{1}(S)|$.

This result follows in a straightforward way from Theorem 3.1 by setting $\mathbf{w}_{_{\mathrm{P}}}$ = 1 for each e ϵ EG.

4. <u>Simple b-matchings</u>.

Throughout this section, let G be a graph, possibly with multiple edges and b = $(b_v: v \in VG)$ a nonnegative integer vector. A simple b-matching of G is a subset M of EG such that each node $v \in VG$ meets at most b, edges in M. A perfect simple b-matching of G (i.e. a simple b-matching which meets each node $\, \, v \, \in \, \, ext{VG} \,\,$ in exactly ${
m b}_{
m v}$ edges) is often called a "b-factor". A simple b-matching will be identified with its 0-1 incidence vector $x = (x_j : j \in EG)$. Given a vector \mathbf{w} = (\mathbf{w}_{e} : \mathbf{e} \in EG) of edge weights, the simple b-matching problem is to maximize wx over all simple b-matchings of G. Tutte [54] described the following construction which reduces a simple b-matching problem to a b-matching problem (cf. Schrijver [82c]). For each edge e = (u,v) of G add vertices u and v to VG and replace e by the edges (u, u_e) , (u_e, v_e) , (v_e, v) (although G may have multiple edges, for simplicity edges will still be referred to as unordered pairs of vertices). Also, for each $e \in EG$ let $b_u = b_v = 1$ and let $w(u,u_0) = w(u_0,v_0) = w(v_0,v) = w_0$ The maximum weight of a

b-matching in the new graph is exactly $\Sigma\{w_e\colon e\in EG\}$ greater than the maximum weight of a simple b-matching of G. As in Schrijver [82c], this construction, together with Theorem 3.1, implies the following theorem, which follows from a result of Edmonds and Johnson [70].

Theorem 4.1: A totally dual integral defining system for the convex hull of the simple b-matchings of G is

$$0 \le x_e \le 1 \qquad \forall e \in EG$$

$$(4.1) \qquad x(\delta(v)) \le b_v \qquad \forall v \in VG$$

$$x(\gamma(S))+x(J) \le \lfloor \frac{1}{2}(b(S)+|J|) \rfloor \quad \forall S \subseteq VG, J \subseteq \delta(S)$$

<u>Proof</u>: It suffices to show that for each graph G and each nonnegative integer vector w the linear program

$$\min \Sigma\{z_{e} \colon e \in EG\} + \Sigma\{b_{v}y_{v} \colon v \in VG\} + \Sigma\{\lfloor \frac{1}{2}(b(S) + |J|)\rfloor Y_{(S,J)} \colon S \subseteq VG, J \subseteq \delta(S)\}$$

$$s.t. \ z_{e} + y(\Psi(e)) + \Sigma\{Y_{(S,J)} \colon S \subseteq VG, J \subseteq \delta(S), e \in \gamma(S) \cup J\} \ge w_{e}$$

$$\forall e \in EG$$

$$z_{e} \ge 0 \ \forall e \in EG$$

$$y_{v} \ge 0 \ \forall v \in VG$$

$$Y_{(S,J)} \ge 0 \ \forall S \subseteq VG, J \subseteq \delta(S)$$

has an integer optimal solution. Suppose that this is not true and let G and W be a counterexample with $\Sigma\{w_e\colon e\in EG\}$ as small as possible. Let G' be the graph obtained by the above construction of Tutte and let (\bar{y},\bar{Y}) be an integer optimal solution to (3.2) relative to G' and the vector obtained from W by the construction. Let K be the optimal objective value of (4.2).

If $u \in VG$ and $e = (u,v) \in EG$ then it may be assumed that $u \in S$ for each $S \subseteq VG'$ with $\overline{Y}_S > 0$ and $u \in S$. Furthermore, if $e = (u,v) \in EG$ and there exists a set $S \subseteq VG'$ with $\overline{Y}_S > 0$, $u \in S$ and $v \notin S$, then it may be assumed that $v \in S$.

Suppose that there exists a set $S \subseteq VG'$ with $\overline{Y}_S > 0$. Let $w_e' = w_e - 1$ for all $e = (u,v) \in EG$ such that either $u_e \in S$ or $v_e \in S$ (or both). Let $\overline{Y}_S' = \overline{Y}_S - 1$ and $\overline{Y}_T' = \overline{Y}_T$ for all $T \in T \setminus \{S\}$ and let $\overline{y}_V' = \overline{y}_V - 1$ for each edge $e = (u,v) \in EG$ with $u_e \in S$ and $v_e \notin S$ and let $\overline{y}_Q' = \overline{y}_Q$ for all other $q \in VG'$. Also, let $\overline{S} = \{v \in VG: v \in S\}$ and let $\overline{J} = \{e = (u,v) \in EG: u_e \in S \text{ and } v_e \notin S\}$. There exists, by assumption, an integer optimal solution u(z,y,Y) to u(z,y) relative to u(z,y) and u(z,y) with objective value

$$\Sigma\{\mathbf{w}_{\mathbf{e}} : \mathbf{e} \in \mathbf{EG}\} + \mathbf{k} - \lfloor \frac{1}{2}(\mathbf{b}(\overline{\mathbf{S}}) + |\overline{\mathbf{J}}|) \rfloor - |\{\mathbf{e} = (\mathbf{u}, \mathbf{v}) \in \mathbf{EG} : \mathbf{either}\}\}$$

$$\mathbf{u}_{\mathbf{e}} \in \mathbf{S} \quad \text{or} \quad \mathbf{v}_{\mathbf{e}} \in \mathbf{S}\} .$$

Since $\Sigma\{w_e^i\colon e\in EG\}=\Sigma\{w_e\colon e\in EG\}-\big|\{e=(u,v)\in EG\colon either\ u_e\in S\}$ or $v_e\in S\}\big|$, (z,y,Y) gives objective value $k-\lfloor\frac{1}{2}b(\overline{S})+\big|\overline{J}\big|$]. So (z,y,Y') is an integer optimal solution to (4.2) relative to G and W, where $Y'(\overline{S},\overline{J})=Y(\overline{S},\overline{J}')^{+1}$ $(\overline{J}'=\overline{J}\cap\delta(\overline{S}),$ since it may be the case that \overline{J} contains exactly one edge not in $\delta(\overline{S})$ and Y'(A,B)=Y(A,B) for all other $A\subseteq VG$, $B\subseteq \delta(A)$, a contradiction. So it may be assumed that no such set S exists.

Suppose that there exists a node $v \in VG$ with $\overline{y}_v > 0$. It may be assumed that for each $e = (u,v) \in \delta_G(v)$ with $w_e > 0$, we have $\overline{y}_u > 0$. Let $w_e' = \max\{0, w_e-1\}$ for each $e \in \delta_G(v)$ and $w_e' = w_e$ for all other edges $e \in EG$ and let (z,y,Y) be an integer optimal solution to (4.2)

relative to G and w'. Let $\overline{y}_v' = \overline{y}_v - 1$, $\overline{y}_u' = \overline{y}_u - 1$ for each e = (u,v) in $\delta_G(v)$ with $w_e > 0$, and $\overline{y}_q' = \overline{y}_q$ for all other $q \in VG'$. Since $(\overline{y}', \overline{Y})$ is an optimal solution to (3.2) relative to G' and w', (z,y,Y) gives objective value $k-b_v$ for (4.2). So (z,y',Y) is an integer optimal solution to (4.2), where $y_v' = y_v + 1$ and $y_q' = y_q$ for all $q \in VG\setminus\{v\}$, a contradiction. So it may be assumed that no such node exists.

Clearly, $\Sigma\{w_e\colon e\in EG\}>0$. So there exists an edge $e=(u,v)\in EG$ with $\overline{y}_u > 0$ and $\overline{y}_v > 0$. Let $w_e' = w_e - 1$ and $w_f' = w_f$ for all $f\in EG\setminus\{e\}$. An integer optimal solution to (4.2) relative to G and G can be found by taking an integer optimal solution (z,y,Y) to (4.2) relative to G and G and increasing the value of G by 1, a contradiction.

If H is a subgraph of G, then for each $v \in VH$ let $b_v^H = \min\{b_v, d_H(v)\}$. The largest simple b-matching of G is of cardinality at most $\lfloor \frac{1}{2} b^G(VG) \rfloor$. Let H be the set of all connected subgraphs of G with $|VH| \geq 3$. Theorem 4.1 implies that

$$0 \le x_{e} \le 1 \qquad \forall e \in EG$$

$$(4.4) \qquad x(\delta_{G}(v)) \le b_{v} \qquad \forall v \in VG$$

$$x(EH) \le \lfloor \frac{1}{2}b^{H}(VH) \rfloor \quad \forall H \in H.$$

is a totally dual integral defining system for S(G,b), the convex hull of the simple b-matchings of G. By a series of results in this section and in Section 5, the unique minimal subset of these inequalities which

is a totally dual integral defining system for S(G,b) will be characterized, i.e. the Schrijver system for S(G,b) will be obtained.

As in the previous sections, critical graphs play an important role here. If G is connected, then G is simple b-critical if $|VG| \ge 3$ and for each $v \in VG$ there exists a simple b-matching of G of cardinality $\lfloor \frac{1}{2} b^G(VG) \rfloor$ which contains exactly $b_V^G - 1$ edges which meet v. Define simple b-separability in a way analogous to matching separability.

<u>Lemma 4.2:</u> Let H be a connected subgraph of G with $|VH| \ge 3$. If the inequality

(4.5)
$$x(EH) \le \frac{1}{2}b^{H}(VH)$$

is not $x(\delta(v)) \leq b_v$ for some $v \in VG$, then it is essential for S(G,b) only if H is simple b-critical and simple b-nonseparable and there does not exist an edge $(u,v) \in EG \setminus EH$ with $u,v \in VH$ and $d_H(u) \geq b_v$ and $d_H(v) \geq b_v$.

<u>Proof:</u> Suppose that the inequality (4.5) is essential for S(G,b) and that it is not $x(\delta(v)) \leq b_v$ for some $v \in VG$. Clearly, H is simple b-nonseparable and there does not exist an edge $(u,v) \in EG \setminus EH$ with $d_H(u) \geq b_u$ and $d_H(v) \geq b_v$. Let M be the set of simple b-matchings of G for which (4.5) holds as an equality. Since (4.5) is not $x(\delta(v)) \leq b_v$ for some $v \in VG$, for each $v \in VG$ there exist a simple b-matching in M which contains at most b_v -1 edges which meet v. Similarly, since (4.5) is not $x \leq 1$ for some $e \in EG$, for each $e \in EG$ there exists a simple b-matching in M which does not contain e. So

for each $v \in VH$ there exists a simple b-matching of H of cardinality $\lfloor \frac{1}{2} b^H(VH) \rfloor$ which contains exactly $b_v^H - 1$ edges which meet v. So H is simple b-critical.

To prove that an inequality in (4.4) is essential for S(G,b), it suffices to find a vector \bar{x} which does not satisfy that inequality but does satisfy each of the other inequalities in

- (i) $0 \le x_2 \le 1$ $\forall e \in EG$
- (4.6) (ii) $x(\delta(v)) \le b_v \quad \forall v \in VG$
 - (iii) $x(EH) \leq \left[\frac{1}{2}b^{\frac{H}{2}}(VH)\right]$ for each simple b-critical subgraph H of G.

This technique will be used to characterize which inequalities in (4.6)-(i) and (4.6)-(ii) are essential for S(G,b).

<u>Lemma 4.3:</u> For each $e \in EG$, $x_e \ge 0$ is an essential inequality for S(G,b).

<u>Proof:</u> Let $e \in EG$ and let $x_e = -1$ and $x_f = 0$ for each $f \in EG\setminus\{e\}$. The vector x does not satisfy $x_e \ge 0$ but it does satisfy each of the other inequalities in (4.6).

<u>Lemma 4.4:</u> Let $e \in EG$. The inequality $x_e \le 1$ is essential for S(G,b) if and only if e does not meet a node v with $b_v = 1$ and $d_G(v) > 1$.

<u>Proof:</u> If e meets a node v with $b_v = 1$ and $d_G(v) > 1$, then $x(\delta(v)) \le 1$ implies that $x_e \le 1$, which implies that $x_e \le 1$ is not

an essential inequality for S(G,b). Suppose that e does not meet such a node v. Let $x_e = 2$ and $x_f = 0$ for each $f \in EG \setminus \{e\}$. The vector x does not satisfy $x_e \le 1$ but it does satisfy each of the other inequalities in (4.6).

Lemma 4.5: Let $v \in VG$ and let $b_u' = \min\{b_u, |\delta(u) \cap \delta(v)|\}$ for each node u in N(v). The inequality $x(\delta(v)) \leq b_v$ is essential for S(G,b) if and only if one of the following conditions hold:

- (4.7) a) v belongs to a connected component of G containing exactly one other node u and b) u has the property that $b_v = b_u$ and c) if $d_G(v) \le b_v$ then $b_v = 1$.
- (4.8) a) $b'(N(v)) = b_v + 1$ and b) there is no edge $(v_1, v_2) \in \gamma(N(v))$ such that $b_v = b'_1$ and $b_v = b'_2$.
- (4.9) $b'(N(v)) \ge b_v + 2$

Proof: Suppose that $b'(N(v)) \leq b_v$ and that (4.7) does not hold. The inequality $x(\delta(v)) \leq b_v$ is implied by the inequalities $x_e \leq 1$ for each $e \in \delta(v)$ and $x(\delta(u)) \leq b_u$ for each $u \in N(v)$. So $x(\delta(v)) \leq b_v$ is not an essential inequality for S(G,b). Now suppose that (4.7) holds. Let $x_e = 1$ for each $e \in \delta(v)$ (if $d_G(v) = 1$ let $x_e = 2$ for $e \in \delta(v)$) and let $x_e = 0$ for each $e \in EG \setminus \delta(v)$. The vector x does not satisfy $x(\delta(v)) \leq b_v$ but it does satisfy each of the other inequalities in (4.6). So $x(\delta(v)) \leq b_v$ is essential for S(G,b).

Suppose that $b'(N(v)) = b_v + 1$ and that (4.8) does not hold. Let $(v_1, v_2) \in \gamma(N(v))$ be an edge such that $b_v = b_v'$ and $b_v = b_v'$. Now $x(\delta(v)) + x_{(v_1, v_2)} \le b_v$ is a valid inequality for S(G, b) which implies $x(\delta(v)) \le b_v$. So $x(\delta(v)) \le b_v$ is not an essential inequality for S(G, b). Now suppose that (4.8) holds. For each $u \in N(v)$ select b_u' edges from $|\delta(u) \cap \delta(v)|$ and let J be the collection of these edges. Let $x_e = 1$ for each $e \in J$ and let $x_e = 0$ for each $e \in EG \setminus J$. The vector x does not satisfy $x(\delta(v)) \le b_v$ but it does satisfy each of the other inequalities in (4.6). So $x(\delta(v)) \le b_v$ is essential for S(G, b).

Suppose that $b'(N(v)) \ge b_v + 2$. Let J be a collection of $b_v + 2$ edges in $\delta(v)$ such that $|J \cap \delta(u)| \le b_u'$ for each $u \in N(v)$. Let $x_e = b_v/(b_v + 1)$ for each $e \in J$ and let $x_e = 0$ for each $e \in EG \setminus J$. The vector x does not satisfy $x(\delta(v)) \le b_v$ but it does satisfy all other inequalities in (4.6) - (i) and (4.6) - (ii). Let $ax \le \alpha$ be an inequality in (4.6) - (iii). Let S be a proper subset of J. If $|S| \le b_v$, then $x(S) \le |S|$ and S is a simple b-matching of G. If $|S| = b_v + 1$, then $x(S) = b_v$ and G contains a simple G and consequently G must be equal to 1 for each G and consequently G must be at least G and G is essential for G and G is each G and G and G is each G and G and G is each G and G and G and G and G and G are each G and G and G and G and

The above lemmas imply that a defining system for S(G,b) is

- (i) $x_{e} \ge 0$ $\forall e \in EG$
- (ii) $x_e \le 1 \quad \forall e \in EG$ such that e does not meet a node v with $b_v = 1$ and $d_G(v) > 1$.
- (iii) $x(\delta(v)) \le b_v \quad \forall v \quad VG \quad \text{for which either (4.7), (4.8)}$ (4.10) or (4.9) holds for v.
 - (iv) $x(EH) \leq \lfloor \frac{1}{2} b^H(VH) \rfloor$ for all simple b-critical simple b-nonseparable subgraphs H of G such that there does not exist an edge $(u,v) \in EG \setminus EH$ with $u,v \in VH$ and $d_H(u) \geq b_H$ and $d_H(v) \geq b_V$.

This linear system is not in general totally dual integral (consider the complete graph on four nodes with $b_i = 2$ for each node i and each edge e receiving weight 1). Thus the Schrijver system for S(G,b) is not identical to the minimal defining system for S(G,b) scaled so that the left hand sides are 0-1 valued.

Remark 4.6: Using a generalization of Lovász' [72a] ear-decomposition of factor-critical graphs (see also Cornuéjols and Pulleyblank [81]), it can be proven that (4.10) is the unique minimal defining system for S(G,b), up to positive scalar multiples of the inequalities. This result will appear in Cook and Pulleyblank [83], but it will not be used here.

To obtain the Schrijver system for S(G,b) the notion of a bicritical graph is needed. The graph G is <u>simple b-bicritical</u> if it is connected, $|VG| \geq 3$, $b^G(VG)$ is even, and for each $v \in VG$ there exists a b-matching of G of cardinality $\frac{1}{2}b^G(VG)-1$ which contains exactly $b^G_{\mathbf{v}}-2$ edges

which meet v. The complete graph on four nodes with $b_i = 2$ for each node i is an example of a simple b-bicritical graph.

A variation of Tutte's b-matching Theorem (Theorem 3.5) will be used to show the relation of simple b-bicritical graphs to totally dual integral systems. The transformation of Tutte [54] mentioned earlier reduces any simple b-matching problem to a b-matching problem. So, as presented in Schrijver [82c], to determine if G has a perfect simple b-matching, this transformation can be applied to G to obtain a new graph G' and then Tutte's b-matching Theorem can be applied to G'. Suppose that $S \subseteq VG$ and $T \subseteq VG \setminus S$. Let

(4.11)
$$Q(S,T) = \sum \{b_{v} - d_{G[VG \setminus S]}(v) : v \in T\}$$

and let $G^T(VG \setminus S)$ be the graph obtained from $G[VG \setminus S]$ by taking each node $v \in T$ and splitting it into $d_{G[VG \setminus S]}(v)$ nodes, each with $b_i = 1$ (i.e. replace v by nodes v_1, \ldots, v_k , where $k = d_{G[VG \setminus S]}(v)$, and replace the edges (u_1, v) , (u_2, v) ,..., (u_k, v) by the edges (u_i, v_i) , $i = 1, \ldots, k$, and let $b_{v_i} = 1$, $i = 1, \ldots, k$). Let $\mathcal{D}_1(S, T)$ denote the set of odd connected components of $G^T(VG \setminus S)$ which contain at least two nodes. (A connected component G_i of $G^T(VG \setminus S)$ is odd if $b(VG_i)$ is odd.) Notice that each connected component of $G^T(VG \setminus S)$ corresponds to a subgraph of $G[VG \setminus S]$. Using Tutte's b-matching Theorem, the following result can be proven.

Theorem 4.7: There exists a perfect simple b-matching of G if and only if $\forall S \subseteq VG$ and $\forall T \subseteq VG \setminus S$: $b(S) \not \supseteq Q(S,T) + |\mathcal{D}_1(S,T)|$.

<u>Proof:</u> Suppose that G has a perfect simple b-matching M and let $S \subseteq VG$ and $T \subseteq VG \setminus S$. Let $M' = M \cap \gamma(VG \setminus S)$. Since M' corresponds to a simple b-matching of $G^T(VG \setminus S)$ of cardinality |M'|, we have $b(VG^T(VG \setminus S)) - 2|M'| \ge |\mathcal{D}_1(S,T)|$. Since $b(VG^T(VG \setminus S)) = b(VG \setminus S) - Q(S,T)$, this implies that $b(VG \setminus S) - 2|M'|$ must be at least $Q(S,T) + |\mathcal{D}_1(S,T)|$. Now since M is a perfect simple b-matching of G, $b(S) \ge b(VG \setminus S) - 2|M'| \ge Q(S,T) + |\mathcal{D}_1(S,T)|$.

Conversely, suppose that G does not have a perfect simple b-matching. Let G' be the graph obtained from G by replacing each edge $e = (u,v) \in EG$ by the edges (u,u,), (u,v,), (v,v) and adding u and v to VG with $b_u = b_v = 1$. Since G does not have a perfect simple b-matching, G' does not have a perfect b-matching. So, by Tutte's b-matching Theorem, there exists a set $X \subseteq VG'$ such that $b(X) < b(U\{R: R \in C^{\circ}(X)\}) + |C^{1}(X)|$. Let X be such a subset of VG' and let $S = \{v \in VG: v \in X\}$. It may be assumed that for each edge $e \in (u,v) \in EG$, if $u \in S$ and $v \notin S$ then $v_e \in X$ and $u_e \notin X$ It may also be assumed that for each edge $e = (u,v) \in EG$, if $u \in S$ and $v \in S$ then neither u nor v is in X. Furthermore, it may be assumed that for each edge $e = (u,v) \in EG$ $u \notin S$ and $v \notin S$ then $u_e \in X$ only if $u_f \in X$ for each edge f = (u,t) such that $t \in VG \setminus S$ Let $T = \{v \in VG \setminus S: v \in C^{\circ}(X)\}$, i.e. T is the set of nodes $v \in VG\backslash S$ that are isolated in G'[VG'\X]. Since $b(X) < b(C^{\circ}(X)) + |C^{1}(X)|$, we have $b(S) < Q(S,T) + |\mathcal{D}_{1}(S,T)|$

Corollary 4.8: If G is simple b-bicritical then G has a perfect simple b^G -matching.

Proof: Suppose G is simple b-bicritical. Let $S \subseteq VG$ and $T \subseteq VG \setminus S$. By Theorem 4.7, it suffices to show that $b^G(S) \geq Q^G(S,T) + |\mathcal{D}_1^G(S,T)|$. Suppose that $S \neq \emptyset$ and let $v \in S$. Let $b_v' = b_v^G - 2$ and $b_u' = b_u^G$ for all $u \in VG \setminus \{v\}$. Since G has a perfect simple b'-matching, $b'(S) \geq Q'(S,T) + |\mathcal{D}_1'(S,T)|$. Now $b^G(S) = b'(S) + 2$, $Q'(S,T) = Q^G(S,T)$, and $\mathcal{D}_1'(S,T) = \mathcal{D}_1^G(S,T)$, so $b^G(S) \geq Q^G(S,T) + |\mathcal{D}_1^G(S,T)|$. Suppose that $S = \emptyset$. If T = VG then $Q^G(S,T) \leq 0$ and $|\mathcal{D}_1^G(S,T)| = 0$, which implies that $b^G(S) \geq Q^G(S,T) + |\mathcal{D}_1^G(S,T)|$. Suppose that $VG \setminus T \neq \emptyset$ and let $v \in VG \setminus T$. Define b' as above. Again, Theorem 4.7 implies that $b'(S) \geq Q'(S,T) + |\mathcal{D}_1'(S,T)|$. Since $b'(S) = b^G(S)$, $Q'(S,T) = Q^G(S,T)$, and $\mathcal{D}_1'(S,T) = \mathcal{D}_1^G(S,T)$, $b^G(S) \geq Q^G(S,T) + |\mathcal{D}_1^G(S,T)|$

Theorem 4.7 will be used in Section 5 to prove the following lemma.

Lemma 4.9: Let H be a connected subgraph of G with $|VH| \ge 3$. The inequality $x(EH) \le l\frac{1}{2}b^H(VH)$ is in the Schrijver system for S(G,b) only if either it is in (4.6)-(ii) or H is a simple b-critical simple b-nonseparable graph or a simple b-bicritical simple b-nonseparable graph.

Using Corollary 4.8 and Lemma 4.9, the Schrijver system for S(G,b) may now be characterized.

Theorem 4.10: The unique minimal totally dual integral system with

integer left hand sides which defines the convex hull of the simple b-matchings of G is (4.10) together with

(4.12) $x(EH) \le \frac{1}{2}b^{H}(VH)$ for each connected component of G which is a simple b-nonseparable simple b-bicritical graph.

<u>Proof:</u> Since (4.4) is a totally dual integral defining system for S(G,b), an inequality is in the Schrijver system for S(G,b) only if it is in (4.4). It follows from the proofs of Lemma 4.4 and Lemma 4.5 that an inequality in (4.6)-(i) or (4.6)-(ii) is in the Schrijver system for S(G,b) only if it is in (4.10)-(i), (4.10)-(ii), or (4.10)-(iii). Furthermore, by Lemma 4.3, Lemma 4.4, and Lemma 4.5, each inequality in (4.10)-(i), (4.10)-(ii), and (4.10)-(iii) is essential for S(G,b) and hence is in the Schrijver system for S(G,b).

Let H be a connected subgraph of G with $|VH| \ge 3$. Suppose that $b^H(VH)$ is odd. By Lemma 4.9,

(4.13)
$$x(EH) \leq \lfloor \frac{1}{2}b^{H}(VH) \rfloor$$

is in the Schrijver system for S(G,b) only if H is simple b-critical simple b-nonseparable. If there exists an edge $(u,v) \in EG \setminus EH$ with $d_H(u) \ge b_u$ and $d_H(v) \ge b_v$ then $x(EH) + x_{(u,v)} \le [\frac{1}{2}b^H(VH)]$ is a valid inequality for S(G,b). So (4.13) is in the Schrijver system for S(G,b) only if it is in (4.10) - (iv). Conversely, Lemma 2.7 implies that if (4.13) is in (4.10) - (iv) then (4.13) is in the Schrijver system for S(G,b).

Now suppose that $b^{H}(VH)$ is even. By Lemma 4.9, (4.13) is in the Schrijver system for S(G,b) only if H is simple b-bicritical simple b-nonseparable so suppose that H is simple b-bicritical simple

b-nonseparable. Corollary 4.8 implies that H has a perfect simple b-matching. It follows that if there exists a node $v \in VH$ for which $d_H(v) \leq b_V$, then $(e,EH\setminus\{e\})$ is a simple b-separation of H, where e is an edge in EH that meets v. So $d_H(v) > b_V$ for each $v \in VH$. If H is not a connected component of G then there exists an edge $e \in EG\setminus EH$ which meets a node in VH. If e is such an edge then $x(EH)+x_e \leq \lfloor \frac{1}{2}b^H(VH) \rfloor$ is a valid inequality for S(G,b). So (4.13) is in the Schrijver system for S(G,b) only if it is in (4.12). Conversely, Lemma 2.7 implies that each inequality in (4.12) is in the Schrijver system for S(G,b).

5. Capacitated b-matchings.

Capacitated b-matchings are a generalization of simple b-matchings and b-matchings. Let G be a graph, possibly with multiple edges, b = (b_v: v \in VG) a nonnegative integer vector, and c = (c_e: e \in EG) a positive integer vector of edge capacities. A c-capacitated b-matching (b,c)-matching) of G is a b-matching x such that $x_e \leq c_e$ for each e \in EG. If c_e = 1 for all e \in EG then a (b,c)-matching of G is a simple b-matching of G. If $\beta = \max\{b_v: v \in VG\}$ and $c_e = \beta$ for all e \in EG then x is a (b,c)-matching of G if and only if x is a b-matching of G.

It is easy to reduce a capacitated b-matching to a simple b-matching by replacing each edge $e \in EG$ with c_e edges each of which has the same end nodes as e. In fact, Tutte's construction for reducing simple b-matching problems to b-matching problems can be used to reduce a capacitated b-matching problem immediately to a b-matching problem: For each edge e = (u,v) of G add vertices u_e and v_e to v_e and replace e by the edges (u,v_e) , (v_e,v_e) , (v_e,v_e) . For each $e \in EG$ let v_e v_e v

Theorem 5.1: A totally dual integral defining system for the convex hull of the (b,c)-matchings of G is

$$0 \le x_{e} \le c_{e} \qquad \forall e \in EG$$

$$x(\delta(v)) \le b_{v} \qquad \forall v \in VG$$

$$x(\delta(s))+x(J) \le \left[\frac{1}{2}(b(s)+c(J))\right] \forall S \subseteq VG, J \subseteq \delta(s).$$

<u>Proof:</u> Follow the proof of Theorem 4.1, using the above construction of Tutte.

If H is a subgraph of G, let $b_v^{(H,c)} = \min\{b_v, \Sigma\{c_e : e \in EH \cap \delta(v)\}\}$ for each $v \in VH$. The largest (b,c)-matching of G is of cardinality at most $\lfloor \frac{1}{2}b^{(G,c)}(VG) \rfloor$. Let P(G,b,c) denote the convex hull of the (b,c)-matchings of G.

Corollary 5.2: A totally dual integral defining system for P(G,b,c) is

- (i) $0 \le x_e \le c_e \quad \forall e \in EG$ (ii) $x(\delta(v)) \le b_v \quad \forall v \in VG$ (5.2) (iii) $x(EH) \le \lfloor \frac{1}{2}b^{(H,c)}(VH) \rfloor$ for all connected subgraphs H of G with $|VH| \ge 3$.
- If G is connected then G is (b,c)-critical if $|VG| \ge 3$ and for each $v \in VG$ there exists a (b,c)-matching \bar{x} of G such that $\bar{x}(\delta(v)) = b_v^{(G,c)} 1$ and $\bar{x}(\delta(u)) = b_u^{(G,c)}$ for each $u \in VG \setminus \{v\}$. Let (b,c)-separability be defined in a way analogous to matching separability. Using the technique used in proving Lemma 4.2 the following result can be proven.

<u>Lemma 5.3:</u> Let H be a connected subgraph of G with $|VH| \ge 3$.

If the inequality

(5.3)
$$x(EH)^{-1} \le \lfloor \frac{1}{2}b^{(H,c)}(VH) \rfloor$$

is not in (5.2)-(ii) then it is essential for P(G,b,c) only if H is (b,c)-critical and (b,c)-nonseparable and there does not exist any edge $(u,v) \in EG\setminus EH$ with $v,u \in VH$ and with $b_u^{(H,c)} = b_u$ and $b_v^{(H,c)} = b_v$.

Let G' be the graph obtained from G by replacing each edge $e \in EG$ with edges e_1, \dots, e_c , each of which has the same end nodes as e (i.e. $\Psi(e_i) = \Psi(e)$ for $i = 1, \dots, c_e$). Each subgraph H of G corresponds to a subgraph H' of G' (VH' = VH and EH' = U{\{e_1, \dots, e_c\}: e \in EH\}}). The following lemma will be used to characterize which inequalities in (5.2)-(ii) are essential for P(G,b,c).

Lemma 5.4: Let H be a subgraph of G. If $x(EH') \le \alpha$ is an essential inequality for S(G,b) then $x(EH) \le \alpha$ is an essential inequality for P(G,b,c).

Proof: Suppose that $x(EH') \le \alpha$ is essential for S(G,b). Let M' be a collection of |EG'| affinely independent simple b-matchings of G' for which $x(EH') \le \alpha$ holds as an equality. For each simple b-matching x' in M', let x be the (b,c)-matching of G obtained by letting $x = \sum \{x' : i = 1, \ldots, c'\}$ for each $x \in EG$. Consider the set of (b,c)-matchings $M = \{x : x' \in M'\}$. Each (b,c)-matching in M satisfies $x(EH) \le \alpha$ with equality. Furthermore, since M' is an affinely independent set of vectors, M contains |EG| affinely independent vectors. So $x(EH) \le \alpha$ is essential for P(G,b,c).

The inequalities in (5.2)-(i) and (5.2)-(ii) that are essential for P(G,b,c) are characterized in the following three lemmas.

<u>Lemma 5.5:</u> For each $e \in EG$, $x_e \ge 0$ is an essential inequality for P(G,b,c).

<u>Proof:</u> Let $e \in EG$ and let $x_e = -1$ and $x_f = 0$ for each $f \in EG \setminus \{e\}$. The vector x does not satisfy $x_e \ge 0$ but it does satisfy each of the other inequalities in (5.2).

<u>Lemma 5.6:</u> Let $e \in EG$. The inequality $x_e \le c_e$ is essential for P(G,b,c) if and only if e does not meet a node $v \in VG$ with $b_v < c_e$ or one with $b_v = c_e$ and $d_G(v) \ge 2$.

<u>Proof:</u> Clearly, if e meets a node $v \in VG$ with $b_v < c_e$ or one with $b_v = c_e$ and $d_G(v) \ge 2$, then $x_e \le c_e$ is not essential for P(G,b,c). Suppose that e does not meet such a node v. Let $x_e = c_e + 1$ and $x_f = 0$ for each $f \in EG \setminus \{e\}$. The vector x does not satisfy $x_e \le c_e$, but it does satisfy all other inequalities in (5.2). \square

Lemma 5.7: Let $v \in VG$ and let $b_u' = \min\{b_u, \Sigma\{c_e : e \in \delta(u) \cap \delta(v)\}\}$ for each node u in N(v). The inequality $x(\delta(v)) \leq b_v$ is essential for P(G,b,c) if and only if one of the following conditions holds.

 $(5.4) \qquad \begin{array}{lll} b'(N(v)) = b_v & \text{and} & v & \text{is in a two node connected component} \\ \text{of } G \text{ with } b_u = b_v, & \text{where } u & \text{is the other node of the} \\ \text{component, and if } \Sigma\{c_e \colon e \in \delta(v)\} = b_v & \text{then } d_G(v) = 1. \end{array}$

(5.5)
$$b'(N(v)) = b + 1 \text{ and there does not exist an edge} \\ (v_1, v_2) \in \gamma(N(v)) \text{ with } b'_1 = b_v \text{ and } b'_2 = b_v.$$

(5.6)
$$b'(N(v)) \ge b_v + 2.$$

Proof: As in the proof of Lemma 4.5, it is easy to check that if $x(\delta(v)) \leq b_v$ is essential for P(G,b,c) then one of the three conditions holds. If (5.5) or (5.6) holds then Lemma 4.5 and Lemma 5.4 together imply that $x(\delta(v)) \leq b_v$ is essential for P(G,b,c). Suppose that (5.4) holds. If $\Sigma\{c_e\colon e\in \delta(v)\} > b_v$, let $x_e = c_e$ for each $e\in \delta(v)$ and let $x_e = 0$ for each $e\in EG\setminus \delta(v)$. If $\Sigma\{c_e\colon e\in \delta(v)\} = b_v$, let $x_e = c_e+1$ for the edge e which meets e0 for each e1 for each e2 for each e3 for each e4 for the edge e5 which meets e5 for each e6 for each e6 for each inequality in (5.2) other than e6 for each e6 for each inequality in (5.2) other

These lemmas imply that the following linear system defines P(G,b,c).

- (i) $x_e \ge 0$ $\forall e \in EG$
- (ii) $x_e \le c_e \quad \forall e \in EG \quad \text{such that} \quad e \quad \text{does not meet a}$ node $v \in VG \quad \text{with} \quad b_v < c_e \quad \text{or one with} \quad b_v = c_e \quad \text{and} \quad d_G(v) \ge 2.$
 - (iii) $x(\delta(v)) \le b_v \quad \forall v \in VG$ for which either (5.4), (5.5), or (5.6) holds.
 - (iv) $x(EH) \le \lfloor \frac{1}{2}b^{(H,c)}(VH) \rfloor$ for all (b,c)-critical, (b,c)-nonseparable subgraphs H of G such that there does not exist an edge $(u,v) \in EG \setminus EH$ with $v,u \in VH$ and with $b_u^{(H,c)} = b_u$ and $b_v^{(H,c)} = b_v$.

The example given in the previous section shows that this linear system is not in general totally dual integral.

Remark 5.8: Lemma 5.4 and the result mentioned in Remark 4.6 together imply that (5.8) is the unique minimal defining system (unique up to positive scalar multiples of the inequalities) for P(G,b,c). Again, this result will appear in Cook and Pulleyblank [83], but will not be used here.

If G is connected and $|VG| \ge 3$, then G is $\underline{(b,c)}$ -bicritical if for each $v \in VG$ there exists a (b,c)-matching x of G such that $x(\delta(v)) = b_v^{(G,c)} - 2$ and $x(\delta(u)) = b_u^{(G,c)}$ for each $u \in VG\setminus\{v\}$. To obtain a totally dual integral defining system for P(G,b,c) it suffices to add an inequality $x(EH) \le \lfloor \frac{1}{2}b^{(H,c)}(VH) \rfloor$ for each (b,c)-bicritical, (b,c)-nonseparable subgraph H of G. To see this, another variation of Tutte's b-matching Theorem is needed. If H is a subgraph of G let $d_H^C(v) = \Sigma\{c_e : e \in \delta_H(v)\}$ for each $v \in VH$. Suppose that $S \subseteq VG$ and $T \subseteq VG\setminus S$. Let

$$(5.9) Q_{c}(S,T) = \Sigma\{b_{v} - d_{G[VG \setminus S]}^{c}(v) : v \in T\}$$

and let $G_c^T(VG\backslash S)$ be the graph obtained from $G[VG\backslash S]$ by taking each node $v \in T$ and performing the following three operations:

- (i) replace v by the nodes $\{v_e : e \in \delta(v)\}$
- (ii) replace each edge $e = (u,v) \in \delta(v)$ by the edge $e' = (u,v_e)$
- (iii) let $b_v = c_e$ for each $e \in \delta(v)$.

Let $\mathcal{D}_1(S,T,c)$ denote the set of odd connected components of $G_c^T(VG\backslash S)$ which contain at least two nodes.

Theorem 5.9: There exists a perfect (b,c)-matching of G if and only if $\forall S \subseteq VG$ and $T \subseteq VG \setminus S$: $b(S) \ge Q_c(S,T)+[\mathcal{D}_1(S,T,c)]$.

<u>Proof:</u> Let G' be the graph obtained from G by replacing each edge $e \in EG$ with edges e_1, \ldots, e_c each of which has the same end nodes as e. The graph G' has a perfect simple b-matching if and only if G has a perfect (b,c)-matching. Applying Theorem 4.7 to G' gives the result stated above.

Note that if c_e is sufficiently large for each $e \in EG$ then x is a (b,c)-matching of G if and only if it is a b-matching of G and if for some $S \subseteq VG$ and some $v \in T \subseteq VG \setminus S$ it is the case that $d_{G[VG \setminus S]}(v) > 0$, then $b(S) \ge Q_c(S,T) + |\mathcal{D}_1(S,T,c)|$. Thus, Theorem 5.9 reduces to Tutte's b-matching Theorem in this case.

An immediate consequence of Corollary 4.8 is the following result, which may also be proven directly from Theorem 5.9.

<u>Lemma 5.10:</u> If G is (b,c)-bicritical, then G has a perfect (b,c)-matching.

This result and the following lemmas will be used to characterize the Schrijver system for P(G,b,c).

Lemma 5.11: Let H be a connected subgraph of G with $b^{(H,c)}(VH)$ odd and $|VH| \ge 3$. The inequality $x(EH) \le \lfloor \frac{1}{2}b^{(H,c)}(VH) \rfloor$ is in the Schrijver system for P(G,b,c) only if either it is in (5.2)-(ii) or H is a (b,c)-critical, (b,c)-nonseparable graph.

<u>Proof:</u> Suppose that $x(EH) \le \lfloor \frac{1}{2}b^{(H,c)}(VH) \rfloor$ is not in (5.2)-(ii) and

that H is not a (b,c)-critical, (b,c)-nonseparable graph. If H is (b,c)-separable then Lemma 2.7 implies that the inequality is not in the Schrijver system for P(G,b,c). Suppose that H is not (b,c)-critical. Let $u \in VG$ be a node for which there does not exist a perfect (b',c)-matching of H where $b'_{u} = b'_{u} -1$ and $b'_{v} = b'_{v}$ for each $v \in VH \setminus \{u\}$. By Theorem 5.9 there exists $S \subseteq VH$ and $T \subseteq VH \setminus S$ such that

(5.10) b'(S) <
$$Q_c'(S,T)+|\mathcal{D}_1'(S,T,c)|$$

where $Q_c^{\,\prime}(S,T)$ and $\mathcal{D}_1^{\,\prime}(S,T,c)$ are $Q_c^{\,\prime}(S,T)$ and $\mathcal{D}_1^{\,\prime}(S,T,c)$ with respect to $b^{\,\prime}$ and H. Let S and T be such subsets of VH (it may be assumed that if $v \in VH \setminus S$ and $d_{H[VH \setminus S]}(v) = 0$ then $v \in T$).

Let \mathcal{H} be the set of all connected subgraphs \overline{H} of G with $|V\overline{H}| \geq 2$. Corollary 5.2 implies that for each integer vector w, the linear program

$$\min \Sigma\{c z_e : e \in EG\} + \Sigma\{b_y y_v : v \in VG\} + \Sigma\{[\frac{1}{2}b^{(\overline{H},c)}(V\overline{H})]Y_{\overline{H}} : \overline{H} \in H\}$$

s.t.
$$z_e^{+y}(\Psi(e))+Y(H(e)) \ge w_e^{-\forall e \in EG}$$
 $z_e^{-\geq 0} \quad \forall e \in EG$
 $y_v^{\geq 0} \quad \forall v \in VG$
 $Y_{\overline{H}}^{-\geq 0} \quad \forall \overline{H}^{-\epsilon} \in H$

has an integer optimal solution. To prove that $x(EH) \le \frac{1}{2}b^{(H,c)}(VH)$ is not in the Schrijver system for P(G,b,c) it suffices to show that for each integer vector w the linear program (5.11) has an integer

optimal solution with $Y_{H} = 0$.

Let w be an integer vector and let (z,y,Y) be an integer optimal solution to (5.11). Suppose that $Y_H > 0$. The inequality (5.10) implies that

(5.12)
$$b^{(H,c)}(S) \leq Q_c^{(H,c)}(S,T) + |\mathcal{D}_1^{(H,c)}(S,T,c)|,$$

where $Q_c^{(H,c)}(S,T)$ and $\mathcal{D}_1^{(H,c)}(S,T,c)$ are $Q_c(S,T)$ and $\mathcal{D}_1(S,T,c)$ with respect to $b^{(H,c)}$ and H. Let \mathcal{D} be the set of connected components of $H_c^T(VH \setminus S)$ that contain at least two nodes (so $\mathcal{D}_1^{(H,c)}(S,T,c) \subseteq \mathcal{D}$). Each connected component \widetilde{H}_i of $H_c^T(VH \setminus S)$ corresponds to a subgraph H_i of H. Note that if \widetilde{H}_i is in \mathcal{D} then $b^{(H,c)}(VH_i) \leq b^{(H,c)}(VH_i)$.

From the definition of $\,{\rm Q}_{_{{\bf C}}}({\rm S},{\rm T})\,\,$ and $\,{\cal D}\,\,$ the following equality holds:

(5.13)
$$b^{(H,c)}(VH) = b^{(H,c)}(S) + Q_c^{(H,c)}(S,T) + b^{(H,c)}(D)$$
,

where $b^{(H,c)}(\mathcal{D}) = \Sigma\{b^{(H,c)}(\tilde{V}_{\mathbf{i}}): \tilde{H}_{\mathbf{i}} \in \mathcal{D}\}.$ Now (5.12) implies that

(5.14)
$$b^{(H,c)}(VH) \ge b^{(H,c)}(VH) + b^{(H,c)}(S) - Q_c^{(H,c)}(S,T) - |\mathcal{D}_1^{(H,c)}(S,T,c)|.$$

Combining (5.13) and (5.14) gives

(5.15)
$$b^{(H,c)}(VH) \ge 2b^{(H,c)}(S)+b^{(H,c)}(D)-|D_1^{(H,c)}(S,T,c)|.$$

It follows that

(5.16)
$$\left[\frac{1}{2}b^{(H,c)}(VH)\right] \ge b^{(H,c)}(S) + \frac{1}{2}b^{(H,c)}(D) - \frac{1}{2}\left[D_1^{(H,c)}(S,T,c)\right]$$

which is equivalent to

$$(5.17) \qquad \left[\frac{1}{2}b^{(\mathrm{H,c})}(\mathrm{VH})\right] \geq b^{(\mathrm{H,c})}(\mathrm{S}) + \Sigma\left\{\left[\frac{1}{2}b^{(\mathrm{H,c})}(\mathrm{V}\widehat{\mathrm{H}}_{\mathbf{i}})\right] : \widetilde{\mathrm{H}}_{\mathbf{i}} \in \mathcal{D}\right\}.$$

This inequality implies that

$$(5.18) \qquad \left[\frac{1}{2}b^{(\mathrm{H,c})}(\mathrm{VH})\right] \geq b^{(\mathrm{H,c})}(\mathrm{S}) + \Sigma\left\{\left[\frac{1}{2}b^{(\mathrm{H_i,c})}(\mathrm{VH_i})\right]: \widetilde{\mathrm{H}}_{i} \in \mathcal{D}\right\}.$$

Using this it is easy to construct an integer optimal solution to (5.11) with $Y_H = 0$. Define $(\bar{z}, \bar{y}, \bar{Y})$ as follows

$$\begin{split} \overline{Y}_{H} &= 0 \\ \overline{Y}_{H_{1}} &= Y_{H_{1}} + Y_{H} \quad \text{for each} \quad \widetilde{H}_{1} \in \mathcal{D} \\ \overline{Y}_{K} &= Y_{K} \quad \text{for each} \quad \text{other} \quad K \in \mathcal{H} \\ \overline{y}_{V} &= y_{V} + Y_{H} \quad \text{for each} \quad V \in S \quad \text{with} \quad b_{V}^{(H,c)} = b_{V} \\ \overline{y}_{V} &= y_{V} \quad \text{for each} \quad \text{other} \quad V \in VG \\ \overline{z}_{e} &= z_{e} + Y_{H} \quad \text{for each} \quad e \in EH \quad \text{which meets a node} \quad u \in S \quad \text{with} \\ b_{u}^{(H,c)} &< b_{u}. \\ \overline{z}_{e} &= z_{e} \quad \text{for each} \quad \text{other} \quad e \in EG. \end{split}$$

It is straightforward to check that $(\bar{z},\bar{y},\bar{Y})$ is a feasible solution to (5.11) (note that since $b^{(H,c)}(VH)$ is odd, H is not H_i for some $\tilde{H}_i \in \mathcal{D}$). The inequality (5.18) implies that it is also an optimal solution to (5.11).

Lemma 5.12: Let H be a connected subgraph of G with $b^{(H,c)}$ (VH) even and $|VH| \ge 3$. The inequality $x(EH) \le \lfloor \frac{1}{2}b^{(H,c)}$ (VH) j is in the Schrijver system for P(G,b,c) only if either it is in (5.2)-(ii) or

H is a (b,c)-bicritical, (b,c)-nonseparable graph.

<u>Proof:</u> Suppose that $x(EH) \leq \frac{1}{2}b^{(H,c)}(VH)J$ is not in (5.2)-(ii) and that H is not a (b,c)-bicritical, (b,c)-nonseparable graph. Again, if H is (b,c)-separable then Lemma 2.7 implies that the inequality is not in the Schrijver system for P(G,b,c). Suppose that H is not (b,c)-bicritical. Let $u \in VG$ be a node which which there does not exist a perfect (b,c)-matching of H where $b_u^{'}=b_u^{(H,c)}-2$ and $b_v^{'}=b_v^{(H,c)}$ for each v VH\{u} (it may be assumed that $b_u^{'}\geq 0$). Let $S\subseteq VH$ and $T\subseteq VH\setminus S$ be sets such that

(5.19)
$$b'(S) < Q'_{c}(S,T) + |\mathcal{D}'_{1}(S,T,c)|$$
.

This inequality implies that

(5.20)
$$b(H,c)(S) \le Q_c^{(H,c)}(S,T) + |\mathcal{D}_1^{(H,c)}(S,T,c)| + 1.$$

Now since $b^{(H,c)}(VH)$ is even, (5.13) implies that $b^{(H,c)}(S)+Q_c^{(H,c)}(S,T)+|\mathcal{D}_1^{(H,c)}(S,T,c)|$ is even. So (5.20) implies (5.12). So the lemma can be proven in the same manner as Lemma 5.11.

The above two lemmas together imply Lemma 4.9, which completes the proof of Theorem 4.10.

The following result characterizes the Schrijver system for P(G,b,c).

Theorem 5.13: The unique minimal totally dual integral system with integer left hand sides which defines the convex hull of the c-capacitated b-matchings of G is (5.8) together with

(5.21) $x(EH) \leq \frac{1}{2}b^{(H,c)}(VH)$ for all (b,c)-bicritical,

(b,c)-nonseparable subgraphs H of G such that there does not exist an edge (u,v) \in EG\EH with u,v \in VH and with $b_u^{(H,c)} = b_u$ and $b_v^{(H,c)} = b_v$ and such that there does not exist an edge (u,v) \in EG with u \in VH, v \notin VH and either $c_{(u,v)} = 1$ or $b_v = 1$.

<u>Proof:</u> If an inequality is not in (5.2), then it is not in the Schrijver system for P(G,b,c). It is easy to check that an inequality in (5.2)-(i) or (5.2)-(ii) is in the Schrijver system for P(G,b,c) only if it is in (5.8)-(i), (5.8)-(ii), or (5.8)-(iii). Conversely, by Lemma 5.5, Lemma 5.6 and Lemma 5.7, each inequality in (5.8)-(i), (5.8)-(ii), and (5.8)-(iii) is essential for P(G,b,c) and hence is in the Schrijver system for P(G,b,c).

Let H be a connected subgraph of G with $|VH| \ge 3$. Suppose that $b^{(H,c)}(VH)$ is odd. Using Lemma 5.11, it is easy to check that

(5.22)
$$x(EH) \le \frac{1}{2}b^{(H,c)}(VH)$$

is in the Schrijver system for P(G,b,c) only if it is in (5.8)-(iv). Conversely, Lemma 2.7 implies that each inequality in (5.8)-(iv) is in the Schrijver system for P(G,b,c). Suppose that $b^{(H,c)}(VH)$ is even. Lemma 5.12 implies that (5.22) is in the Schrijver system for P(G,b,c) only if H is (b,c)-bicritical, (b,c)-nonseparable. Suppose that H is (b,c)-bicritical, (b,c)-nonseparable. Lemma 2.7 implies that (5.22) is in the Schrijver system for P(G,b,c) only if it is in (5.21). Conversely, Lemma 2.7, together with Lemma 5.10, implies that each inequality in (5.21) is in the Schrijver system for P(G,b,c).

Theorem 4.10 may be obtained from the above result by letting $c_e = 1$ for each $e \in EG$. To obtain the characterization of the Schrijver system for the convex hull of the b-matchings of G, given in Theorem 3.3, from the above result let $c_e = \alpha$ for each $e \in EG$ where $\alpha = 2 \cdot \max\{b_v \colon v \in VG\}$. Now a vector x is a b-matching of G if and only if x is a (b,c)-matching of G. So (5.8) together with (5.21) is the Schrijver system for the convex hull of the b-matchings of G. It is straightforward to check that this linear system is identical to (3.3) together with (3.4).

It is possible to test in polynomial time whether or not a given subgraph of G is (b,c)-critical or (b,c)-bicritical. Indeed, using Edmonds' blossom algorithm (Edmonds [65,65a]) and a "scaling" argument similar to the one used by Edmonds and Karp [72] to solve min cost flows, Cunningham and Marsh (see Marsh [79]) found a polynomial-time algorithm for the b-matching problem. Using the construction of Tutte, this algorithm can be used to solve the (b,c)-matching problem in polynomial time, which implies that it is possible to test in polynomial time whether or not a graph is (b,c)-critical or (b,c)-bicritical. The polynomial-time algorithm of Cunningham and Marsh, together with Theorem 3.3, also gives a polynomial-time algorithm to test whether or not a graph is b-separable and, hence, whether or not an inequality is in the Schrijver system for the convex hull of the b-matchings of G. A characterization of (b,c)-nonseparability which yields a polynomial-time test for (b,c)-nonseparability is given in Cook and Pulleyblank [83].

Remark 5.14: An alternative polynomial-time algorithm for the b-matching problem has been found by Anstee[83]. Anstee's algorithm uses a polynomial-time min cost flow algorithm as a subroutine and thus avoids a separate "scaling" argument.

6. Triangle-free 2-matchings

Let G be a graph. A 2-matching of G is a b-matching with $b_v = 2$ for each $v \in VG$. Motivated by the fact that the 2-matching problem is a relaxation of the travelling salesman problem, Cornuéjols and Pulleyblank [80] considered a constrained variation of 2-matchings called triangle-free 2-matchings. A 2-matching, x, is a triangle-free 2-matching if $x_0 + x_1 + x_2 \le 2$ for each triple of edges $\{e_1, e_2, e_3\} \subseteq EG$

which form the edges of a triangle of G (a triangle is a circuit of length 3). Cornvéjols and Pulleyblank found a polynomial-time algorithm for solving the triangle-free 2-matching problem and characterized the unique minimal defining system for T(G), the convex hull of the triangle-free 2-matchings of G. The Schrijver system for T(G) will be characterized in this section.

Let \mathcal{T} be the set of all triangles of G. The following result is a consequence of the Cornvéjols-Pulleyblank algorithm.

Theorem 6.1: A totally dual integral defining system for T(G) is

$$x_{e} \ge 0 \qquad \forall e \in EG$$

$$x(\delta(v)) \le 2 \qquad \forall v \in VG$$

$$x(ET) \le 2 \qquad \forall T \in T$$

$$x(\gamma(S)) \le |S| \qquad \forall S \subseteq VG.$$

The following result of Cornuejols and Pulleyblank [80] follows very easily from the above theorem.

Theorem 6.2: The unique minimal defining system (unique up to positive scalar multiples of the inequalities) for the convex hull of the

triangle free 2-matchings of G is

 $\begin{array}{c} x_e \geq 0 \quad \forall e \in EG \\ \\ x(\delta(v)) \leq 2 \quad \text{for all} \quad v \in VG \quad \text{such that either} \quad d_G(v) \geq 3 \quad \underline{\text{or}} \\ \\ d_G(v) = 2 \quad \text{and } v \text{ is not the node of a triangle } \underline{\text{or}} \quad d_G(v) = 1 \text{ and} \\ \\ v \quad \text{is in a 2-node connected component of } G. \end{array}$

 $x(ET) \leq 2 \quad \forall T \in T.$

If G is a circuit of length 5, then (6.2) is not totally dual integral. This is an example of a triangle-free bicritical graph. If x is a triangle-free 2-matching such that $x(\delta(v)) = 2$ for each $v \in VG$, then x is a perfect triangle-free 2-matching. If G is connected and $|VG| \ge 4$, then G is triangle-free-bicritical if for each $v \in VG$ the graph obtained by deleting v from G has a perfect triangle-free 2-matching. A triangle T of a connected graph G is a pendent triangle of G if T contains a cutnode of G and T contains two nodes v_1, v_2 with $d_G(v_1) = d_G(v_2) = 2$.

Theorem 6.3: The unique minimal totally dual integral system with integer left hand sides that defines the convex hull of the triangle-free 2-matchings of G is (6.2) together with

(6.3) $x(\gamma(S)) \le |S|$ for all $S \subseteq VG$ such that G[S] is triangle-free-bicritical and contains no triangle T which is a pendent triangle of G[S].

<u>Proof:</u> Let $S \subseteq VG$. Suppose that G[S] is not triangle-free-bicritical. Let $v \in S$ be a node such that $G[S \setminus \{v\}]$ does not have a perfect triangle-free 2-matching. The inequality $x(\gamma(S)) \leq |S|$ can be obtained by summing the valid inequalities $x(\delta_{G[S]}(v)) \leq 2$ and $x(\gamma(S \setminus \{v\})) \leq |S|-2$.

So $x(\gamma(S)) \le |S|$ is not in the Schrijver system for T(G). Using this, it follows from Theorem 6.1 that (6.2) together with (6.3) is a totally dual integral defining system for T(G). So an inequality is in the Schrijver system for T(G) only if it is in (6.2) or (6.3).

 \sim Each inequality in (6.2) is essential for T(G) and hence is in the Schrijver system for T(G). Let $S \subseteq VG$ and suppose that G[S]is a triangle-free-bicritical graph which contains no triangle T which is a pendent triangle of G[S]. Let $V' \subseteq VG$, $T' \subseteq T$, $S' \subseteq \{R: R \subseteq VG\}$. Call (V', T', S') a cover of G[S] if for each edge $e \in \gamma(S)$ either $e \in \delta(v)$ for some $v \in V'$ or $e \in ET$ for some $T \in T'$ or $e \in \gamma(S')$ for some $S' \in S'$. If (V', T', S') is a cover of G[S] then the weight of (V',T',S') is w(V',T',S') = 2|V'| + $+2|T'|+\Sigma\{|S'|: S' \in S'\}$. To show that $x(\delta(S)) \leq |S|$ is in the Schrijver system for T(G) it suffices to show that the weight of any cover of G[S] such that $S \notin S'$ is greater than S. Let (V', T', S')be a cover of G[S] of minimum weight (since $(\emptyset,\emptyset,\{S\})$) is a cover of G[S], $w(V',T',S') \leq |S|$). Suppose that $S \notin S'$ and assume that of all minimum weight coverings of G[S] with $S \notin S'$, |T'|is as small as possible.

Since G[S] is triangle-free-bicritical, V' must be the empty set. Suppose that S' is also the empty set. Since |T'| is as small as possible, $|VT_1 \cap VT_2| \le 1$ for all $T_1, T_2 \in T'$. Now since there is no triangle T in G[S] which is a pendent triangle of G[S] and since $|S| \ge 4$, there exists a circuit C in G[S] which is not a triangle in T'. Let $\{v_0, \dots, v_{k-1}\}$ be the node set of C, where v_i is adjacent to v_{i+1} , for $i = 0, \dots, k-1$ (subscripts should be taken modulo k). There exist distinct triangles T_0, \dots, T_{k-1} in T' such that $VT_i = \{v_i, v_{i+1}, u_i\}$, where $u_i \notin VC$ and $u_i \ne u_j$ for $i \ne j$. Let G' be the graph with node set $\cup \{VT_i: i = 0, \dots, k-1\}$ and edge set

Corollary 6.4: If G is triangle-free-bicritical with no pendent triangle, then G has a perfect triangle-free 2-matching.

<u>Proof:</u> Suppose that G is triangle-free-bicritical with no pendent triangle. Theorem 6.3 implies that $x(EG) \le |VG|$ is in the Schrijver system for T(G). Hence, G must have a triangle-free 2-matching of size |VG|.

Corollary 6.5: A graph G is triangle-free nonseparable if and only if either G is isomorphic to $K_{1,n}$ for some n or G is triangle-

free-bicritical with no pendent triangle.

Proof: This follows from Theorem 6.3 and Lemma 2.7.

REFERENCES

- [81] M. Akgül, <u>Topics in Relaxation and Ellipsoidal Methods</u>, Phd. Thesis, University of Waterloo, 1981.
- [83] R.P. Anstee, "A polynomial algorithm for b-matching; an alternative approach", Research report CORR 83-22, University of Waterloo, 1983.
- [82] R.P. Anstee and M. Farber, "Characterizations of totally balanced matrices", Research report CORR 82-5, University of Waterloo (Submitted to Journal of Algorithms), 1982.
- [77] A. Bachem, "The theorem of Minkowski for polyhedral monoids and aggregated linear diophantine systems", Report No. 7765-OR, Institut für Ökonometrie und Operations Research, University of Bonn, 1977.
- [82] A. Bachem and M. Grötschel, "New aspects of polyhedral theory", in B. Korte (ed.), Modern Applied Mathematics Optimization and Operations Research (North Holland, 1982), 51-106.
- [79] A. Bachem and R. Kannan, "Applications of polynomial Smith normal form calculations", in L. Collatz et al (eds.), Numerische

 Methoden bei graphentheoretischen und kombinatorischen

 Problemen, Band 2 (Birkhäuser Verlag, Basel, 1979), 195-215.
- [79] A. Bachem and R. von Randow, "Integer theorems of Farkas lemma type", Operations Verfahren, Vol 32 (1979), 19-28.
- [82] E. Balas and W.R. Pulleyblank, "The perfectly matchable subgraph polytope of a bipartite graph", to appear in Networks (1982).
- [72] M.L. Balinski, "Establishing the matching polytope", JCT (B) 13 (1972), 1-13.

- [73] Zs. Baranyai, "On the factorization of the complete uniform hypergraph", in A. Hajnal, R. Rado, and V.T. Sós, eds.,

 <u>Infinite and Finite Sets</u> (Proc. Intern. Coll. Keszthely,
 1973), North-Holland, Amsterdam.
- [77] S. Baum and L. Trotter, "Integral rounding and polyhedral decomposition of totally unimodular systems", in <u>Optimization and Operations Research</u> (Proc. Bonn 1977: R. Henn, B. Korte, and W. Oettli, eds.), Lecture Notes in Economic and Math. Systems 157 (Springer, Berlin, 1977), 25-33.
- [82] S. Baum and L.E. Trotter, Jr., "Finite checkability for integer rounding properties in combinatorial programming problems", Math. Prog. 22 (1982), 141-147.
- [58] C. Berge, "Sur le couplage maximum d'un graphe", C. R. Acad. Sci. Paris 247 (1958), 258-259.
- [62] C. Berge, "Sur une conjecture relative au probleme des codes optimaux", Comm. 13e assemblée générale de l'URSI, Tokyo, 1962.
- [72] C. Berge, "Balanced matrices", Math. Prog. 2 (1972), 19-31.
- [73] C. Berge, "Balanced hypergraphs and some applications to graph theory", in J.N. Srivastava et al., eds., <u>A Survey of Combinatorial Theory</u> (North-Holland, 1973).
- [80] C. Berge, "Balanced matrices and property (G)", Math. Prog. Study 12 (1980), 163-175.
- [70] C. Berge and M. Las Vergnas, "Sur un theoreme du type König pour hypergraphes", Annals of the New York Academy of Sciences 175 (1970), 32-40.
- [81] R.G. Bland, D. Goldfarb, and M.J. Todd, "The ellipsoid method: a survey", Operations Research Vol. 29, No. 9 (1981), 1039-1091.
- [76] J.A. Bondy and U.S.R. Murty, <u>Graph Theory with Applications</u> (Macmillan, London, 1976).

- [79] M. Boulala and J.P. Uhry, "Polytope des indépendants dans un graphe sêrie parallèle", Discrete Math. 27 (1979), 225-243.
- [81] M. Burlet, "Etude algorithmique de certaines classes de graphes parfaits", Thèse troisième cycle, Université Scientifique et Mèdicale de Grenoble, Mathématiques appliquées, Grenoble, 1981.
- [82] M. Burlet and J. Fonlupt, "Polynomial algorithm to recognize a Meyniel graph", Research report no. 303, Grenoble, May 1982.
- [82] M. Burlet and J.P. Uhry, "Parity graphs", Annals of Dis. Math. 16 (1982), 1-26.
- [82] K. Cameron, <u>Polyhedral and Algorithmic Ramifications of Antichains</u>, Ph.D. Thesis, Univ. of Waterloo, 1982.
- [81] R. Chandrasekaran, "Polynomial algorithms for totally dual integral systems and extensions", Annals of Dis. Math 11 (1981), 39-51.
- [82] R. Chandrasekaran and S. Shirali, "Total weak unimodularity: testing and applications", Research report, The University of Texas at Dallas, 1982.
- [73] V. Chvatal, "Edmonds polytopes and a hierarchy of combinatorial problems", Dis. Math. 4 (1973), 305-337.
- [75] V. Chvatal, "On certain polytopes associated with graphs", JCT(B) 18 (1975), 135-154.
- [81] V. Chvatal, "Perfectly ordered graphs", Technical Report SOCS-S1.28, McGill Univ., 1981.
- [71] S.A. Cook, "The complexity of theorem-proving procedures", in Proc. 3rd Ann. ACM Symp. on Theory of Computing (ACM, New York, 1971), 151-158.
- [81] W. Cook, "A minimal totally dual integral defining system for the b-matching polyhedron", Report CORR 81-15, 1981 (to appear in SIAM Journal on Alg. and Disc. Methods).

- [83] W. Cook, "Operations that preserve total dual integrality",
 Operations Research Letters, Vol. 2, No. 1 (1983), 31-35.
- [83] W. Cook, J. Fonlupt, and A. Schrijver, "An integer analogue of Carathéodory's theorem", to appear.
- [82] W. Cook, L. Lovász, and A. Schrijver, "A polynomial-time test for total dual integrality in fixed dimension", to appear.
- [83] W. Cook and W.R. Pulleyblank, "Linear systems for constrained matching problems", in preparation.
- [80] G. Cornuéjols and W. Pulleyblank, "A matching problem with side conditions", Dis. Math. 29 (1980), 135-159.
- [80a] G. Cornuéjols and W. Pulleyblank, "The travelling salesman polytope and {0,2}-matchings", to appear in Annals of Discrete Math. (1980).
- [81] G. Cornuéjols and W.R. Pulleyblank, "Critical graphs, matchings and tours: a hierarchy of relaxations for the travelling salesman problem", to appear in Combinatorica (1981).
- [73] W.H. Cunningham, <u>A Combinatorial Decomposition Theory</u>, Ph.D. Thesis, Univ. of Waterloo, 1973.
- [81] W. Cunningham, "Testing membership in matroid polyhedra", Research
 Report 81207-OR, Inst. für Operations Research, Univ. Bonn, 1981.
- [82] W. Cunningham and A. Frank, "A primal-dual algorithm for submodular flows", to appear in Math. of Op. Res. (1982).
- [78] W.H. Cunningham and A.B. Marsh, III, "A primal algorithm for optimum matching", Math. Prog. Study 8 (1978), 50-72.
- [63] G.B. Dantzig, <u>Linear Programming and Extensions</u> (Princeton Univ. Press, Princeton, New Jersey, 1963).

- [65] J. Edmonds, "Maximum matching and a polyhedron with 0,1 vertices", J. Res. Nat. Bur. Standards 69B (1965), 125-130.
- [65a] J. Edmonds, "Paths, trees and flowers", Canadian J. Math. 17 (1965), 449-467.
- [65b] J. Edmonds, "Minimum partion of a matroid into independent sets", J. Res. Nat. Bur. Standards 69B (1965), 67-72.
- [67] J. Edmonds, "Systems of distinct representatives and linear algebra", J. Res. Nat. Bur. Standards 71B (1967), 241-245.
- [70] J. Edmonds, "Submodular functions, matroids, and certain polyhedra", in R. Guy, H. Hanani, N. Sauer and J Schonheim, eds.,

 <u>Combinatorial Structures and Their Applications</u> (Gordon and Breach, New York, 1970), 69-87.
- [71] J. Edmonds, "Matroids and the greedy algorithm", Math. Prog. 1 (1971), 127-136.
- [72] J. Edmonds, "Edge-disjoint branchings", Combinatorial Algorithms (Algorithmic Press, New York, 1972), 91-96.
- [77] J. Edmonds and R. Giles, "A min-max relation for submodular functions on graphs", Annals of Dis. Math 1 (1977), 185-204.
- [70] J. Edmonds and E. Johnson, "Matching: a well-solved class of integer linear programs", in R. Guy, H. Hanani, N. Sauer and J. Schonheim, eds., <u>Combinatorial Structures and Their</u> Applications (Gordon and Breach, New York, 1970), 89-92.
- [73] J. Edmonds and E.L. Johnson, "Matching, Euler tours and the Chinese Postman", Math. Prog. 5 (1973), 88-124.
- [72] J. Edmonds and R.M. Karp, "Theoretical improvements in algorithmic efficiency for network flow problems", J. ACM 19 (1972), 248-264.

- [82] J. Edmonds, L. Lovász, and W.R. Pulleyblank, "Brick decompositions and the matching rank of graphs", Research Report CORR 82-17, Univ. of Waterloo, 1982 (to appear in Combinatorica).
- [31] E. Egerváry, "Matrixok kombinatorius tulajdonsagairol", Mat. es Fiz. Lapok 38 (1931), 16-28.
- [81] R. Euler, "Perfect independence systems", Report 81-19, Mathematisches Institüt, Univ. of Köln, 1981.
- [82] P. Feofiloff and D.H. Younger, "Directed cut transversal packing for source-sink connected graphs", Research Report CORR 82-33, Univ. of Waterloo, 1982.
- [82] J. Fonlupt and J.P. Uhry, "Transformations which preserve perfectness and h-perfectness of graphs", Annals of Dis. Math. 16 (1982), 83-95.
- [62] L.R. Ford, Jr. and D.R. Fulkerson, Flows in Networks (Princeton Univ. Press, Princeton, 1962).
- [79] A. Frank, "Kernel systems of directed graphs", Acta Sci. Math (Szeged) 41 (1979), 63-76.
- [81] A. Frank, "Finding feasible vectors of Edmonds-Giles polyhedra",

 Report 81205-OR, Institut für Operations Research, Bonn, 1981.
- [81a] A. Frank, "Generalized polymatroids", Report 81206-OR, Institut für Operations Research, Bonn, 1981.
- [82] A. Frank, "An algorithm for submodular functions on graphs", Annals of Dis. Math. 16 (1982), 189-212.
- [81] A. Frank and É. Tardos, "Matroids from crossing families", to appear in the proceedings of the Sixth Hungarian Combinatorial Colloquium, Eger, 1981.
- [76] M.A. Frumkin, "An application of modular arithmetic to the construction of algorithms for solving systems of linear equations", Soviet Math. Dokl., Vol. 17 (1976), No. 4, 1165-1168.

- [78] S. Fujishige, "Algorithms for solving the independent flow problem", J. Oper. Res. Soc. Japan 21 (1978), 189-203.
- [68] D.R. Fulkerson, "Networks, frames, and blocking systems", in <u>Mathematics of the Decision Sciences</u> Part 1, Lectures in Applied Math., Vol II (Am. Math. Soc., Providence, RI, 1968), 303-335.
- [70] D.R. Fulkerson, "Blocking polyhedra", in <u>Graph Theory and its</u>

 <u>Applications</u>, B. Harris, ed. (Academic Press, 1970), 93-112.
- [71] D.R. Fulkerson, "Blocking and anti-blocking pairs of polyhedra", Math. Prog. 1 (1971), 168-194.
- [72] D.R. Fulkerson, "Anti-blocking polyhedra", JCT(B) 12 (1972), 50-71.
- [73] D.R. Fulkerson, "On the perfect graph theorem", in T.C. Hu and S.M. Robinson, eds., <u>Mathematical Programming</u> (Proc. of an Advanced Seminar, Madison, Wisconsin, 1972), Academic Press, New York, 1973.
- [74] D.R. Fulkerson, "Packing rooted directed cuts in a weighted directed graph", Math. Prog. 6 (1974), 1-13.
- [74] D.R. Fulkerson, A.J. Hoffman, and R. Oppenheim, "On balanced matrices", Math. Prog. Study 1 (1974), 120-132.
- [51] D. Gale, H.W. Kuhn, and A.W. Tucker, "Linear programming and the theory of games", in T.C. Koopmans, ed., <u>Activity Analysis of</u>
 Production and Allocation (John Wiley and Sons, New York, 1951).
- [50] T. Gallai, "On factorisation of graphs", Acta Mathematica (Budapest) 1 (1950), 133-153.
- [79] M.R. Garey and D.S. Johnson, <u>Computers and Intractability: A Guide</u>
 to the Theory of NP-Completeness (W.H. Freeman and Co.,
 San Francisco, 1979).

- [76] J. v. z. Gathen and M. Sieveking, Chapter 4 of <u>Komplexitaet von</u> <u>Entscheidungsproblemen</u>, edited by E. Specker and V. Strassen, Lecture Notes in Computer Science 43 (Springer-Verlag, New York, 1976).
- [75] R. Giles, <u>Submodular Functions</u>, <u>Graphs</u>, and <u>Integer Polyhedra</u>, Ph.D. Thesis, Univ. of Waterloo, 1975.
- [81] R. Giles and J.B. Orlin, "Verifying total dual integrality", manuscript, 1981.
- [79] F.R. Giles and W.R. Pulleyblank, "Total dual integrality and integer polyhedra", Lin. Alg. and its Appl. 25 (1979), 191-196.
- [63] R. Gomory, "An algorithm for integer solutions to linear programs", in R.L. Graves and P. Wolfe, eds., Recent Advances in Mathematical Programming (McGraw-Hill, NY, 1963), 269-302.
- [80] J. Green-Krotki, "Matching polyhedra", M.S. Thesis, Carleton Univ., Dept. of Math. and Stat., 1980.
- [82] H. Groflin and A.J. Hoffman, "Lattice polyhedra II: generalization, constructions and examples", Annals of Dis. Math. 15 (1982), 189-203.
- [81] M. Grotschel, L. Lovász, and A. Schrijver, "The ellipsoid method and its consequences in combinatorial optimization", Combinatorica 1 (1981), 169-197.
- [81a] M. Grotschel, L. Lovász, and A. Schrijver, "Polynomial algorithms for perfect graphs", Report No. 81176-OR, Univ. Bonn, 1981.
- [82] M. Grotschel, L. Lovász, and A. Schrijver, "Geometric methods in combinatorial optimization", to appear in the proceedings of the Silver Jubilee Conference on Combinatorics, Univ. of Waterloo (1982).
- [78] R. Hassin, On Network Flows, Ph.D. Thesis, Yale Univ., 1978.

- [82] R. Hassin, "Minimum cost flow with set constraints", Networks, Vol. 12 (1982), 1-21.
- [56] I. Heller and C.B. Tompkins, "An extension of a theorem of Dantzig's", in Ann. Math. Study 38 (Princeton Univ. Press, NJ, 1956), 247-254.
- [51] C. Hermite, "Sur 1'introduction des variables continues dans la theorie des nombres", J.R. Angew. Math. 41 (1851), 191-216.
- [90] D. Hilbert, "Uber die Theorie der algebraischen Formem", Math. Ann. 36 (1890), 473-534.
- [60] A.J. Hoffman, "Some recent applications of the theory of linear inequalities to extremal combinatorial analysis", in <u>Proc.</u> <u>Symp. Appl. Math 10, Combinatorial Analysis</u> (Providence, RI, 1960), 113-127.
- [74] A.J. Hoffman, "A generalization of max flow-min cut", Math. Prog. 6 (1974), 352-359.
- [76] A.J. Hoffman, "Total unimodularity and combinatorial theorems", Lin. Alg. and its Appl. 13 (1976), 103-108.
- [76a] A.J. Hoffman, "On lattice polyhedra II: constructions and examples", IBM Research Report RC 6268 (1976).
- [78] A.J. Hoffman, "On lattice polyhedra III: blockers and anti-blockers of lattice clutters", Math. Prog. Study 8 (1978), 197-207.
- [79] A.J. Hoffman, "The role of unimodularity in applying linear inequalities to combinatorial theorems", Annals of Dis. Math 4 (1979), 73-84.
- [82] A.J. Hoffman, "Ordered sets and linear programming", in I. Rival (ed.),

 Ordered Sets (D. Riedel, Dordrecht, Holland, 1982), 619-654.

- [82] A.J. Hoffman, A. Kolen, and M. Sakarovitch, "Totally-balanced and greedy matrices", submitted to SIAM J. on Alg. and Dis. Math. (1982).
- [56] A.J. Hoffman and J.B. Kruskal, "Integral boundary points of convex polyhedra", in <u>Ann. Math. Study 38</u> (Princeton Univ. Press, NJ, 1956), 223-247.
- [78] A.J. Hoffman and R. Oppenheim, "Local unimodularity in the matching polytope", Ann. of Dis. Math. 2 (1978), 201-209.
- [77] A.J. Hoffman and D.E. Schwartz, "On partitions of a partially ordered set", JCT(B) 23 (1977), 3-13.
- [78] A.J. Hoffman and D.E. Schwartz, "On lattice polyhedra", in A. Hajnal and V.T. Sos, eds., <u>Combinatorics</u> (North-Holland, Amsterdam, 1978), 593-598.
- [78] R. Jeroslow, "Some basis theorems for integral monoids", Math. of Op. Res., Vol. 3, No. 2, May 1978.
- [75] E. Johnson, "On cut set integer polyhedra", Cahiers du Centre de Recherche Operationelle (Actes Journées Franco-Belges) 17 (1975), 235-251.
- [79] R. Kannan and A. Bachem, "Polynomial algorithms for computing the Smith and Hermite normal forms of an integer matrix", SIAM

 J. Computing, Vol. 8, No. 4, November 1979, 499-507.
- [72] R.M. Karp, "Reducibility amoung combinatorial problems", in
 R.E. Miller and J.W. Thatcher (eds.), Complexity of Computer
 Computations (Plenum Press, New York, 1972).
- [75] R.M. Karp, "On the complexity of combinatorial problems", Networks 5 (1975), 45-68.
- [79] L.G. Khachiyan, "A polynomial algorithm in linear programming", Soviet Math. Dokl. 20 (1979), 191-194.

- [82] E. Korach, <u>Packing of T-cuts and other aspects of dual integrality</u>, Ph.D. Thesis, Univ. of Waterloo, 1982.
- [82] E.L. Lawler, "Generalizations of polymatroidal network flows", Report BW 158, Mathematisch Centrum, Amsterdam, 1982.
- [82] E.L. Lawler and C.U. Martel, "Computing maximal 'polymatroidal' network flows", Math. of Op. Res., Vol. 7, No. 3, August 1982.
- [79] A. Lehman, "On the width-length inequality", Math. Prog. 17 (1979), 403-417.
- [82] A.K. Lenstra, H.W. Lenstra, Jr., and L. Lovász, "Factoring polynomials with rational coefficients", Report 82-05, Mathematisch Institut, Amsterdam, 1982.
- [81] H.W. Lenstra, Jr., "Integer programming with a fixed number of variables", Report 81-03, Dept. of Math., Univ. of Amsterdam, 1981.
- [69] C.G. Lekkerkerker, Geometry of Numbers (North Holland, Amsterdam, 1969).
- [70] L. Lovász, "Subgraphs with prescribed valencies", JCT 8 (1970), 391-416.
- [72] L. Lovász, "Normal hypergraphs and the perfect graph conjecture", Dis. Math. 2 (1972), 253-267.
- [72a] L. Lovász, "A note on factor-critical graphs", Studia Scientiarum Math. Hungarica 7 (1972), 279-280.
- [75] L. Lovász, "2-matchings and 2-covers of hypergraphs", Acta. Math. Acad. Sci. Hungar. 26 (1975), 433-434.
- [76] L. Lovász, "On two minimax theorems in graph theory", JCT(B) 21 (1976), 96-103.
- [79] L. Lovász, "Graph theory and integer programming", Ann. of Dis. Math. 4 (1979), 141-158.

- [82] A. Lubiw, "T-free matrices", M.Math. Thesis, Univ. of Waterloo, 1982.
- [76] C.L. Lucchesi, A Minimax Equality for Directed Graphs, Ph.D. Thesis, Univ. of Waterloo, 1976.
- [78] C.L. Lucchesi and D.H. Younger, "A minimax theorem for directed graphs", J. London Math. Soc. (2), 17 (1978), 369-374.
- [81] A. Mandel, "On TDI'ness and real linear systems", unpublished paper, 1981.
- [79] A.B. Marsh III, <u>Matching Algorithms</u>, Ph.D. Thesis, The Johns Hopkins Univ., 1979.
- [96] H. Minkowski, Geometrie der Zahlen (Teubner, Leipzig, 1896).
- [75] M. Padberg, "Characterization of totally unimodular, balanced and perfect matrices", in <u>Combinatorial Programming: Methods and Applications</u> (D. Reidel Publishing Company, Dordrechet, 1975), 275-284.
- [73] W.R. Pulleyblank, <u>Faces of Matching Polyhedra</u>, Ph.D. Thesis, Univ. of Waterloo, 1973.
- [80] W. Pulleyblank, "Dual integrality in b-matching problems", Math. Prog. Study 12 (1980), 176-196.
- [81] W.R. Pulleyblank, "Total dual integrality and b-matchings", Operations Res. Letters, Vol. 1, No. 1 (1981), 28-30.
- [82] W.R Pulleyblank, "Polyhedral combinatorics", to appear in the proceedings of the XI International Symposium on Mathematical Programming, Bonn, W. Germany (1982).
- [74] W.R. Pulleyblank and J. Edmonds, "Facets of 1-matching polyhedra", in C. Berge and D.K. Ray-Chaudhuri, eds., <u>Hypergraph Seminar</u>. Springer Verlag (1974), 214-242.

- [70] R.T. Rockafeller, Convex Analysis (Princeton Univ. Press, Princeton, New Jersey, 1970).
- [81] N. Sbihi and J.P. Uhry, "A class of h-perfect graphs", Rapport de Rech. No. 236, IMAG, Grenoble, 1981.
- [80] P. Schonsleben, "Ganzzahige polymatroid-intersections-algorithmen", Ph.D. Thesis, ETH, Zurich, 1980.
- [79] A. Schrijver, "Fractional packing and covering", in A. Schrijver, ed.,
 Packing and Covering (Mathematisch Centrum, Amsterdam, 1979).
- [80] A. Schrijver, "On cutting planes", Annals of Dis. Math. 9 (1980), 291-296.
- [80a] A. Schrijver, "Theory of integer linear programming", preliminary version of a tract on polyhedral methods in combinatorics, 1980.
- [81] A. Schrijver, "On total dual integrality", Linear Alg. and its Appl. 38 (1981), 27-32.
- [81a] A. Schrijver, "Short proofs of the matching polyhedron", to appear in JCT(B), 1981.
- [82] A. Schrijver, "Min-max relations for directed graphs", Annals of Dis. Math. 16 (1982), 261-280.
- [82a] A. Schrijver, "Proving total dual integrality with cross-free families a general framework", Report AE 5/82, Univ. of Amsterdam, 1982 (to appear in Math. Prog.).
- [82b] A. Schrijver, "Total dual integrality from directed graphs, crossing families, and sub- and supermodular functions", to appear in the proceedings of the Silver Jubilee Conference on Combinatorics, University of Waterloo, 1982.

- [82c] A. Schrijver, "Min-max results in combinatorial optimization", to appear in the proceedings of the XI International Symposium on Math. Prog., Bonn, W. Germany, 1982.
- [83] A. Schrijver, "Packing and covering of crossing families of cuts", to appear in JCT(B), 1983.
- [77] A. Schrijver and P.D. Seymour, "A proof of total dual integrality of matching polyhedra", Report ZN79/77, Mathematisch Centrum, Amsterdam, 1977.
- [77] P.D. Seymour, "The matroids with the max-flow min-cut property", JCT(B) 23 (1977), 189-222.
- [79] P.D. Seymour, "On multicolorings of cubic graphs and conjectures of Fulkerson and Tutte", Proc. London Math. Soc. (3) 38 (1979), 423-460.
- [80] P.D. Seymour, "Decomposition of regular matroids", JCT(B) 28 (1980), 305-359.
- [81] P.D. Seymour, "On odd cuts and plane multicommodity flows", Proc. London Math. Soc. (3) 42 (1981), 178-192.
- [61] H.J.S. Smith, "On systems of indeterminate equations and congruences", Phil. Trans. Roy. Soc. London 151 (1861), 293-326.
- [70] J. Stoer and C. Witzgall, <u>Convexity and Optimization in Finite</u>
 Dimensions I (Springer, Berlin, 1970).
- [78] K. Truemper, "Algebraic characterizations of unimodular matrices", SIAM J. Appl. Math., Vol. 35, No. 2, Sept. 1978, 328-332.
- [78] K. Truemper and R. Chandrasekaran, "Local unimodularity of matrix-vector pairs", Lin. Alg. and its Appl. 22 (1978), 65-78.
- [47] W.T. Tutte, "The factorization of linear graphs", J. London Math. Soc. 22 (1947), 107-111.

- [52] W.T. Tutte, "The factors of graphs", Canad. J. Math. 4 (1952), 314-328.
- [54] W.T. Tutte, "A short proof of the factor theorem for finite graphs", Canad. J. Math. 6 (1954), 347-352.
- [81] W.T. Tutte, "Graph factors", Combinatorica 1 (1981), 79-97.
- [40] B.L. van der Waerden, Moderne Algebra II (Springer, Berlin, 1940).
- [68] A. Veinott and G.B. Dantzig, "Integral extreme points", SIAM Review, Vol. 10, No. 3, July 1968, 371-372.
- [75] K. Vidyasankar and D.H. Younger, "A minimax equality related to the longest directed path in an acyclic graph", Can. J. Math. 27 (1975), 348-351.
- [35] H. Weyl, "Elementary theorie der Konvexen Polyeder", Commentarii Math. Helvetici 7 (1935), 290-306.
- [69] D.H. Younger, "Maximum families of disjoint directed cut sets", in W.T. Tutte, ed., <u>Progress in Combinatorics</u> (Academic Press, New York, 1969), 329-333.
- [82] U. Zimmermann, "Augmenting circuit methods for submodular flow problems", Mathematisches Institut, Univ. of Köln, 1982.